SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 8-K

CURRENT REPORT
PURSUANT TO SECTION 13 OR 15(D) OF THE
SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported)
December 17, 2002

Phillips-Van Heusen Corporation
-----(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation)

1-724 13-1166910

(Commission File Number) (IRS Employer Identification Number)

200 Madison Avenue, New York, New York 10016
-----(Address of Principal Executive Offices) (Zip Code)

Registrant's telephone number, including area code (212)-381-3500 $\,$

ITEM 5. OTHER EVENTS

Pursuant to a Stock Purchase Agreement, dated December 17, 2002 ("CKI Purchase Agreement"), among Phillips-Van Heusen Corporation ("PVH"), a Delaware corporation, Calvin Klein, Inc. ("CKI"), a New York corporation, Calvin Klein (Europe), Inc. ("CK Europe I"), a Delaware corporation, Calvin Klein (Europe II) Corp. ("CK Europe II"), a Delaware corporation, Calvin Klein Europe S.r.l. ("CK Italy"), a limited liability company organized under the laws of Italy, CK Service Corp. ("Service Corp." and, together with CKI, CK Europe I, CK Europe II and CK Italy, the "CK Companies"), a Delaware corporation, Calvin Klein, Barry Schwartz and certain family members and affiliated trusts (collectively, the "Sellers"), PVH will purchase from the Sellers all of the issued and outstanding stock of the CK Companies for \$400 million in cash as well as \$30 million of PVH common stock (to be valued based on a 10-trading day period immediately prior to the closing). The transaction also includes, in consideration of certain other rights held by Mr. Klein, a nine-year warrant in favor of Mr. Klein to purchase 320,000 shares of PVH common stock at \$28.00 per share, and contingent purchase price payments to Mr. Klein based on future sales of the Calvin Klein brand.

In order to obtain funds to pay a portion of the purchase price for the CK Companies, PVH entered into a Securities Purchase Agreement (the "Preferred Stock Purchase Agreement") with Lehman Brothers Inc. ("Lehman Brothers") and affiliates of Apax Managers, Inc. and Apax Partners Europe Managers Ltd. (collectively, the "Apax Entities") on December 16, 2002. Pursuant to the Preferred Stock Purchase Agreement, Lehman Brothers will act as a dealer and purchase from PVH and sell to the Apax Entities \$250 million in newly issued convertible preferred stock of PVH. The conversion price for the convertible preferred stock is \$14.00 per share of PVH common stock. The convertible preferred stock has a dividend rate of 8% per annum, payable in kind or cash at PVH's option. In addition, Apax Managers, Inc. has agreed to provide a loan to PVH in the amount of \$125 million which matures in two years. The loan is secured and bears interest at a rate of 10% per annum for the first year and at 15% per annum for the second year.

The transactions are expected to close within 60 days, subject to normal regulatory approvals and customary closing conditions.

The Board of Directors of PVH has approved an amendment to the Rights Agreement, dated as of June 10, 1986, as amended, by and between PVH and The Bank of New York (successor to The Chase Manhattan Bank, N.A.), as rights agent. The amendment will make the provisions of the rights agreement inapplicable to the transactions contemplated by the Preferred Stock Purchase Agreement.

This description of the CKI Purchase Agreement and the Preferred Stock Purchase Agreement are qualified in their entirety by reference to the full text of such documents, copies of which are filed as exhibits to this Form 8-K filing.

- ITEM 7. FINANCIAL STATEMENTS, PRO FORMA FINANCIAL INFORMATION AND EXHIBITS
 - (c) Exhibits:

EXHIBIT DESCRIPTION

- 10.1 Stock Purchase Agreement, dated December 17, 2002, among Phillips-Van Heusen Corporation, Calvin Klein, Inc., Calvin Klein (Europe), Inc., Calvin Klein (Europe II) Corp., Calvin Klein Europe S.r.l., CK Service Corp., Calvin Klein, Barry Schwartz, Trust for the Benefit of the Issue of Calvin Klein, Trust for the Benefit of the Issue of Barry Schwartz, Stephanie Schwartz-Ferdman and Jonathan Schwartz.
- 10.2 Securities Purchase Agreement, dated December 16, 2002, among Phillips-Van Heusen Corporation, Lehman Brothers Inc. and the Investors named therein.
- 99.1 Press Release, dated December 17, 2002.

SIGNATURES

Pursuant to the requirements of the Securities and Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Phillips-Van Heusen Corporation

By: /s/ Mark D. Fischer

Mark D. Fischer, Vice President

Date: December 20, 2002

STOCK PURCHASE AGREEMENT

DATED AS OF DECEMBER 17, 2002

AMONG

PHILLIPS-VAN HEUSEN CORPORATION,

CALVIN KLEIN, INC.,

CALVIN KLEIN (EUROPE), INC.,

CALVIN KLEIN (EUROPE II) CORP.,

CALVIN KLEIN EUROPE S.R.L.,

CK SERVICE CORP.,

CALVIN KLEIN,

BARRY SCHWARTZ,

TRUST FOR THE BENEFIT OF THE ISSUE OF CALVIN KLEIN,
TRUST FOR THE BENEFIT OF THE ISSUE OF BARRY SCHWARTZ,

STEPHANIE SCHWARTZ-FERDMAN

AND

JONATHAN SCHWARTZ

TABLE OF CONTENTS

| | | PAGE |
|---------|--|------|
| ARTICLE | I. CLOSING; SALE AND PURCHASE | 1 |
| 1.1 | The Closing | 1 |
| 1.2 | Sale and Purchase of the Purchase Shares | 2 |
| 1.3 | Delivery of Purchase Price and Stock Certificates | 2 |
| 1.4 | Purchase Price Adjustment | 2 |
| 1.5 | Purchase of Design Services Agreement | 4 |
| 1.6 | Satisfaction of Debt | 8 |
| 1.7 | Releases | 9 |
| | | |
| 1.8 | Actions Simultaneous | 10 |
| ARTICLE | II. REPRESENTATIONS AND WARRANTIES OF THE SELLERS WITH | |
| | RESPECT TO THE PURCHASE SHARES | 10 |
| 2.1 | Authority; Execution and Delivery; Enforceability | 10 |
| 2.2 | Non-Contravention | 10 |
| 2.3 | Title to Purchase Shares | 11 |
| 2.4 | Consents and Approvals | 11 |
| 2.5 | Litigation and Claims | 11 |
| 2.6 | Investment Representations | 11 |
| 2.7 | No Finder | 12 |
| 2.1 | NO FINGEL | 12 |
| ARTICLE | III. REPRESENTATIONS AND WARRANTIES OF THE CK COMPANIES, | |
| | MR. KLEIN AND MR. SCHWARTZ WITH RESPECT TO THE | |
| | CK COMPANIES AND THE CK TRUST | 12 |
| 3.1 | Organization; Good Standing | 12 |
| 3.2 | Equity Interests | 13 |
| 3.3 | Authority; Execution and Delivery; Enforceability | 13 |
| 3.4 | Non-Contravention | 13 |
| 3.5 | Corporate and Trust Documents; Books and Records | 14 |
| 3.6 | Capitalization; Options; Trust Interests | 14 |
| 3.7 | Consents and Approvals | 15 |
| 3.8 | Title to Assets | 15 |
| 3.9 | Real Property | 15 |
| 3.10 | Employment Related Agreements and Actions | 15 |
| 3.11 | | 18 |
| 3.12 | Contracts | |
| | Intellectual Property | 20 |
| 3.13 | Insurance | 22 |
| 3.14 | Financial Statements; Liabilities | 22 |
| 3.15 | Tax Matters | 23 |
| 3.16 | Absence of Certain Changes and Events | 24 |
| 3.17 | Litigation and Claims | 26 |
| 3.18 | Governmental Permits; Compliance with Laws | 26 |
| 3.19 | Environmental Matters | 27 |
| 3.20 | Employee Plans | 27 |
| 3 21 | No Finder | 3.0 |

| ARTICLE | IV. REPRESENTATIONS AND WARRANTIES OF PVH | 30 |
|---------|---|----|
| 4.1 | Organization; Good Standing | 30 |
| 4.2 | Authority; Execution and Delivery; Enforceability | 30 |
| 4.3 | Non-Contravention | 30 |
| 4.4 | Consents and Approvals | 31 |
| 4.5 | Equity Matters | 31 |
| 4.6 | Litigation and Claims | 31 |
| 4.7 | SEC Reports | 31 |
| 4.8 | Private Offering | 32 |
| 4.9 | Capitalization | 32 |
| 4.10 | Listing 34 | |
| 4.11 | Financial Statements | 33 |
| 4.12 | Tax Matters | 33 |
| 4.13 | Intellectual Property | 34 |
| 4.14 | Employee Plans | 34 |
| 4.15 | Compliance with Laws; Permits | 37 |
| 4.16 | Environmental Protection | 38 |
| 4.17 | No Finder | 38 |
| 4.18 | No Reliance | 38 |
| 4.10 | NO VETTURE | 50 |
| ARTICLE | V. ACTION PRIOR TO THE CLOSING DATE | 39 |
| 5.1 | Conduct of Business | 39 |
| 5.2 | No Breach of Representations and Warranties; | |
| | Notification of Certain Matters | 42 |
| 5.3 | Access | 42 |
| 5.4 | Standstill | 42 |
| 5.5 | Notice of Litigation | 43 |
| 5.6 | Fulfillment of Conditions to PVH's Obligations | 43 |
| 5.7 | Fulfillment of Conditions to Sellers' Obligations | 43 |
| 5.8 | Governmental Consents | 43 |
| 5.9 | Third Party Consents | 43 |
| 5.10 | Publicity | 43 |
| 5.11 | Transfer of Certain Assets | 44 |
| ARTICLE | VI. OTHER AGREEMENTS OF THE PARTIES | 44 |
| 6.1 | Cooperation in Litigation; Retention of Employees | 44 |
| 6.2 | Employees | 44 |
| 6.3 | Non-Competition, Non-Solicitation, | |
| | Non-Derogation and Confidentiality | 46 |
| 6.4 | Tax Matters | 48 |
| 6.5 | Access | 51 |
| 6.6 | Directors' and Officers' Indemnification | |
| | and Insurance | 51 |
| 6.7 | Offices | 51 |
| 6.8 | Legend | 51 |
| 6.9 | Certain Matters. Each party agrees to comply with | |
| | Section 6.9 of the CK Companies Disclosure Schedule | 52 |
| 6.10 | Further Assurances | 52 |

| ARTICLE | VII. CONDITIONS PRECEDENT TO OBLIGATIONS OF PVH | 52 |
|---------|--|-----|
| 7.1 | Representations and Warranties | 52 |
| 7.2 | Performance | 52 |
| 7.3 | No Material Adverse Effect | 53 |
| 7.4 | Certificates | 53 |
| 7.5 | No Injunction | 53 |
| 7.6 | CK Debt Obligations; Liens Discharged | 53 |
| 7.7 | HSR Act Clearance | 53 |
| 7.8 | Government Approvals | 53 |
| 7.9 | Third Party Consents | 53 |
| 7.10 | Consulting Agreement | 53 |
| 7.11 | Agreement and Assignment | 53 |
| 7.12 | Notarial Deed | 53 |
| 7.13 | Stock Certificates | 54 |
| 7.14 | Good Standing | 54 |
| 7.15 | Resignations | 54 |
| 7.16 | Stockholder Loans | 54 |
| 7.17 | Termination under Design Services Agreement | 54 |
| 7.18 | Escrow Agreement. The Sellers shall have entered | 5 1 |
| 7.10 | into the Escrow Agreement | 54 |
| | THEO THE ESCIOW AGREEMENT | 74 |
| ARTICLE | VIII. CONDITIONS PRECEDENT TO OBLIGATIONS OF THE SELLERS | 54 |
| 8.1 | Representations and Warranties | 54 |
| 8.2 | Performance | 54 |
| 8.3 | Certificates | 54 |
| 8.4 | No Injunction | 55 |
| 8.5 | HSR Act Clearance | 55 |
| 8.6 | Government Approvals | 55 |
| 8.7 | Consulting Agreement | 55 |
| 8.8 | Agreement and Assignment | 55 |
| 8.9 | Equity Matters | 55 |
| 8.10 | Notarial Deed | 55 |
| 8.11 | Termination under Design Services Agreement | 55 |
| 8.12 | Security Agreements | 55 |
| 8.13 | Registration Rights Agreement | 55 |
| 8.14 | Escrow Agreement. PVH shall have entered into | |
| | the Escrow Agreement | 55 |
| 8.15 | Opinion | 55 |
| ARTICLE | IX. INDEMNIFICATION | 56 |
| 9.1 | Survival | 56 |
| 9.2 | Indemnification by the Sellers | 56 |
| 9.3 | Indemnification by PVH | 58 |
| 9.4 | Limitations on Indemnification | 59 |
| 9.5 | Tax Treatment of Indemnity Payments | 60 |
| 9.6 | Tax Refunds | 60 |
| 9.7 | Notice of Claims | 61 |
| 9.7 | | 61 |
| 9.0 | Third Party Claims | OΤ |

| ARTICLE X. | TERMINATION | 62 |
|------------|-------------------------------------|----------|
| 10.1 | Termination Effects of Termination | 62 62 |
| 10.2 | Effects of Termination | 02 |
| ARTICLE XI | I. MISCELLANEOUS | 63 |
| 11.1 | Expenses of the Transaction | 63 |
| 11.2 | Notices | 63 |
| 11.3 | No Modification Except in Writing | 64 |
| 11.4 | Entire Agreement | 64 |
| 11.5 | Severability | 64 |
| 11.6 | Assignment | 65 |
| 11.7 | Governing Law; Jurisdiction | 65 |
| 11.8 | Specific Performance | 65 |
| 11.9 | Headings; References | 66 |
| 11.10 | Interpretation | 66 |
| 11.11 | Schedules | 66 |
| 11.12 | Third Parties | 66 |
| 11.13 | Counterparts | 66 |
| | | |

APPENDIX

| APPENDIX A. DE | EFINITIONS | A-1 |
|-----------------|----------------------------------|-----------|
| | SCHEDULES | |
| SCHEDULE I. | CK COMPANIES DISCLOSURE SCHEDULE | I-1 |
| SCHEDULE II. | PVH DISCLOSURE SCHEDULE. | II-1 |
| SCHEDULE III. | SELLERS DISCLOSURE SCHEDULE | III-1 |
| SCHEDULE IV. | BALANCE SHEET RULES | IV-1 |
| SCHEDULE V. | LIQUIDATED DAMAGES STATEMENT | V-1 |
| | EXHIBITS | |
| ALLOCATION AMON | NG SELLERS | EXHIBIT A |
| FORM OF ESCROW | AGREEMENT | EXHIBIT E |
| FORM OF DESIGN | SERVICES PAYMENT STATEMENT | EXHIBIT (|
| FORM OF CONSULT | FING AGREEMENT | EXHIBIT I |
| FORM OF AGREEME | ENT AND ASSIGNMENT | EXHIBIT E |
| FORM OF NOTARIA | AL DEED | EXHIBIT E |
| FORM OF SCHWART | IZ DESIGN SERVICES FORM | EXHIBIT O |
| FORM OF REGISTE | RATION RIGHTS AGREEMENT | EXHIBIT F |
| FORM OF DESIGN | SERVICES SECURITY AGREEMENT | EXHIBIT I |
| FORM OF WARRANT | r | EXHIBIT 3 |
| FORM OF INTERCE | REDITOR AGREEMENT | EXHIBIT F |
| FORM OF GUARANT | FEE | EXHIBIT I |

STOCK PURCHASE AGREEMENT

STOCK PURCHASE AGREEMENT ("Agreement"), dated as of December 17, 2002, among PHILLIPS-VAN HEUSEN CORPORATION, a Delaware corporation ("PVH"), CALVIN KLEIN, INC., a New York corporation ("CKI"), CALVIN KLEIN (EUROPE), INC., a Delaware corporation ("CK Europe I"), CALVIN KLEIN (EUROPE II) CORP., a Delaware corporation ("CK Europe II"), CALVIN KLEIN EUROPE S.R.L., a limited liability company organized under the laws of Italy ("CK Italy"), CK SERVICE CORP., a Delaware corporation ("Service Corp." and, together with CKI, CK Europe I, CK Europe II and CK Italy, the "CK Companies"), CALVIN KLEIN ("Mr. Klein"), BARRY SCHWARTZ ("Mr. Schwartz"), TRUST FOR THE BENEFIT OF THE ISSUE OF CALVIN KLEIN (the "Klein Trust"), TRUST FOR THE BENEFIT OF THE ISSUE OF BARRY SCHWARTZ (the "Schwartz Trust"), STEPHANIE SCHWARTZ-FERDMAN ("Schwartz-Ferdman") and JONATHAN SCHWARTZ ("J. Schwartz" and, together with Mr. Klein, Mr. Schwartz, the Klein Trust, the Schwartz Trust and Schwartz-Ferdman, the "Sellers").

WITNESSETH:

WHEREAS, the Sellers are the owners of all of the issued and outstanding shares of the capital stock (or all of the outstanding parts of capital in the case of CK Italy) of each CK Company (the "Purchase Shares");

WHEREAS, PVH desires to purchase and acquire from the Sellers, and the Sellers desire to sell and transfer to PVH, the Purchase Shares on the terms and subject to the conditions hereinafter set forth;

WHEREAS, PVH desires to purchase and acquire from Mr. Klein, and Mr. Klein desires to sell and transfer to PVH, his rights under the Design Services Agreement (as herein defined); and

WHEREAS, terms used in this Agreement and not otherwise defined in this Agreement are defined in Appendix A hereto.

NOW, THEREFORE, the parties hereto hereby agree as follows:

ARTICLE I.

CLOSING; SALE AND PURCHASE

1.1 The Closing. The closing (the "Closing") of the transactions contained in this Article I shall take place at 10:00 A.M., Eastern Time, on the second Business Day after all of the conditions contained in Articles VII and VIII have been satisfied or waived (other than those conditions which will be satisfied at the Closing Time), provided, that, if the HSR Clearance in connection with the Apax Transaction has not been received by such second Business Day, then (i) the conditions contained in Sections 7.1, 7.8 and 7.9 shall be deemed to have been satisfied as of such second Business Day (the "HSR Date") and (ii) the Closing shall take place at 10:00

- A.M., Eastern Time, on the second Business Day after such HSR Clearance has been received; provided, further, that if all the conditions contained in Articles VII and VIII have been satisfied or waived, the Closing shall take place not later than the Termination Date, or at such other time or such other date as PVH and the Sellers may agree, at the offices of Katten Muchin Zavis Rosenman, 575 Madison Avenue, New York, New York. (Hereinafter, such date is referred to as the "Closing Date" and such time on the Closing Date is referred to as the "Closing Time.")
- 1.2 Sale and Purchase of the Purchase Shares. Upon the terms and subject to the conditions set forth herein, at the Closing, the Sellers agree to sell, convey, transfer, assign and deliver to PVH certificates (or such other appropriate evidences of ownership with respect to CK Italy) representing the Purchase Shares, duly endorsed in blank or accompanied by stock or other appropriate powers in blank with all appropriate transfer stamps affixed thereto (the "Stock Certificates"), and PVH (and its designee in the case of CK Italy) agrees to purchase from the Sellers the Purchase Shares, for an aggregate purchase price consisting of (i) \$400,000,000 in cash (the "Cash Purchase Price"), as adjusted in accordance with Section 1.4, and (ii) such number of shares of PVH Common Stock equal to the quotient of (x) \$30,000,000 over (y) the PVH Common Stock Closing Price (the "PVH Shares" and, together with the Cash Purchase Price, the "Purchase Price").
- 1.3 Delivery of Purchase Price and Stock Certificates. (a) At the Closing, (i) the Cash Purchase Price shall be paid by PVH to the Sellers by wire transfer of immediately available funds to accounts designated in writing by the Sellers at least two Business Days prior to the Closing, (ii) subject to Section 1.3(b), the PVH Shares shall be delivered by PVH to the Sellers, and (iii) the Stock Certificates shall be delivered by the Sellers to PVH, and in the case of (i) and (ii) in the proportions as set forth on Exhibit A.
- (b) At the Closing Time, the Holdback Shares shall be deposited in escrow and held pursuant to the terms of an escrow agreement in substantially the form attached hereto as Exhibit B (the "Escrow Agreement"). The Sellers agree, that, in accordance with the terms of the Escrow Agreement, all of the certificates representing the Holdback Shares shall be delivered, along with stock powers executed by such Sellers in blank, to the escrow agent named therein.
- 1.4 Purchase Price Adjustment. In accordance with the procedures set forth in this Section 1.4, the Cash Purchase Price shall be adjusted as follows:
- (a) At least five Business Days prior to the Closing Date, the chief financial officer of CKI, on behalf of the Sellers, shall deliver to PVH a good faith estimate (as adjusted, the "Estimated Closing Net Book Value") of the Closing Date Net Book Value prepared in accordance with GAAP and the resulting Net Book Value Overage or Net Book Value Underage. A "Net Book Value Overage" shall exist when (and shall be equal to the amount by which) the Estimated Closing Net Book Value exceeds the Base Amount. A "Net Book Value Underage" shall exist when (and shall be equal to the amount by which) the Base Amount exceeds the Estimated Closing Net Book Value. Within five Business Days following the Closing Date, in the event of a Net Book Value Overage, PVH shall pay the amount of such Net Book Value Overage to the Sellers in accordance with the procedures set forth in Section 1.3 (subject to adjustment only in the event that the Estimated Closing Net Book Value, on its face,

fails to properly apply GAAP or contains material mathematical error) and in the event of a Net Book Value Underage, PVH shall receive, in accordance with the Escrow Agreement, such number of Holdback Shares equal to the quotient of (i) an amount equal to the Net Book Value Underage, over (ii) the PVH Common Stock Closing Price (subject to adjustment only in the event that the Estimated Closing Net Book Value, on its face, fails to properly apply GAAP or contains material mathematical error). Within 90 days after the Closing Date, PVH shall deliver to the Sellers a combined balance sheet for the CK Companies as of the Closing Date and after giving effect to any payments made or other actions taken by the Sellers to satisfy or discharge the CK Debt Obligations pursuant to Section 1.6 or in satisfaction of the condition contained in Section 7.16 (and prior to giving effect to any action by PVH or its Affiliates that would impact such balance sheet) prepared in accordance with GAAP (the "Closing Date Balance Sheet").

(b) The Closing Date Balance Sheet shall become final and binding upon the parties on the 30th day following delivery thereof, unless the Sellers give written notice of their disagreement with the Closing Date Balance Sheet (a "Notice of Disagreement") to PVH prior to such date. Any Notice of Disagreement shall (i) specify in reasonable detail the nature of any disagreement so asserted and (ii) only include disagreements based on mathematical errors or based on the Closing Date Balance Sheet not being calculated pursuant to this Section 1.4 or in accordance with GAAP. If a Notice of Disagreement is received by PVH in a timely manner, then the Closing Date Balance Sheet (as revised in accordance with this sentence) shall become final and binding upon PVH and the Sellers on the earlier of (A) the date PVH and the Sellers resolve in writing any differences they have with respect to the matters specified in the Notice of Disagreement and (B) the date any disputed matters are finally resolved in writing by the Accounting Firm. During the 30-day period following the delivery of a Notice of Disagreement, PVH and the Sellers shall seek in good faith to resolve in writing any differences that they may have with respect to the matters specified in the Notice of Disagreement. At the end of such 30-day period, PVH and the Sellers shall submit to an independent accounting firm (the "Accounting Firm") for arbitration, in accordance with the standards set forth in this Section 1.4, only matters that remain in dispute and were properly included in the Notice of Disagreement in accordance with this Section 1.4, in the form of a written brief. The Accounting Firm shall be KPMG LLP or, if such firm is unable or unwilling to act, such other nationally recognized independent public accounting firm as shall be agreed upon by PVH and the Sellers in writing. PVH and the Sellers shall use their commercially reasonable efforts to cause the Accounting Firm to render a written decision resolving the matters submitted to the Accounting Firm within 30 days of the receipt of such submission. The scope of the disputes to be resolved by the Accounting Firm shall be limited to whether the items in dispute that were properly included in the Notice of Disagreement were determined pursuant to this Section 1.4 and in accordance with GAAP and whether there were mathematical errors in the Closing Date Balance Sheet, and the Accounting Firm is not to make any other determination, including any determination as to whether the Base Amount is correct. The Accounting Firm's decision shall be based solely on written submissions by PVH and the Sellers and their respective representatives and not by independent review. The Accounting Firm shall address only those items in dispute and may not assign a value greater than the greatest value for such item claimed by either party or smaller than the smallest value for such item claimed by either party. Judgment may be entered upon the determination of the Accounting Firm in any court having jurisdiction over the party against

which such determination is to be enforced. The fees and expenses of the Accounting Firm incurred pursuant to this Section 1.4 shall be borne by PVH and the Sellers equally.

- (c) If the Base Amount is less than the Closing Date Net Book Value, PVH shall, and if the Base Amount is more than the Closing Date Net Book Value, the Sellers shall, within ten Business Days after the Closing Date Balance Sheet becomes final and binding on the parties, make payment to the other party by wire transfer in immediately available funds of the amount of such difference, net of any adjustment made as a result of any Net Book Value Overage or Net Book Value Underage, as applicable, without interest (the "Adjustment Payment"). If the Sellers are required to make an Adjustment Payment to PVH in accordance with this Section 1.4, and such Adjustment Payment is not received by PVH within such ten Business Day period, then, PVH shall receive such number of Holdback Shares equal to the quotient of (i) an amount equal to such Adjustment Payment over (ii) the PVH Common Stock Closing Price, in accordance with the Escrow Agreement. The parties agree that if PVH receives Holdback Shares as a result of a Net Book Value Underage and if PVH is required to make an Adjustment Payment in accordance with this subsection (c), then, PVH shall, first, be entitled to return to the Sellers such number of Holdback Shares equal to the quotient of (i) an amount equal to such Adjustment Payment over (ii) the PVH Common Stock Closing Price, and, second, with respect to the balance of such Adjustment Payment, if any, make payment to the Sellers in accordance with the first sentence of this subsection (c).
- (d) Following the Closing, PVH shall not, and shall not permit the CK Companies to, take any action with respect to the accounting books and records of the CK Companies, or the items reflected thereon, on which the Closing Date Balance Sheet is to be based, that is inconsistent with GAAP or which would impair the ability of any party to make any calculation, or determine the grounds for any Notice of Disagreement, contemplated by this Section 1.4. During the period of time from and after the Closing Date through the final determination of the Closing Date Net Book Value in accordance with this Section 1.4, PVH shall afford, and shall cause the CK Companies to afford, to the Sellers and any accountants, counsel or financial advisers retained by the Sellers in connection with the determination of the Closing Date Net Book Value in accordance with this Section 1.4, access during normal business hours and upon reasonable prior notice to the books, records, personnel and representatives (including the CK Companies' accountants) of the CK Companies relevant to the preparation of the Closing Date Balance Sheet and the determination of the Closing Date Net Book Value in accordance with this Section 1.4.

1.5 Purchase of Design Services Agreement.

(a) Mr. Klein hereby agrees to sell, assign and transfer to PVH at the Closing all of his rights under and pursuant to the Design Services Agreement. In consideration of the foregoing, (i) effective at the Closing Time, PVH shall deliver to Mr. Klein the Warrant, and CKI shall release Mr. Klein from his obligations under the Design Services Agreement, and (ii) within 45 days after the end of each PVH fiscal quarter, commencing with the quarter during which the Closing occurs (the "Initial Quarter") and ending with the first PVH fiscal quarter that ends after the Payment Period, PVH shall make payments (the "Design Services Purchase Payments") to Mr. Klein or his heirs, successors or assigns (the "Klein Heirs") equal to the

Design Services Rate multiplied by the Total Sales during such quarter; provided, however, that with respect to the Initial Ouarter, PVH shall make such Design Services Purchase Payment to Mr. Klein or the Klein Heirs, as the case may be, equal to the product of (x) the Design Services Rate multiplied by Total Sales for the Initial Quarter, and (y) a fraction, (i) the numerator of which is the number of days from and including the Closing Date to and including the last day of the Initial Quarter, and (ii) the denominator of which is the number of days in the Initial Quarter; provided, further, however, that with respect to the quarter during which the Payment Period ends (the "Final Quarter"), PVH shall make such Design Services Purchase Payment to Mr. Klein equal to the product of (x) the Design Services Rate multiplied by Total Sales for the Final Quarter, and (y) a fraction, (i) the numerator of which is the number of days from and including the first day of the Final Quarter to and including the last day of the Payment Period, and (ii) the denominator of which is the number of days in the Final Quarter. For purposes of this Section 1.5, Total Sales shall mean (without duplication), (A) total worldwide wholesale "net sales" (i.e., invoiced price less actual returns, allowances, customary and usual trade discounts granted and itemized insurance, taxes and freight) by PVH and its Affiliates of all products Bearing the Marks marketed, distributed or sold during the applicable PVH fiscal quarter ("Wholesale Direct Sales"), (B) total worldwide wholesale net sales or retail net sales, as the case may be, of all products Bearing the Marks under licenses and other agreements between PVH and its Affiliates, on the one hand, and its licensees and the others to whom PVH or any of its Affiliates has granted the right to use the Marks, on the other hand, which net sales for purposes hereof shall mean the net sales reported by such licensees and the other Persons (taking into account any increases or decreases in such reported amounts through the exercise of audit rights or any other adjustments) during the applicable PVH fiscal quarter ("Licensee Sales"), (C) total worldwide retail sales (i.e., cost to PVH or its Affiliates plus 15% for all regular items and transfer pricing for surplus goods, close-outs and other distressed goods) by PVH and its Affiliates of products Bearing the Marks ("Retail Direct Sales"), (D) total worldwide retail sales (i.e., cost to PVH or its Affiliate plus 15% for all regular items and transfer pricing for surplus goods, close-outs and other distressed goods) by PVH and its Affiliates of all products not Bearing the Marks but which are sold under the CK Umbrella, (E) with respect to goods and services other than products in the Restricted Business, total Wholesale Direct Sales, Licensee Sales and Retail Direct Sales, as the case may be, of all goods and services and other worldwide revenue from goods and services either Bearing the Marks or marketed under the Marks (i.e, the basis upon which PVH or an Affiliate receives payments from a licensee or other authorized user, for example, food and beverage sales in a "CK Restaurant" or hotel rooms in a "CK Hotel"), and (F) with respect to the Administration Agreement, dated March 14, 1994, between CKI and Warnaco Inc., net sales of Women's Intimate Apparel and Men's Underwear (as such terms are defined therein) by Warnaco Inc. and their affiliates and net proceeds of any licensees and sub-licensees of such Warnaco entities, in each case as and to the extent provided in said Administration Agreement. For purposes of this Section 1.5, "Bearing the Marks" shall mean a Mark displayed on the product in question or on its labels or packaging. It is being acknowledged that on or after the Closing Date, PVH may change its corporate name to a name including a Mark (the "New Corporate Name") and after adoption of the New Corporate Name, if the New Corporate Name is displayed on the labeling or packaging of a product to identify (in a manner customary in the industry) the New Corporate Name as merely the manufacturer, distributor or seller of such product or the owner of a trademark, tradename or service mark other than a Mark (e.g., Izod,

manufactured by CKI), then such use of the Corporate Name and such product shall not be included within this definition of Bearing the Marks. For purposes of this Agreement, "CK Umbrella" shall include the marketing, distribution or sale (i) in any store, catalogue, web site or other direct-to-consumer channel (each, a "Channel") controlled by PVH or its Affiliates (a "PVH Channel") the primary branding of which PVH Channel includes one of the Marks or where one of the Marks is a principal element in the marketing or advertising for such PVH Channel; (ii) in any PVH Channel where at least 50% of the net sales are derived from products Bearing the Marks; or (iii) in any Channel where there is a "shop-in-shop" or similar dedicated area for the sale of products Bearing the Marks and PVH or its Affiliates have directed or permitted the operator of such Channel to include for sale within such area other products of PVH not Bearing the Marks. All Design Services Purchase Payments due to Mr. Klein or the Klein Heirs, as the case may be, pursuant to this Section 1.5 shall be paid to Mr. Klein or such Klein Heirs at the address designated pursuant to Section 11.2 or by wire transfer of funds to an account designated by Mr. Klein or such Klein Heirs in a notice to PVH pursuant to Section 11.2. Except as otherwise provided in this Section 1.5, PVH shall not be obligated to pay interest on the Design Services Purchase Payments and shall report the payment of the Design Services Purchase Payments in accordance with the Code.

- (b) At the time each Design Services Purchase Payment is due, PVH shall deliver to Mr. Klein or the Klein Heirs, as the case may be, a written statement (each, a "Payment Statement") setting forth the amount of the Design Services Purchase Payment payable pursuant to Section 1.5(a) with respect to such quarter and the calculation thereof. Each Payment Statement shall be delivered to Mr. Klein or the Klein Heirs, as the case may be, in substantially the form attached hereto as Exhibit C.
- (c) Mr. Klein or the Klein Heirs and/or their respective authorized representatives shall have the right one time with respect to each fiscal year of PVH (although such right may also be exercised as to multiple years (up to three years at any one time) at the same time or as to partial years), during normal business hours and upon reasonable prior notice, during the Payment Period and for two years thereafter, to audit and copy those books, records and supporting documentation of PVH (and to the extent PVH can make them available, the workpapers of PVH's accountants) pertaining to transactions arising out of or relating to this Section 1.5, and may have access to employees of PVH and its Affiliates who are responsible for overseeing or are directly involved in calculating Design Services Purchase Payments for purposes of asking questions and obtaining information relating to such calculations. Acceptance by Mr. Klein or the Klein Heirs, as the case may be, of any Design Services Purchase Payment shall not preclude Mr. Klein or the Klein Heirs, as the case may be, from questioning its correctness; provided, however, that once Mr. Klein or the Klein Heirs, as the case may be, has exercised his or their right to audit the books, records, supporting documentation and/or ask questions of PVH employees pursuant to this Section 1.5(c) with respect to any period during the Payment Period, he or they, as the case may be, shall thereafter be barred from disputing the accuracy of any Design Services Purchase Payment or Payment Statement received during the portion of the Payment Period examined, unless Mr. Klein or the Klein Heirs, as the case may be, raises an objection to any such payment or statement within one year after all reasonably requested books, records, supporting documentation and personnel are made available for such audit.

(d) If Mr. Klein or the Klein Heirs, as the case may be, shall raise any objections with respect to any Payment Statement and related Design Services Purchase Payment during the Payment Period or within the two-year period after the Payment Period (but subject to the limitations set forth in Section 1.5(c)), then Mr. Klein or the Klein Heirs, as the case may be, on the one hand, and PVH, on the other hand, shall attempt promptly to resolve the matter or matters in dispute. If the dispute is so resolved, then the applicable Payment Statement shall be revised in accordance with the resolution, and shall be final and binding on all parties hereto. If the resolution of a dispute requires PVH to pay additional amounts (i.e., the revised calculation of the Design Services Purchase Payment exceeds the amount of the Design Services Purchase Payment set forth in the applicable Payment Statement) then PVH shall make payment to Mr. Klein or the Klein Heirs, as the case may be, of such additional amounts. If such dispute cannot be resolved by Mr. Klein or the Klein Heirs, as the case may be, on the one hand, and PVH, on the other hand, within 30 Business Days after the delivery of the objection, then the specific matters in dispute shall be submitted to mediation in accordance with Section 11.7. Notwithstanding Sections 11.1 and 11.7, if the determination of the aggregate amount of the Design Services Purchase Payments by the mediators for the period of time in dispute (i) exceeds the aggregate amount set forth in the Payment Statements by \$500,000 or more, PVH shall bear all the fees and expenses of the mediation, or (ii) is less than \$100,000 in excess of the aggregate amount thereof set forth in the Payment Statements, Mr. Klein or the Klein Heirs, as the case may be, shall bear all the fees and expenses of the mediation; otherwise each party shall bear its own expenses of such mediation and share equally the other costs of such mediation. If the resolution of a dispute in accordance with this Section 1.5 requires PVH to pay additional amounts to Mr. Klein or the Klein Heirs, as the case may be, then PVH shall make payment to Mr. Klein or the Klein Heirs, as the case may be, of such additional amounts plus, in the case of any such payment which is greater than \$1,000,000, interest at the rate of 15% per annum, and, in the case of any such payment which is equal to or less than \$1,000,000, interest at the prime lending rate as in effect on the date such payment is due as established by JPMorgan Chase Bank (or its successor) in New York, New York, in each case from the date the original payment was due to Mr. Klein or the Klein Heirs, as the case may be, until such payment is made in accordance with this Section 1.5; provided, however, no such interest shall accrue or be required to be paid by PVH as a result of increases in Design Services Purchase Payments because of changes in amounts reported to PVH by the licensees and other third parties to whom rights to use the Marks have been granted; provided, that Mr. Klein is paid all such amounts as part of the next Design Services Purchase Payment due to Mr. Klein. Any Design Services Purchase Payment which is not made to Mr. Klein or the Klein Heirs, as the case may be, as a result of a default in payment by PVH shall accrue interest at 15% per annum; provided, however, no such interest shall accrue or be required to be paid by PVH as a result of increases in Design Services Purchase Payments because of changes in amounts reported to PVH by the licensees and other third parties to whom rights to use the Marks have been granted. Notwithstanding any Design Services Purchase Payment made hereunder, such payment shall not restrict or compromise PVH's right to dispute (with respect to the three-year period prior to any such dispute) that an overpayment was made to Mr. Klein or the Klein Heirs.

(e) For income Tax purposes, PVH, the Sellers and the CK Companies agree that they will report the Design Services Purchase Payments as made in exchange solely for the surrender by Mr. Klein of all his rights under the Design Services Agreement and that except for

the portion of such payment required to be treated as interest for income Tax purposes and except for changes in Law, the parties agree to file their Tax Returns accordingly.

- (f) If PVH defaults in the payment of any Design Services Purchase Payment for more than 30 days after the same becomes due and payable or any other Event of Default (as defined in the Design Services Security Agreement) occurs (each such default, a "Design Services Purchase Payment Default"), Mr. Klein or the Klein Heirs may, at their option declare that the entire amount of all unpaid Design Services Purchase Payments together with the Liquidated Damages Amount shall be immediately due and payable, in each and every case without presentment, demand, protest or further notice, all of which are hereby waived. The parties acknowledge that, in the event of a Design Services Purchase Payment Default, precise damages would be difficult or impossible to calculate and expressly agree that the Liquidated Damages Amount is a fair and reasonable estimate of such damages and that the Liquidated Damages Amount is not a penalty. The parties acknowledge and agree that the provisions of this Section 1.5(f) are for the mutual benefit of the parties and are intended to avoid a protracted dispute respecting damages occurring by reason of a Design Services Purchase Payment Default. In the event the Design Services Purchase Payments together with the Liquidated Damages Amount shall be declared immediately due and payable pursuant to this Section 1.5(f), PVH shall have the right to make a one-time request within 30 days of such declaration by giving written notice to Mr. Klein or the Klein Heirs, as the case may be, and upon receipt of such notice, Mr. Klein or the Klein Heirs, as the case may be, shall rescind any such declaration and its consequences if (i) the Design Services Purchase Payment Default upon which such declaration was based shall have resulted from the inability of PVH to receive information from third parties necessary to calculate the amount of such Design Services Purchase Payment in a timely manner, (ii) PVH has paid in full all overdue Design Services Purchase Payments and all overdue interest on such Design Services Purchase Payments, (iii) no default or event of default shall have occurred and be continuing with respect to the Design Services Security Agreement, the Guarantee or any Material Indebtedness (as defined in the Design Services Security Agreement), and (iv) no judgment or decree has been entered for the payment of any monies due pursuant to this Agreement. No rescission under this Section 1.5(f) will extend to or affect any subsequent Design Services Purchase Payment Default or impair any right arising in connection therewith.
- (g) In connection with this Section 1.5, at the Closing, PVH and Mr. Klein shall enter into the Design Services Security Agreement.
- (h) Except as otherwise provided in this subsection (h), all amounts payable pursuant to this Section 1.5 shall be paid without reduction, set-off or counterclaim of any kind. Any breach by Mr. Klein of any of his obligations under this Agreement for which amounts are required to be paid to a PVH Group Member, upon and in accordance with a final non-appealable judgment by a court of competent jurisdiction finding such a breach, or otherwise as agreed to in writing by the parties, shall entitle PVH to offset against the Design Services Purchase Payments.
- 1.6 Satisfaction of Debt. The Sellers shall cause all of the obligations (including, without limitation, any contingent obligations, whether by guarantee or otherwise), of each CK

Company in respect of debt for borrowed money (other than any Customs performance bonds, trade letters of credit covering purchases of inventory and stand-by letters of credit covering obligations under certain leases and the capitalized leases, guarantees of employee credit cards and any intercompany indebtedness set forth in the CK Companies Disclosure Schedule), together with any interest accruing on any such obligation (the "CK Debt Obligations") to be repaid in full and cancelled by no later than the Closing Time, or shall cause such CK Company to be released and discharged from such obligations effective prior to the Closing Time and shall cause all Liens granted under the Line Letter to be discharged; it being understood that (i) the CK Debt Obligations may be paid simultaneously at Closing with the proceeds of the Cash Purchase Price, and (ii) PVH shall not acquire or assume any obligations of any CK Company in respect of debt for borrowed money except as provided in this Section 1.6. Set forth on the CK Companies Disclosure Schedule is a list of all of the performance bonds and trade letters of credit of the CK Companies as are in effect on the date hereof. Within ten Business Days prior to the Closing, the Sellers shall provide to PVH a list of each then outstanding letter of credit, stating the letter of credit number, the issuing bank, the name, address and country of the beneficiary, the purchase orders covered, the expiration date and the amount outstanding on the day the list is compiled with respect to each outstanding letter of credit, as well as a list of each Customs performance bond then outstanding, stating the amount, issuer, beneficiary, permitted importers, amount of premium and date of the last premium payment and the due date of the next premium payment with respect to each such bond. Within five Business Days prior to the Closing, PVH shall provide to the Sellers a list of the letters of credit and Customs performance bonds it can and is willing to assume and will replace such letters of credit and performance bonds not being assumed by PVH. Notwithstanding anything to the contrary contained in this Agreement, in no event shall PVH's inability or unwillingness to assume any such letters of credit and Customs performance bonds give rise to any breach on the part of the Sellers or the CK Companies hereunder or the failure of any condition to PVH's obligation to consummate the transactions contemplated hereby as it relates to the obligations of the CK Companies underlying such letters of credit and Customs performance bonds or in respect of which such letters of credit or performance bonds were issued or the impact thereof on the business of the CK

1.7 Releases. Each Seller, on behalf of himself, herself or itself, and also on behalf of his, her or its heirs, successors and/or assigns, unconditionally and irrevocably waives and relinquishes as of the Closing Date, any claims (and resulting damages) which he, she or it has, had, may have or claims to have had against any CK Company, the CK Trust, PVH or PVH's Affiliates (in the case of PVH and its Affiliates (other than any CK Company or the CK Trust) solely as a result of PVH's ownership of the CK Companies after the Closing Date), from the beginning of the world through the date of the Closing, other than claims in respect of, or relating to, this Agreement, each Related Agreement and the transactions contemplated hereby. PVH, on behalf of itself, its Affiliates and the CK Companies (as constituted after the Closing) and also on behalf of their respective heirs, successors and/or assigns, unconditionally and irrevocably waives and relinquishes as of the Closing Date, any claims (and resulting damages) which the CK Companies, the CK Trust, PVH or PVH's Affiliates (in the case of PVH and its Affiliates, solely as a result of PVH's ownership of the CK Companies after the Closing Date) has, had, may have or claims to have had against any Seller from the beginning of the world through the date of the Closing, other than claims in respect of, or relating to, this Agreement, each Related Agreement and the transactions contemplated hereby.

9

1.8 Actions Simultaneous. All actions to be taken and all documents to be executed and delivered by all parties at the Closing shall be deemed to have been taken and executed and delivered simultaneously and no actions shall be deemed to have been taken nor shall any documents be deemed to have been executed and delivered until all actions have been taken and all documents have been executed and delivered.

ARTICLE II.

REPRESENTATIONS AND WARRANTIES OF THE SELLERS WITH RESPECT TO THE PURCHASE SHARES

Each Seller, individually as to such Seller, hereby represents and warrants to PVH as follows (provided, that, Mr. Klein, jointly and severally, makes the representations and warranties contained in this Article II with respect to the other Klein Group Seller, and Mr. Schwartz, jointly and severally, makes the representations and warranties contained in this Article II with respect to the other Schwartz Group Sellers):

- 2.1 Authority; Execution and Delivery; Enforceability.
- (a) Such Seller has full power, authority and capacity to execute and deliver this Agreement and, to the extent a party thereto, the Related Agreements, to perform such Seller's respective obligations hereunder and under such Related Agreements and to consummate the transactions contemplated hereby and by such Related Agreements. Each of this Agreement and (when executed) the Related Agreements has been (or will be) duly executed and delivered by such Seller (to the extent a party thereto), and constitutes (or will, when executed, constitute) the legal, valid and binding obligation of such Seller (to the extent a party thereto), enforceable against such Seller in accordance with its terms, subject to bankruptcy, insolvency, fraudulent transfer, moratorium and other similar Laws of general applicability relating to or affecting creditors' rights and to general equity principles.
- (b) In the case of the Klein Trust and the Schwartz Trust, all acts and other proceedings required to be taken by such Trust to authorize the execution, delivery and performance of this Agreement and the Related Agreements have been duly and properly taken. The Klein Trust was created under agreement dated April 1, 1976 between Mr. Klein, as grantor, and Mr. Schwartz and Charles Ballon, as trustees. The Schwartz Trust was created under agreement dated April 1, 1976 between Mr. Schwartz, as grantor, and Mr. Klein and Charles Ballon, as trustees.

2.2 Non-Contravention.

(a) The execution and delivery of this Agreement and the Related Agreements by such Seller (to the extent a party thereto) does not, and the consummation by such Seller of the transactions contemplated hereby and by such Related Agreements and compliance by such Seller with the terms hereof and of such Related Agreements, will not:

- (i) constitute a default under or a violation or breach of, or result in the acceleration of any obligation under, any provision of any contract or other instrument to which such Seller is a party, except to the extent such default, violation, breach or acceleration does not materially adversely affect such Seller's ability to consummate the transactions contemplated hereby and by such Related Agreements;
- (ii) assuming the consents described in Section 2.4 have been received, violate any Order or any Law affecting such Seller; or
- (iii) result in the creation of any Lien on such Seller's Purchase Shares, other than any Lien created by or through any action of PVH or its Affiliates.
- (b) In the case of the Klein Trust and the Schwartz Trust, the execution and delivery of this Agreement and the Related Agreements by such Trust (to the extent a party thereto) does not, and the consummation of the transactions contemplated hereby and by such Related Agreements and compliance with the terms hereof and of such Related Agreements will not, constitute a violation or breach of the trust agreement governing such Trust or any related trust instrument.
- 2.3 Title to Purchase Shares. Such Seller has good and valid title to the Purchase Shares owned by such Seller as set forth on the Sellers Disclosure Schedule, free and clear of all Liens. Except as set forth in the Sellers Disclosure Schedule, such Seller is not currently bound by any contract, agreement, arrangement, commitment or understanding with, or has not granted any option or right currently in effect or which would arise after the date hereof to, any party other than PVH with respect to the acquisition of any of such Seller's Purchase Shares.
- 2.4 Consents and Approvals. Except as set forth in the Sellers Disclosure Schedule, no consent, approval, license, permit, order or authorization of, or registration, declaration or filing with, any Governmental Authority or Person is required to be obtained by or on behalf of such Seller in connection with the execution, delivery and performance of this Agreement or the Related Agreements (to the extent a party thereto) or the consummation of the transactions contemplated hereby and by such Related Agreements, other than compliance with and filings under the HSR Act and similar Laws of any jurisdiction.
- 2.5 Litigation and Claims. There is no Action pending or, to the Knowledge of such Seller, threatened, against or affecting such Seller that could reasonably be expected to materially adversely affect such Seller's ability to consummate the transactions contemplated hereby or by the Related Agreements (to the extent a party thereto).
- 2.6 Investment Representations. Such Seller understands that the PVH Shares, and Mr. Klein understands that the Warrant and the shares of PVH Common Stock underlying such Warrant, will not have been registered under the Securities Act. Such Seller also understands that the PVH Shares, and Mr. Klein also understands that the Warrant and the shares of PVH Common Stock underlying such Warrant, are being offered and sold pursuant to an exemption from registration contained in the Securities Act based in part upon the representations of such

Seller contained in this Agreement. Such Seller hereby represents and warrants to PVH as follows:

- (i) Such Seller is capable of evaluating the merits and risks of the investment in PVH and has the capacity to protect such Seller's own interests. Such Seller must bear the economic risk of this investment indefinitely unless the relevant securities are registered pursuant to the Securities Act, or an exemption from registration is available. Such Seller also understands that there is no assurance that any exemption from registration under the Securities Act will be available and that, even if available, such exemption may not allow such Seller to transfer all or any portion of such securities under the circumstances, in the amounts or at the times such Seller might propose.
- (ii) Such Seller is acquiring the relevant securities for such Seller's own account for investment only, and not with a view towards its distribution.
- (iii) Such Seller represents that such Seller is an "accredited investor" within the meaning of Regulation D under the Securities Act.
- 2.7 No Finder. Except as set forth in the Sellers Disclosure Schedule, neither such Seller nor any party acting on such Seller's behalf has paid or become obligated to pay any fee or commission to any broker, finder or intermediary (other than counsel) for or on account of the transactions contemplated hereby or by the Related Agreements.

ARTICLE III.

REPRESENTATIONS AND WARRANTIES OF THE CK COMPANIES, MR. KLEIN AND MR. SCHWARTZ WITH RESPECT TO THE CK COMPANIES AND THE CK TRUST

Each of Mr. Klein and Mr. Schwartz and each CK Company, jointly and severally (but subject to the limitations contained in Section 9.4), hereby represents and warrants to PVH as follows:

- 3.1 Organization; Good Standing.
- (a) Each CK Company is a corporation duly organized, validly existing and in good standing under the Laws of the jurisdiction in which it is organized. Each CK Company has full corporate power and authority to conduct all of the business and activities conducted by it, and to own or lease all of the assets owned or leased by it; and is duly licensed or qualified to do business and is in good standing as a foreign corporation in all jurisdictions in which the nature of the business and activities conducted by it, and/or the character of the assets owned or leased by it, makes such qualification or license necessary, except for those jurisdictions in which the failure to be so qualified or licensed would not, individually or in the aggregate, limit any CK Company's ability to consummate the transactions contemplated hereby and by the Related Agreements or have a CK Companies Material Adverse Effect.

- (b) The CK Trust is a Delaware business trust, was duly established under the CK Trust Agreement and remains in good standing under the Laws of the State of Delaware.
- 3.2 Equity Interests. Except as set forth in the CK Companies Disclosure Schedule, none of the CK Companies, directly or indirectly, own any capital stock of or other equity interests in any Person. Except as set forth in the CK Companies Disclosure Schedule, none of the CK Companies is a participant in any joint venture, partnership, or similar arrangement, it being acknowledged and agreed by PVH that the license agreements for the Marks do not create such an arrangement as a result of CKI being entitled to share in certain revenue, or have licensees or others share in certain costs.
- 3.3 Authority; Execution and Delivery; Enforceability. Each CK Company has full corporate power and authority to execute and deliver this Agreement and, to the extent a party thereto, the Related Agreements, to perform its respective obligations hereunder and under such Related Agreements and to consummate the transactions contemplated hereby and by such Related Agreements. All corporate acts and other proceedings required to be taken by each CK Company to authorize the execution, delivery and performance of this Agreement and such Related Agreements have been duly and properly taken. Each of this Agreement and (when executed) the Related Agreements has been (or will be) duly executed and delivered by each CK Company (to the extent a party thereto), and constitutes (or will, when executed, constitute) the legal, valid and binding obligation of each CK Company (to the extent a party thereto), enforceable against each CK Company in accordance with its terms, subject to bankruptcy, insolvency, fraudulent transfer, moratorium and other similar Laws of general applicability relating to or affecting creditors' rights and to general equity principles.
- 3.4 Non-Contravention. The execution and delivery of this Agreement and the Related Agreements by each CK Company (to the extent a party thereto) does not, and the consummation by any such CK Company of the transactions contemplated hereby and by such Related Agreements and compliance by any such CK Company with the terms hereof and of such Related Agreements, will not:
 - (i) constitute a violation or breach of the articles or certificate of incorporation, as applicable, or the by-laws of any CK Company;
 - (ii) except as set forth in the CK Companies Disclosure Schedule, constitute a default under or a violation or breach of, or result in the acceleration of any obligation of a CK Company under, or a change in any right or obligation of, a CK Company or counterparty, under, any provision of any Contract required to be listed in a Schedule to this Agreement (including, without limitation, the CK Trust Agreement) or by which (a) any of the assets of any CK Company or (b) the Purchase Shares may be affected or secured;
 - (iii) assuming the consents described in Section 3.7 have been received, violate any Order or any Law affecting any CK Company or the CK Trust, or the assets of any CK Company or the CK Trust;

- (iv) result in the creation of any Lien on any of the assets of any CK Company or the CK Trust, other than a Permitted Lien or any Lien created by or through any action of PVH or its Affiliates; or
- (v) assuming the consents described in Section 3.7 have been received, result in the termination of (A) any license agreement for any Mark, (B) any material license (excluding licenses for the Marks), franchise, lease or permit to which any CK Company or the CK Trust is a party or by which it is bound, or (C) any material right of any CK Company or the CK Trust under any of the foregoing.
- 3.5 Corporate and Trust Documents; Books and Records. The Sellers have made available to PVH complete and correct copies of the articles or certificate of incorporation, as applicable, by-laws and stock transfer books of each CK Company, and of the certificate of trust relating to the CK Trust, the CK Trust Agreement and any related agreements thereto. The Sellers have made available to PVH complete and correct in all material respects copies of all minute books and all other existing records of any meeting of the board of directors or other similar governing body (and any committee thereof) or stockholders (or partholders in the case of CK Italy) of each CK Company.
 - 3.6 Capitalization; Options; Trust Interests.
- (a) The CK Companies Disclosure Schedule sets forth for each CK Company the amount of its authorized capital stock, the amount of its outstanding capital stock and the record and beneficial owners of its outstanding capital stock. All the outstanding shares of capital stock (and all of the outstanding parts of capital in the case of CK Italy) of each CK Company have been duly authorized and validly issued and are fully paid and non-assessable, free and clear of all Liens. There are no outstanding warrants, options, contracts, rights (preemptive or otherwise), calls or commitments of any character binding on a CK Company relating to any authorized and issued or unissued shares of capital stock (or parts of capital in the case of CK Italy) of any CK Company or other instruments binding on any CK Company convertible into or exchangeable for such stock, or which obligate any CK Company to seek authorization to issue additional shares of any class of stock (or parts of capital in the case of CK Italy), nor will any be created by virtue of this Agreement or the Related Agreements or the transactions contemplated hereby or by the Related Agreements. The Purchase Shares constitute all of the outstanding shares of capital stock (and all of the outstanding parts of capital in the case of CK Italy) of the CK Companies.
- (b) The CK Trust estate is divided into three series of beneficial ownership: Class A, Class B and Class C, with separate rights with respect to that portion of the CK Trust estate allocated to that series. The CK Companies Disclosure Schedule sets forth for the CK Trust the outstanding ownership interests and the record and beneficial owners of its outstanding ownership interests. All the outstanding ownership interests in the CK Trust have been duly authorized and validly issued in compliance with the CK Trust Agreement, free and clear of all Liens, and CKI has made all required contributions to the CK Trust under the CK Trust Agreement. Except as set forth in the CK Companies Disclosure Schedule, there are no other classes of ownership interest of the CK Trust authorized or issued. There are no outstanding

warrants, options, contracts, rights (preemptive or otherwise), calls or commitments of any character binding on the CK Trust relating to any interests of the CK Trust or other instruments convertible into or exchangeable for such interests, or which obligate the CK Trust to seek authorization to issue additional interests, nor will any be created by virtue of this Agreement or the Related Agreements or the transactions contemplated hereby or by the Related Agreements.

- 3.7 Consents and Approvals. Except as set forth in the CK Companies Disclosure Schedule, no consent, approval, license, permit, order or authorization of, or registration, declaration or filing with, any Governmental Authority, and no consent or approval of any Person under any Contract required to be listed in a Schedule to this Agreement, is required to be obtained by or on behalf of any CK Company or the CK Trust in connection with the execution, delivery and performance of this Agreement or the Related Agreements (to the extent a party thereto) or the consummation of the transactions contemplated hereby and by such Related Agreements, other than compliance with and filings under the HSR Act and similar Laws of any jurisdiction.
 - 3.8 Title to Assets.
- (a) Each of the CK Companies and the CK Trust has good and valid title to (or sufficient rights to use), free and clear of all Liens (other than Permitted Liens and Liens granted under the Line Letter), all of its assets, not including the CK Intellectual Property Rights, the representations with respect to which are set forth in Section 3.12.
- (b) Except as set forth in the CK Companies Disclosure Schedule, none of the Sellers or any of their Affiliates (other than the CK Companies and the CK Trust) own any assets primarily used in or necessary to conduct the business of any CK Company.
- (c) All of the material tangible personal property of the CK Companies has been maintained in all material respects in accordance with generally accepted industry practice. All of the material leased personal property of the CK Companies is in all material respects in the condition required of such property by the terms of the lease applicable thereto during the relevant term of the lease.
 - 3.9 Real Property.
 - (a) None of the CK Companies owns any real property.
- (b) The CK Companies Disclosure Schedule lists all Real Property Leases. Complete and correct copies of each Real Property Lease have been provided to PVH.
 - 3.10 Employment Related Agreements and Actions.
- (a) The CK Companies Disclosure Schedule contains a complete and correct list, as of the date hereof, of the directors and the officers of each CK Company.
- (b) The CK Companies Disclosure Schedule contains a complete and correct list, as of the date hereof, of all Contracts currently in effect with current or former employees,

consultants, or independent contractors of each CK Company in each case which provides for payments in excess of \$75,000 per annum or \$250,000 in the aggregate (in the case of Contracts providing for "at-will" employment only to the extent such payments are guaranteed following termination of employment).

- (c) Except as set forth in the CK Companies Disclosure Schedule and with respect to current employees of any CK Company whose primary work location is in the United States:
 - (i) none of the CK Companies is a party to any collective bargaining agreement or other contract with any labor organization or other representative of the CK Companies' employees, nor is any such contract presently being negotiated, and no question concerning representation exists or has been raised with respect to any of the employees of the CK Companies since January 1, 1999, nor, to the Knowledge of the Sellers or any CK Company, are there any campaigns being conducted to solicit cards from the CK Companies' employees to authorize representation by any labor organization;
 - (ii) there is no labor strike, union-related slowdown, work stoppage, lockout or other union-related labor controversy in effect, or to the Knowledge of the Sellers or any CK Company, threatened, against or otherwise affecting any CK Company, and none of the CK Companies has experienced any such event since January 1, 1999, and there is no unfair labor practice, charge or complaint pending or, to the Knowledge of the Sellers or any CK Company, threatened, against any CK Company;
 - (iii) no material Action by or before any Governmental Authority brought by or on behalf of any employee, prospective employee, former employee, retiree, labor organization or other representative of the employees of any CK Company is pending or, to the Knowledge of the Sellers or any CK Company, threatened, against any CK Company, and none of the CK Companies is a party to, or otherwise bound by, any consent decree with, or citation by, any Governmental Authority relating to employees or employment practices at any CK Company;
 - (iv) no grievance is pending or, to the Knowledge of the Sellers or any CK Company, threatened, by any Person or Persons at any CK Company against any CK Company which would reasonably be expected to have a CK Companies Material Adverse Effect;
 - (v) each CK Company is in compliance in all material respects with all applicable Laws, Contracts and policies relating to employment, employment practices, wages, hours, occupational safety and health, the withholding and payment of Taxes from or with respect to the compensation of employees, and terms and conditions of employment;
 - (vi) each CK Company has paid in full or accrued (subject to any good faith dispute) to all of its employees all wages, salaries, commissions, bonuses, benefits and

other compensation due to such employees or otherwise arising under any policy, practice, agreement, plan, program, statute or other applicable Law:

- (vii) none of the CK Companies is closing, or since January 1, 1999, has closed any Facility, effectuated any company-wide layoffs of employees or implemented any early retirement, separation or window program, nor has any CK Company planned or announced any such action or program for the future; and
- (viii) each CK Company is in compliance in all material respects with its obligations pursuant to WARN, and all other notification and bargaining obligations arising under any collective bargaining agreement or statute.
- (d) With respect to current employees of any CK Company whose primary work location is outside the United States ("International Employees") and except as set forth in the CK Companies Disclosure Schedules:
 - (i) none of the CK Companies is a party to any collective bargaining agreement or other contract with any labor organization or other representative of the International Employees, recognized for the purposes of the collective bargaining of terms and conditions, nor is any agreement presently being negotiated, and no question concerning representation exists or has been raised with respect to any of the International Employees since January 1, 1999, nor, to the Knowledge of the Sellers or any CK Company, are there any campaigns being conducted to solicit registration with any union or subscription to collective bargaining agreement from the International Employees to authorize representation by any labor organization;
 - (ii) there is no labor strike, union-related slowdown, work stoppage, lockout or other union-related labor controversy in effect, or to the Knowledge of the Sellers or any CK Company, threatened, against or otherwise affecting any CK Company, and none of the CK Companies has experienced any such event since January 1, 1999, and there is no unfair labor practice, charge or complaint pending or, to the Knowledge of the Sellers or any CK Company, threatened, against any CK Company;
 - (iii) no material Action by or before any Governmental Authority or union representative, brought by or on behalf of any International Employee, prospective employee, former employee, retiree, labor organization or other representative of the International Employees is pending or, to the Knowledge of the Sellers or any CK Company, threatened, against any CK Company, and none of the CK Companies is a party to, or otherwise bound by, any consent decree with, or citation by, any Governmental Authority or union representative relating to employees or employment practices at any CK Company;
 - (iv) no formal grievance is pending or, to the Knowledge of the Sellers or any CK Company, threatened, by any Person or Persons at any CK Company against any CK Company which would reasonably be expected to have a CK Companies Material Adverse Effect;

- (v) each CK Company is in compliance in all material respects with all applicable Laws and individual and/or collective Contracts, and policies including, but not limited to, relating to employment, employment practices, wages, hours, holidays, occupational safety and health, social security contribution, accruals for termination indemnity, the withholding and payment of Taxes from or with respect to the compensation of International Employees, and terms and conditions of employment;
- (vi) each CK Company has paid in full or accrued (subject to any good faith dispute) to all of its International Employees all wages, salaries, commissions, bonuses, benefits, holidays, termination notices, indemnity, refunds and reinbursements, and other compensation due to such employees or otherwise arising under any policy, practice, individual and/or collective agreement, plan, program, statute or other applicable Law;
- (vii) none of the CK Companies is closing, or since January 1, 1999, has closed any Facility, effectuated any company-wide and partial layoffs of International Employees or implemented any early retirement, incentive to resignation, separation or window program, nor has any CK Company planned or announced any such action or program for the future; and
- (viii) each CK Company is in compliance in all material respects with its obligations pursuant to any applicable Law relating to factory or facility closings and all other notification and bargaining obligations arising under any collective bargaining agreement or statute.

3.11 Contracts.

- (a) The CK Companies Disclosure Schedule contains a complete and correct list, as of the date hereof, of all Contracts (other than Contracts between or among the CK Companies) that involve payments by, or to, any CK Company or the CK Trust, of more than \$250,000 per annum or \$500,000 in the aggregate and all Contracts without regard to dollar amount, or such lower amount expressly set forth, in the following categories:
 - (i) commitments or agreements for services for which a prepayment or advance has been made to, or by, or on behalf of, any CK Company or the CK Trust in excess of \$100,000 per commitment or agreement (other than inventory or advertising related purchase order or commitments, licensing arrangements or minimum guaranteed payments for advertising in the ordinary course of business consistent with past practice (collectively, "Permitted Arrangements"));
 - (ii) partnership or joint venture Contracts or arrangements or any other agreements involving a sharing of revenue or profits (other than license agreements described below and sales associate and agency commission arrangements);
 - (iii) Contracts restricting any CK Company or the CK Trust from carrying on its business or activities, as the case may be, in any material respect in its usual and customary manner in any jurisdiction;

- (iv) any non-competition agreements in favor of any CK Company (other than employment or consulting agreements);
- (v) except as set forth in clause (ix) below, each Contract between any CK Company or the CK Trust, on the one hand, and any Affiliate of any CK Company or the CK Trust, on the other hand;
- (vi) any Contracts for the sale or other disposition by any CK Company or the CK Trust of any of its assets in excess of \$50,000 other than in the ordinary course of business, consistent with past practice, and in no event in excess of \$100,000 (other than the sale or disposition of inventory in the ordinary course of business consistent with past practice);
- (vii) each Contract granting to a third party the right to use any CK Intellectual Property Rights to manufacture and sell products, other than Contracts that may be terminated upon less than 90 days' notice by a CK Company granting the right to manufacture and sell products for seasonal purposes, and each co-existence agreement relating to any Mark;
- (viii) any letters of credit, Customs performance bonds or capitalized leases which will not be satisfied at or prior to Closing; and
- (ix) each outstanding loan or advance made by any CK Company to any director, officer, employee, stockholder or other Affiliate of such CK Company (other than any intercompany indebtedness and any business-related advances to employees made in the ordinary course of business, consistent with past practice and in an amount not in excess of \$25,000 per employee or \$100,000 in the aggregate).
- (b) Except as set forth in the CK Companies Disclosure Schedule, each CK Company and the CK Trust has, with respect to all Contracts required to be listed in any Schedule to this Agreement, performed in all material respects all obligations required to be performed by it, and is not in default in any material respect under any such Contract, and, to the Knowledge of the Sellers or any CK Company, no other party to any such Contract is in default in any material respect under any such Contract. Except as set forth in the CK Companies Disclosure Schedule, no event has occurred which, with the lapse of time or the giving of notice or both, would constitute a default in any material respect by any CK Company or the CK Trust, or, to the Knowledge of the Sellers or any CK Company, by any other party to any such Contract. Solely for purposes of Section 7.1, the term "Contract" used in Sections 3.4(ii), 3.7 and this Section 3.11(b) shall also include any Contract entered into after the date of this Agreement that would have been required to be listed in a Schedule to this Agreement had such Contract been in effect as of the date of this Agreement. For purposes of Sections 3.4(ii), 3.7 and this Section 3.11(b), representations that are qualified by Knowledge with respect to another party's compliance shall be deemed not to include the Knowledge qualifier if such other party to the Contract is an Affiliate required to be listed on the CK Companies Disclosure Schedule pursuant to Section 3.2.

3.12 Intellectual Property.

- (a) For the purposes of this Agreement, "Intellectual Property Rights" shall mean (i) all trademarks, service marks, trade dress, design marks, logos, trade names, domain names, web-sites, brand names and corporate names, whether registered or unregistered, active or inactive, together with all goodwill associated therewith, and all applications, registrations and renewals in connection therewith, (ii) all inventions and designs (whether patentable or unpatentable and whether or not reduced to practice), all improvements thereto, and all patents, patent applications, and patent disclosures, together with all reissuances, continuations, continuations-in-part, revisions, extensions and reexaminations thereof, (iii) all artwork, photographs, advertising and promotional materials and computer software and all copyright applications, registrations and renewals in connection therewith, (iv) all trade secrets and confidential business information (including ideas, research and development, know-how, formulas, compositions, manufacturing and production processes and techniques, technical data, designs, drawings, specifications, customer and supplier lists, pricing and cost information), and (v) all other rights in all of the foregoing, including such rights as are provided by treaties, conventions and common law; and the "CK Intellectual Property Rights" shall mean all of the foregoing, including the Marks, and, (vi) the library of historical examples of prior "Calvin Klein" women's collection apparel lines (and certain other "Calvin Klein" apparel items), as well as the CAD systems with historical data and information relating to such prior lines, and (vii) all rights to pursue, recover and retain damages and costs and attorneys' fees for past, present and future infringement of any of the foregoing, in each instance owned or used by the CK Companies or the CK Trust. Notwithstanding the foregoing, the definition of "CK Intellectual Property Rights" shall exclude (A) any Intellectual Property Rights owned by, beneficially owned by and/or licensed for use by the CK Trust to the extent evidenced by its Class B and Class C ownership interests and (B) computer software that may be purchased over the counter.
- (b) The CK Companies Disclosure Schedule sets forth a complete and correct list, as of the date hereof of all: (i) subsisting registrations and applications for registration in the name of any CK Company or the CK Trust for trademarks, service marks, trade names, corporate names, brand names, logos, domain names, patents and copyrights included in the CK Intellectual Property Rights; (ii) unregistered and/or common law Intellectual Property Rights included in the CK Intellectual Property Rights and material to the business of the CK Companies as presently conducted; and (iii) pending Actions (including, but not limited to those in Patent and Trademark Offices and courts) directly related to any CK Intellectual Property Rights and material Actions threatened in writing within the six years in respect of patents, and the three years in respect of any other CK Intellectual Property Rights, in each case, prior to the date of this Agreement (e.g., via cease-and-desist letters) directly relating to any CK Intellectual Property Rights.
- (c) The CK Trust and the CK Companies own, beneficially own and/or are licensed or otherwise have the right to use, all Intellectual Property Rights necessary to conduct, or material to, the businesses of the CK Companies as they are currently conducted. The CK Trust is the sole and exclusive title owner of the Marks other than as set forth on the CK Companies Disclosure Schedule, free and clear of all Liens other than the Contracts listed on the CK Companies Disclosure Schedule granting third parties rights to use such Marks. All material

applications and registrations for Intellectual Property Rights included in the CK Intellectual Property Rights, including the Marks, as set forth in the CK Companies Disclosure Schedule, are as of the date hereof, valid, subsisting, in full force and effect; have not been assigned and have been properly maintained by the filing of all necessary declarations and renewals in compliance in all material respects with the CK Trust Agreement and the Servicing Agreement. Except as disclosed on the CK Companies Disclosure Schedule, the CK Trust owns all of the Marks and continues to own, preserve, maintain, defend and protect the Marks owned by the CK Trust in compliance in all material respects with the CK Trust Agreement and the Servicing Agreement. CKI is the servicer of the CK Trust and effects all maintenance of the Marks on behalf of the CK Trust, in compliance in all material respects with the Servicing Agreement. CKI continues to own, preserve, maintain, defend and protect the Marks owned by CKI as a nominee for the benefit of the CK Trust, as set forth in the CK Companies Disclosure Schedule. Except as disclosed in the CK Companies Disclosure Schedule, and to the Knowledge of the Sellers or any CK Company, there is no Intellectual Property Right necessary to conduct or material to the businesses of the CK Companies as currently conducted whether used by any Seller, any CK Company or any Person authorized by any of the foregoing that is not owned, or beneficially owned, by the CK Trust or a CK Company or that the CK Trust or the CK Companies are not properly authorized to use. The CK Companies and/or the CK Trust have taken all necessary actions to maintain and protect in all material respects the Marks and each other material item of CK Intellectual Property Rights in the ordinary course of business.

- (d) The continued operation of the business of the CK Companies as presently conducted does not infringe, misappropriate or make unauthorized use of, in any material respect, any Intellectual Property Rights of third parties. Except as disclosed in the CK Companies Disclosure Schedule, there are no material Actions pending or threatened in writing that alleges that the use or exploitation of the CK Intellectual Property Rights infringes, misappropriates or constitutes the unauthorized use of any Intellectual Property Rights of third parties. To the Knowledge of the Sellers or any CK Company, and except as disclosed in the CK Companies Disclosure Schedule, the CK Intellectual Property Rights are not being infringed by any Person. Except as disclosed in the CK Companies Disclosure Schedule, there are no material Actions pending for which notice has been provided to any Seller, any CK Company or the CK Trust, or Actions threatened in writing, challenging any CK Company's or the CK Trust's ownership of, right to use, or the validity or enforceability or patentability of any CK Intellectual Property Rights.
- (e) None of the CK Companies or the CK Trust is in breach of or default in any material respect, or is alleged in writing, to be in breach of or default, in any material respect, under any CK Intellectual Property Rights Contract, nor has an event or condition occurred (or been alleged by any other party in writing to have occurred) which would constitute a material breach

or event of default on the part of any CK Company or the CK Trust or would provide a basis for a valid claim, acceleration, additional fees or termination by any other party under any CK Intellectual Property Rights Contract. Except as disclosed in the CK Companies Disclosure Schedule, to the Knowledge of Sellers or any CK Company, no other party is in breach of or default, in any material respect, under any CK Intellectual Property Rights Contract, nor, to the Knowledge of the Sellers or any CK Company, has any event or condition occurred (or been alleged by any other party in writing to have occurred) which would constitute a material breach or event of default on the part of such other party under any such Contract. Except as disclosed in the CK Companies Disclosure Schedule, no material waiver or deferral of enforcement of the rights or benefits of any CK Company has been provided under any CK Intellectual Property Rights Contract since January 1, 2000. Except as disclosed in the CK Companies Disclosure Schedule, the consummation of the transactions contemplated hereby and the Related Agreements shall not result in the loss or impairment of any CK Company's or the CK Trust's rights in the CK Intellectual Property Rights.

- (f) None of the CK Companies or the CK Trust has given to any Person an indemnity in connection with any Intellectual Property Right, other than indemnities that, individually or in the aggregate, would not reasonably be expected to result in liability to the CK Companies or the CK Trust in excess of \$100,000, except as otherwise disclosed in the CK Companies Disclosure Schedule.
- 3.13 Insurance. The CK Companies Disclosure Schedule contains a complete and correct list, as of the date hereof (together with their respective termination dates), of all policies of fire, casualty, general liability, defamation, general media liability, personal injury, property damage, workers' compensation and all other forms of insurance carried by each CK Company or pursuant to which any CK Company is a named beneficiary or pursuant to which the business or properties of each CK Company is insured and complete and correct copies of which have been provided to PVH. All of such policies and any substantially equivalent replacement coverages are in full force and effect and no notice of cancellation or termination has been received with respect to such policies. Each CK Company has notified such insurers of any material claim arising since January 1, 2000 known to it which it believes is covered by any such insurance policy and has provided PVH with a list of such claim which is pending as of the date hereof.

3.14 Financial Statements; Liabilities.

(a) CKI has provided to PVH the audited combined balance sheets of CKI and its affiliates as of, and the audited combined statements of operations and cash flows of CKI and its affiliates for, the fiscal years ended December 29, 2001, December 30, 2000 and January 1, 2000, together with the notes thereto and the opinions of PricewaterhouseCoopers LLP thereon (collectively, the "Audited Financial Statements"). Except as set forth in the CK Companies Disclosure Schedule, the Audited Financial Statements have been prepared from the books and records of CKI and its affiliates, and present fairly in all material respects, in conformity with GAAP, the financial condition, results of operations and cash flows of CKI and its affiliates on a combined basis for the periods and dates covered

thereby. Additionally, CKI has provided to PVH the unaudited combined balance sheets of CKI and its affiliates as of, and the unaudited combined statements of operations and cash flows of CKI and its affiliates for, the nine-month period ended September 28, 2002 (collectively, the "Unaudited Financial Statements"). Except as set forth in the CK Companies Disclosure Schedule, the Unaudited Financial Statements have been prepared from the books and records of CKI and its affiliates, and present fairly in all material respects, in conformity with GAAP, the financial condition, results of operations and cash flows of CKI and its affiliates on a combined basis for such period and date covered thereby, except for adjustments and accruals which are normally made at year end and the absence of footnote disclosures thereto.

- (b) Except as set forth in the CK Companies Disclosure Schedule (which, notwithstanding Section 11.11, with respect to clause (iii) shall mean any section of the CK Companies Disclosure Schedule), CKI and its affiliates do not have any liabilities or obligations of any kind, whether absolute, accrued, asserted or unasserted, contingent or otherwise, except liabilities, obligations and contingencies, that (i) are reflected on or accrued or reserved against in the 2001 Balance Sheet, or reflected in any notes thereto, or the September 28 Balance Sheet, or reflected in any notes thereto, (ii) were incurred since September 28, 2002 in the ordinary course of business (iii) were incurred since September 28, 2002 not in the ordinary course of business but only if reflected on, or accrued or reserved against, the Closing Date Balance Sheet, or (iv) with respect to liabilities and obligations of a type not required to be disclosed by GAAP in a combined balance sheet of the CK Companies, would not reasonably be expected to have a CK Companies Material Adverse Effect.
- 3.15 Tax Matters. Except as set forth in the CK Companies Disclosure Schedule:
 - (i) For all periods beginning after December 31, 1994, each of the CK Companies is not and has not at any time during the period for which any Tax adjustment could be made been a member of any consolidated, combined, unitary or similar group with respect to any Taxes. Each of the CK Companies currently qualifies as an S corporation for federal income Tax purposes and has so qualified since the applicable date stated on the CK Companies Disclosure Schedule. The states in which each of the CK Companies qualifies as an S corporation and the date from which it so qualified are stated on the CK Companies Disclosure Schedule. Each of the CK Companies qualifies as an S corporation in every jurisdiction in which it files income or franchise Tax Returns that permits corporations to elect to qualify as S corporations.
 - (ii) All Tax Returns required to be filed by, or with respect to, each of the CK Companies have been filed when due, all such Tax Returns were correct and complete in all material respects, and all Taxes shown on said Tax Returns have been paid or, to the extent not paid, accrued in the September 28 Balance Sheet. The Sellers have made available to PVH complete and correct copies of (i) all income and franchise Tax Returns filed by each of the CK Companies for the last three taxable years, and (ii) all Tax examination reports and statements of deficiencies assessed with respect to each of the CK Companies for the last five taxable years.
 - (iii) No Tax proceeding is currently being conducted with respect to any of the CK Companies, no issues that had been raised by a Tax Authority with respect to any of them are pending and none of them has received written notification from any Tax Authority that it intends to commence a Tax proceeding with respect to any Tax Return.
 - (iv) There are no outstanding waivers of the time for assessment of any Taxes relating to any of the CK Companies. There are no Liens for any Tax (other than Taxes not yet due and payable) on the assets of any of the CK Companies.

- (v) To the Knowledge of the Sellers or any CK Company, no claim has been made by any Tax Authority in a jurisdiction where Tax Returns are not filed, that any of the CK Companies is or may be subject to taxation by that jurisdiction.
- (vi) None of the CK Companies will be required to recognize taxable income on or after the Closing Date in connection with a transaction with respect to which money or other property which is the subject of such taxable income was received and distributed to the Sellers prior to the Closing Date.
- (vii) There is not, and will not be as of the Closing, any agreement or consent made under Section 341(f) of the Code affecting any of the CK Companies.
- (viii) None of the CK Companies has assets that constitute tax-exempt bond financed property or tax-exempt use property within the meaning of Section 168 of the Code, and none of such assets is subject to a lease, safe-harbor lease, or other arrangement as a result of which the purported owner is not treated as the owner for federal income Tax purposes.
- (ix) None of the CK Companies is a party to any Tax sharing or Tax indemnification agreement or arrangement, whether formal or informal, and from and after the Closing Date, none of them shall have any rights, obligations or liabilities under any such agreement or arrangement in existence prior to the Closing.
- (x) None of the CK Companies has undergone a change in accounting method that currently requires, or will require, an adjustment to taxable income under Section 481 of the Code for any period following the Closing Date.
- (xi) All material Taxes that any of the CK Companies is or was obligated to withhold from amounts paid prior to the date hereof to any Person have been fully paid or accrued on the CK Companies' financial statements.
- (xii) None of the CK Companies is or has been a "United States real property holding company" within the meaning of the Code during the applicable period specified in Section 897(c) (1) (A) (ii) of the Code.
- (xiii) The CK Trust has been classified and treated as a grantor trust for federal income tax purposes at all times since inception.
- (xiv) For purposes of the representations set forth in (ii) through (vi) and (viii) through (xi) of this Section 3.15, the term "CK Companies" shall include the CK Trust.
- 3.16 Absence of Certain Changes and Events. Since September 28, 2002, except as set forth in the CK Companies Disclosure Schedule, each CK Company has conducted its business in the ordinary course thereof consistent with past practice and from such date through the date of this Agreement, with respect to any CK Company or the CK Trust, as the case may be, there has not been any:

- (i) change in the business, assets, liabilities, results of operations or financial condition of the CK Companies, or any event, condition or contingency (either individually or taken together) that constitutes a CK Companies Material Adverse Effect;
- (ii) (A) incurrence, payment or discharge of any liability or obligation (absolute, accrued, contingent or otherwise), (B) sale or transfer of any property, or (C) acquisition or sale, lease, grant of interest in, or other disposition of, any assets or businesses, in each of clauses (A), (B) and (C), other than in the ordinary course of business, consistent with past practice;
- (iii) (A) guarantee or any other assumption of the obligations of any Person (other than another CK Company) or (B) making of any loan or advance to any Person (other than business-related advances to employees in the ordinary course of business, consistent with past practice and in an amount not in excess of \$25,000 per employee or \$100,000 in the aggregate);
- (iv) settlement or compromise of any Action if the amount of such settlement will not be paid in full prior to the Closing or which settlement or compromise would reasonably be expected to have a continuing adverse impact on the business of the CK Companies after the Closing;
- (v) Tax election or change in a Tax election or the filing for any change in any material respect of any method of accounting with the Internal Revenue Service, except as required by any change in Law;
- (vi) change in any method of accounting applied in the preparation of the Financial Statements, other than a change which is required by reason of a concurrent change in Law or GAAP;
- (vii) (A) adoption of or amendment in any material respect to any benefit plan or bonus, profit sharing, deferred compensation, incentive, stock option or stock purchase plan, program or commitment, paid time off for sickness or other plan, program or arrangement for the benefit of its employees, consultants or directors, or (B) grant of any increase (other than increases required under any Contract and annual increases in the ordinary course of business, consistent with past practice) in the compensation of its employees, officers or directors (including any such increase pursuant to any bonus, profit sharing or other compensation or incentive plan, program or commitment);
- (viii) material change or modification in any of the Contracts, nor has any of the CK Companies or the CK Trust entered into any Contract, except, in each case, in the ordinary and regular course of its business and in no event calling for annual payments by, or to, any CK Company in excess of \$250,000 (other than with respect to Permitted Arrangements);
- (ix) issuance or sale by any CK Company or the CK Trust of any capital stock or trust interest of any CK Company or the CK Trust, as the case may be, or any security convertible into or exchangeable for, any shares of such capital stock or trust interests;

- (x) declaration, distribution or the setting aside for distribution of any property other than cash, cash equivalents and readily marketable securities of any CK Company or the CK Trust, or directly or indirectly, the redemption, purchase or otherwise acquisition of any shares of capital stock or trust interests for property other than cash, as the case may be; and
- (xi) agreement, whether in writing or otherwise, to take any action described in this Section 3.16.
- 3.17 Litigation and Claims. Except as set forth in the CK Companies Disclosure Schedule, there is no material Action pending or, to the Knowledge of the Sellers or any CK Company, threatened, against or affecting any CK Company, the CK Trust or the Purchase Shares, and there is no material Action pending or, to the Knowledge of the Sellers or any CK Company, threatened, against any CK Company or the CK Trust affecting the propriety or validity of the transactions contemplated hereby or by the Related Agreements. Except as set forth in the CK Companies Disclosure Schedule, none of the CK Companies or the CK Trust is subject to or in default under or with respect to any Order.
 - 3.18 Governmental Permits; Compliance with Laws.
- (a) Each CK Company owns, holds or possesses all material Governmental Permits which are necessary to entitle it to own or lease, operate and use its assets and to carry on and conduct its business substantially as currently conducted, except for such Governmental Permits which would be obtainable in due course by any qualified applicant without any undue burden or cost in the event of any failure to apply, lapse, termination, cancellation or forfeiture thereof.
- (b) The CK Companies Disclosure Schedule sets forth a complete and correct list and brief description of each Governmental Permit owned, held or possessed by the CK Companies, as of the date hereof, except for such Governmental Permits which would be obtainable in due course by any qualified applicant without any undue burden in the event of any lapse, termination, cancellation or forfeiture thereof. Except as set forth in the CK Companies Disclosure Schedule, (i) each CK Company has fulfilled and performed in all respects its obligations under each of the Governmental Permits which it owns, holds or possesses other than where the failure to perform could not reasonably be expected to have a CK Companies Material Adverse Effect, and (ii) no written notice of cancellation, of default or of any material dispute concerning any Governmental Permit, or of any event, condition or state of facts described in the preceding clause, has been received by the Sellers or any CK Company. Solely for purposes of Section 7.1, the term "Governmental Permit" used in this Section 3.18(b) shall also include any Governmental Permit owned, held or possessed by any CK Company after the date of this Agreement that would have been required to be listed in a Schedule to this Agreement had such Governmental Permit been owned, held or possessed by such CK Company as of the date of this Agreement.
- (c) Each CK Company is in compliance in all respects with all Laws which are applicable to its business other than any failure to comply which could not reasonably be expected to have a CK Companies Material Adverse Effect.

- 3.19 Environmental Matters. Except as set forth in the CK Companies Disclosure Schedule:
 - (i) Each CK Company is in compliance in all material respects with its obligations under applicable Environmental Laws;
 - (ii) There are no Environmental Conditions present at, on, or under, any Facility, as a result of activities of any CK Company or any of their employees or agents or, to the Knowledge of the Sellers or any CK Company, as a result of activities of any other Person, in each case in amounts exceeding the levels permitted by applicable Environmental Law or under circumstances that would reasonably be expected to result in liability in any material respect under or relating to Environmental Law;
 - (iii) None of the CK Companies has disposed of, arranged for the disposal of, released, threatened to release, or transported any Hazardous Substances in violation of any applicable Environmental Law or in a manner that would reasonably be expected to result in material Damages to PVH or to any CK Company;
 - (iv) None of the CK Companies is subject to any Actions, is subject to any Order or has received any notice or other communication from any Governmental Authority or the current or prior owner or operator of any of the Facilities or any other Person, in each case with respect to any actual or potential violation or failure to comply with any Environmental Law or of any actual or threatened obligation or liability under any Environmental Law, or regarding any Hazardous Substances; and to the Knowledge of the Sellers or any CK Company, none of the CK Companies is threatened with any such written Action, Order, notice or communication;
 - (v) None of the CK Companies has specifically contractually assumed any liability or obligation under or relating to Environmental Laws or Hazardous Substances; and
 - (vi) Since January 1, 1992, there are no Environmental Reports in the custody or control of the CK Companies relating to the Facilities, business of any CK Company or activities of any CK Company that have not been made available to PVH.

3.20 Employee Plans.

(a) Except as set forth in the CK Companies Disclosure Schedule, none of the CK Companies or the Sellers nor any other Person which together with the CK Companies or any of the Sellers constitutes a member of the CK Companies' or such Seller's "controlled group" or "affiliated service group" (within the meaning of Sections 4001(a)(14) and/or (b) of ERISA and/or Sections 414(b), (c), (m) or (o) of the Code (each such group or groups and each member thereof hereinafter referred to individually and collectively as the "Group")) has at any time adopted or maintained, has any liability or is a fiduciary with respect to or has any present or future obligation to contribute to or make payment under (i) any employee benefit plan (as defined in Section 3(3) of ERISA) which is subject to ERISA, or (ii) any other benefit plan, program, contract or arrangement of any kind whatsoever (whether for the benefit of present,

former, retired or future employees, consultants or independent contractors of the Group, or for the benefit of any other Person or Persons) including, without limitation, plans, programs, contracts or arrangements with respect to pension, retirement, profit sharing, deferred compensation, thrift, savings, stock ownership, stock bonus, restricted stock, health, dental, medical, life, hospitalization, disability, relocation, child care, educational assistance, stock purchase, stock option, incentive, bonus, sabbatical leave, vacation, severance, cafeteria, pre-tax premium, flexible spending or other contribution, benefit or payment of any kind, and plans, programs, contracts or arrangements providing for contributions, benefits or payments in the event of a change of ownership or control in whole or in part of the Group, other than any such benefit plan, program, contract or arrangement maintained outside the United States primarily for the benefit of International Employees (hereinafter individually and collective called the "International Plans"), (all such employee benefit plans and other benefit plans, programs, contracts or arrangements, whether written or oral, other than any International Plan, hereinafter individually and collectively called the "Employee Benefit Plans"). No member of the Group has any obligation other than as required by applicable Law, to amend any Employee Benefit Plan so as to increase benefits thereunder or otherwise.

- (b) No Employee Benefit Plan is subject to Title IV of ERISA, Section 302 of ERISA or Section 412 or 413(c) of the Code, other than the ILGWU National Retirement Fund and NRF-2000.
- (c) Except as set forth in the CK Companies Disclosure Schedule with respect to the filings of Forms 5330, any and all amounts which any member of the Group is required to pay as contributions or otherwise, with respect to the Employee Benefit Plans, have been timely paid.
- (d) Except as disclosed on the CK Companies Disclosure Schedule, each of the Employee Benefit Plans (other than any Multiemployer Plans (as defined in Section 3.20(e)) and any welfare plans (within the meaning of Section 3(1) of ERISA) maintained by a labor union (such welfare plans and Multiemployer Plans that are Employee Benefit Plans hereinafter individually and collectively called "Union Plans")) sponsored or maintained by any CK Company, to which such CK Company has any current or future obligation to contribute or under which any employee of such CK Company (and/or any dependent or beneficiary of such employee) is covered or entitled to benefits by reason of employment with such CK Company (such Employee Benefit Plans hereinafter individually and collectively called "Company Plans") and, to the Knowledge of the Sellers or any CK Company, each of the Union Plans, has been established, maintained, operated and administered in all material respects in accordance with its terms and all applicable Law. Each of the Company Plans which is intended to be "qualified" within the meaning of Sections 401(a) and 501(a) of the Code (a "Qualified Plan") has been determined by the Internal Revenue Service to be so qualified and nothing to the Knowledge of the Sellers or any CK Company has occurred since the issuance of such determination to adversely affect such qualified status. There are no pending, threatened or anticipated proceedings (other than routine claims for benefits) involving any of the Company Plans, or, to the Knowledge of the Sellers or any CK Company, any of the Union Plans, with respect to or affecting any CK Company or any current or former employee of any CK Company. There have been no nonexempt "prohibited transactions" within the meaning of Section 406 of ERISA or Section 4975 of the Code with respect to any of the Employee Benefit Plans with respect to

which all obligations and liabilities of any CK Company has not been satisfied in full prior to November 1, 2002.

- (e) No member of the Group has incurred any withdrawal liability with respect to a "multiemployer plan" within the meaning of Section 3(37) of ERISA (each such plan hereinafter a "Multiemployer Plan") under Title IV of ERISA. No member of the Group is a party to, or participates in, or has any liability or contingent liability with respect to any Multiemployer Plan (other than the ILGWU National Retirement Fund and NRF-2000 as disclosed in Section 3.20(e) of the CK Companies Disclosure Schedule).
- (f) A complete and correct copy of each of the Company Plans and each of the International Plans, and all amendments thereto, whether currently effective or to become effective at a later date, and all contracts and agreements relating thereto, or to the funding thereof (including, without limitation, all trust agreements, insurance contracts, investment management agreements, subscription and participation agreements, administration and recordkeeping agreements) have been provided to PVH. In the case of any Company Plan that is not in written form, an accurate and complete description of such Company Plan has been provided to PVH. With respect to each Company Plan, PVH has been provided with a complete and correct copy of each of (i) the three most recent annual reports (Form 5500 series), and (ii) the most recent summary plan description (including summaries of material modification), and Internal Revenue Service determination letter and/or ruling.
- (g) Except as set forth on the CK Companies Disclosure Schedule, each Company Plan, and all contracts and agreements relating thereto or to the funding thereof, can be unilaterally terminated by such CK Company, without penalty, on no more than 30 days' notice, and all obligations of such CK Company with respect to all other Employee Benefit Plans can be unilaterally terminated by such CK Company, without penalty, on no more than 30 days' notice. There have been no material changes in the financial condition of the respective Company Plans (or other information provided hereunder) from that stated in each Company Plan's most recent of such annual reports.
- (h) Except as set forth on the CK Companies Disclosure Schedule, no Company Plan provides benefits including, without limitation, death or medical benefits (whether or not insured), with respect to any employees, former employees or directors of any CK Company beyond their retirement or other termination of service, other than (i) coverage mandated by applicable Law or (ii) death benefits or retirement benefits under any "employee pension plan," as that term is defined in Section 3(2) of ERISA.
- (i) Except as set forth in the CK Companies Disclosure Schedules, neither the execution and delivery by the Sellers or any CK Company of this Agreement or the Related Agreements nor the consummation by the Sellers of the transactions contemplated hereby or by the Related Agreements shall (either alone or upon the occurrence of additional events or acts) (x) require any CK Company to make any payment to, or obtain any consent or waiver from, any officer, director, employee, consultant or agent of any member of the Group or (y) accelerate vesting or payment of any benefits or any payments, increase the amount or value of any benefit or payment.

- (j) The CK Companies Disclosure Schedule separately identifies each International Plan as well as each severance or other post termination arrangement applicable to International Employees. Any and all amounts that any member of the Group is required to pay as contribution or otherwise with respect to each International Plan have been timely paid. Each International Plan has been established, maintained and operated in all material respects in accordance with their terms and all applicable Law including, without limitation, any Law relating to exemption from Tax for any International Plan intended to be Tax-exempt. There are no pending, threatened or anticipated proceedings (other than routine claims for benefits) involving any of the International Plans.
- 3.21 No Finder. None of the CK Companies or any party acting on their behalf, has paid or become obligated to pay any fee or commission to any broker, finder or intermediary (other than counsel which fee or commission shall be paid in full by the CK Companies prior to the Closing) for or on account of the transactions contemplated hereby or by the Related Agreements.

ARTICLE IV.

REPRESENTATIONS AND WARRANTIES OF PVH

PVH represents and warrants to each CK Company and the Sellers as follows:

- 4.1 Organization; Good Standing. PVH is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware.
- 4.2 Authority; Execution and Delivery; Enforceability. PVH has full corporate power and authority to execute and deliver this Agreement and the Related Agreements, to perform its respective obligations hereunder and under the Related Agreements and to consummate the transactions contemplated hereby and by the Related Agreements. All corporate acts and other proceedings required to be taken by PVH to authorize the execution, delivery and performance of this Agreement and the Related Agreements have been duly and properly taken. Each of this Agreement and (when executed) the Related Agreements has been (or will be) duly executed and delivered by PVH, and constitutes (or will, when executed, constitute) the legal, valid and binding obligation of PVH, enforceable against PVH in accordance with its terms, subject to bankruptcy, insolvency, fraudulent transfer, moratorium and other similar Laws of general applicability relating to or affecting creditors' rights and to general equity principles.
- 4.3 Non-Contravention. The execution and delivery of this Agreement and the Related Agreements by PVH does not, and the consummation of the transactions contemplated hereby and by the Related Agreements and compliance with the terms hereof and of the Related Agreements, will not:
 - (i) constitute a violation or breach of the certificate of incorporation or the by-laws of PVH;

- (ii) except as set forth in the PVH Disclosure Schedule, constitute a default under or a violation or breach of, or result in the acceleration of any obligation under, any provision of any material contract or other instrument to which PVH is a party or by which any of the assets of PVH; or
- (iii) assuming the consents described in Section 4.4 have been received, violate any Order or any Law affecting PVH or the assets of PVH.
- 4.4 Consents and Approvals. Except as set forth in the PVH Disclosure Schedule, no consent, approval, license, permit, order or authorization of, or registration, declaration or filing with, any Governmental Authority or Person is required to be obtained by or on behalf of PVH in connection with the execution, delivery and performance of this Agreement or the Related Agreements or the consummation of the transactions contemplated hereby and by the Related Agreements, other than compliance with and filings under the HSR Act and similar Laws of any jurisdiction.
- 4.5 Equity Matters. The PVH Shares, the Warrant and the shares of PVH Common Stock underlying the Warrant, when issued in accordance with the terms of this Agreement or the Warrant, as the case may be, will be duly and validly issued, fully paid and non-assessable, will be free and clear of any Liens imposed by or through PVH, will not be subject to preemptive rights and will not subject the holder thereof to personal liability by reasons of being such a holder.
- 4.6 Litigation and Claims. There is no Action pending or, to the knowledge of PVH, threatened, against or affecting PVH with respect to the propriety or validity of the transactions contemplated hereby or by the Related Agreements. Except as otherwise disclosed herein or in the PVH Disclosure Schedules, there are no Actions pending or, to the knowledge of PVH, threatened, against or involving PVH or any of its Subsidiaries or any of their respective properties or assets, at Law or in equity, involving claims of more than \$1,000,000 by stockholders of PVH, or claims involving more than \$1,000,000 for product liability, infringement of trademark, patent or intellectual property, violations of health and safety laws covering employees, violations of Environmental Laws, violations of customs laws, sexual harassment or discrimination, or racial discrimination. There is no outstanding or, to the knowledge of PVH, threatened, Order of any Governmental Authority against PVH or any of its Subsidiaries or any of their respective properties or assets, which Order could have a PVH Material Adverse Effect.
- 4.7 SEC Reports. PVH has timely filed all forms, reports and documents required to be filed by it with the Securities and Exchange Commission since February 4, 2001 (the "SEC Reports"). The SEC Reports (i) were prepared in all material respects in accordance with the requirements of the Securities Act or the Exchange Act, as the case may be, and (ii) did not at the time they were filed contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements made therein, in light of the circumstances under which they were made, not misleading. Each of the balance sheets (including the related notes) included in the SEC Reports presents fairly in all material respects the consolidated financial position of PVH as of the respective dates thereof, and the

other related financial statements (including the related notes) included therein presented fairly in all material respects the consolidated results of operations and changes in financial position of PVH for the respective periods indicated, except, in the case of interim financial statements, for year-end audit adjustments, consisting only of normal recurring accruals. Each of the financial statements (including the related notes) included in the SEC Reports has been prepared in accordance with GAAP, except as otherwise noted therein or, in the case of the unaudited financial statements, as permitted by the applicable rules and regulations of the Securities and Exchange Commission.

- 4.8 Private Offering. Subject to the accuracy of the Sellers' representations and warranties contained in Article II, the offer, sale and issuance of the PVH Shares, the Warrant and the shares of PVH Common Stock underlying the Warrant, as contemplated hereby and by the Warrant, is (and will be) exempt from the registration requirements of the Securities Act. PVH has not offered or sold such securities by any form of general solicitation or general advertising, as such terms are used in Rule 502(c) under the Securities Act.
- 4.9 Capitalization. The PVH Disclosure Schedule sets forth, in each case as of the date hereof, (i) the authorized Capital Stock of PVH, the number of shares of each class of Capital Stock issued and outstanding and the number of shares of PVH Common Stock reserved for issuance in connection with PVH's stock option plans, and (ii) all options, warrants, rights to subscribe to, calls, contracts, undertakings, arrangements and commitments to issue which may result in the issuance of Capital Stock of PVH, other than (x) options to purchase 615,887, 1,886,878 and 2,387,564 shares of PVH Common Stock issued and outstanding under PVH's 1987, 1997 and 2000 stock option plans, respectively, and (y) the Rights. All of the issued and outstanding shares of PVH's Capital Stock have been duly and validly authorized and issued and are fully paid and non-assessable and are not subject to any preemptive rights. Except pursuant to this Agreement or in connection with stock option plans, the Rights Agreement and the Apax Transaction, (i) no equity securities of PVH are or may be required to be issued by reason of any options, warrants, rights to subscribe to, calls or commitments of any character whatsoever, (ii) there are outstanding no securities or rights convertible into or exchangeable for shares of any Capital Stock of PVH, and (iii) there are no contracts, commitments, understandings or arrangements by which PVH is bound to issue additional shares of its Capital Stock or securities or rights convertible into or exchangeable for shares of any Capital Stock of PVH, or options, warrants or rights to purchase or acquire any additional shares of its Capital Stock. Neither PVH nor any of its Subsidiaries are subject to any obligation (contingent or otherwise) to repurchase or otherwise acquire or retire any of its Capital Stock. Except as contemplated by the Registration Rights Agreement, there are no contracts between PVH and any Person granting such Person the right to require PVH to file a registration statement under the Securities Act with respect to any securities of PVH owned or to be owned by such Person or to require PVH to include such securities in any other registration statement filed by PVH under the Securities Act.
- 4.10 Listing. PVH is not in violation of the listing requirements of the NYSE in any material respect. PVH has not received any written notice from the NYSE that the PVH Common Stock is to be delisted by the NYSE.

4.11 Financial Statements. PVH has made available to the Sellers copies of (i) the consolidated balance sheet of PVH and its Subsidiaries as of February 3, 2002 and February 4, 2001 and the related consolidated income statements, changes in stockholders' equity and cash flows for the fiscal years ended February 3, 2002, February 4, 2001 and January 30, 2000, as reported in PVH's Annual Report on Form 10-K for the fiscal year ended February 3, 2002, filed by PVH with the SEC under the Exchange Act, and accompanied by the audit report of Ernst & Young LLP, independent public accountants, and (ii) the unaudited consolidated balance sheet of PVH and its Subsidiaries as of November 3, 2002 (the "PVH Balance Sheet") and the related unaudited consolidated income statements and cash flows for the thirteen weeks and thirty-nine weeks ended November 3, 2002, as reported in PVH's Quarterly Report on Form 10-Q for the quarterly period ended November 3, 2002, filed with the SEC under the Exchange Act. Such audited financial statements accurately reflect the books and records of PVH and present fairly, in all material respects, the consolidated financial position of PVH and its Subsidiaries and the consolidated results of their operations and their cash flows for the periods and dates covered thereby, in conformity with GAAP. Such unaudited financial statements accurately reflect the books and records of PVH and present fairly, in all material respects, the consolidated financial position of PVH and its Subsidiaries and the consolidated results of their operations and their cash flows for the period and date covered thereby, in conformity with GAAP, except for changes resulting from year-end adjustments (none of which will be material in amount) and the absence of footnote disclosures thereto. PVH and its Subsidiaries have no liabilities or obligations of a type that GAAP would require to be on the PVH Balance Sheet (absolute, accrued, contingent or otherwise) which are not fully reflected or reserved against in the PVH Balance Sheet in conformity with GAAP, except for liabilities and obligations that may have arisen in the ordinary and usual course of business and consistent with past practice since November 3, 2002 and that, individually or in the aggregate, could not reasonably have a PVH Material Adverse Effect. Neither PVH nor any of its Subsidiaries are currently a party to any Off-Balance Sheet Arrangement, which is not reflected in the financial statements (or the footnotes thereto) referred to in this Section 4.11. During the past three years, PVH has not restated any of its published financial results and PVH is not aware of any facts which may require such restatement.

4.12 Tax Matters.

- (a) For the periods commencing after January 1, 1999, PVH and each of its Subsidiaries has timely filed or caused to be timely filed any and all Tax Returns required to be filed by it under applicable Law, except to the extent that any failure to do so could not reasonably be likely to have a PVH Material Adverse Effect. The reserves for Taxes contained in the financial statements of PVH or carried on the books and records of PVH and its Subsidiaries, as applicable, are in the aggregate adequate to cover all Tax liabilities and deferred Taxes of PVH and its Subsidiaries, as of the date of this Agreement, except to the extent that any inadequacy could not have a PVH Material Adverse Effect.
- (b) For the periods commencing after January 1, 1999, all Taxes shown as being due and owing by PVH or any of its Subsidiaries on any Tax Return have been paid.

- (c) PVH and each of its Subsidiaries have timely withheld and paid all Taxes required to have been withheld and paid in connection with any amounts paid or owing to any employee, independent contractor, creditor, stockholder, or third party, except to the extent that any failure to do so could not have a PVH Material Adverse Effect.
- (d) None of PVH nor any of its Subsidiaries have waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency, in each case except as to Taxes that are disclosed in the PVH Disclosure Schedules or in the financial statements of PVH and its Subsidiaries, as applicable, or that, if assessed could not have a PVH Material Adverse Effect.
- (e) PVH has not been a "United States real property holding corporation" within the meaning of Section 897 of the Code during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code.

4.13 Intellectual Property.

- (a) For the purposes of this Section 4.13, "PVH Intellectual Property" shall mean (i) all PVH Marks, trade dress, logos, trade names, domain names, web-sites, brand names and corporate names (and all licenses or other rights relating to any of the foregoing), together with all goodwill associated therewith, and all applications, registrations and renewals in connection therewith, (ii) all inventions and designs (whether patentable or unpatentable and whether or not reduced to practice), all improvements thereto, and all patents, patent applications, and patent disclosures, together with all reissuances, continuations, continuations-in-part, revisions, extensions and reexaminations thereof, and (iii) all copyrightable works, all copyrights and all applications, registrations and renewals in connection therewith, in each case which is owned by or licensed to PVH and its Subsidiaries as of the date hereof.
- (b) Each of PVH and its Subsidiaries owns, or is licensed to use, all PVH Intellectual Property material to its business, and to the knowledge of PVH, the use thereof by PVH and its Subsidiaries does not infringe upon the trademark, copyright or proprietary rights of any other Person, except for any such infringements that, individually or in the aggregate, could not reasonably be likely to have a PVH Material Adverse Effect.
- (c) Except as set forth in the PVH Disclosure Schedules, no PVH Mark licensed to or used by PVH or any of its Subsidiaries is scheduled to expire (without the right of extension on the part of PVH) within 12 months from the date hereof, which such expiration could reasonably be likely to have a PVH Material Adverse Effect.

4.14 Employee Plans.

(a) The PVH Disclosure Schedule sets forth a complete and correct list of every Employee Program which is maintained, administered, sponsored or contributed to by PVH or any of its Subsidiaries, which covers any employee of PVH or any of its Subsidiaries or with respect to which an obligation of PVH or any of its Subsidiaries to make any contribution exists, other than a Foreign Plan. Any Employee Program listed in such schedule (other than any

Employee Program that is a multiemployer plan, as defined in Section 3(37) of ERISA (a "Multiemployer Plan")) is referred to below as a "Company Employee Plan."

- (b) PVH has made available to the Sellers with respect to each Company Employee Plan complete and correct copies of (i) all written documents comprising such Company Employee Plan (including amendments, individual agreements, service agreements, trusts and other funding agreements), (ii) the three most recent annual returns in the Federal Form 5500 series (including all schedules thereto) filed with respect to such Company Employee Plan, (iii) the three most recent audited financial statements and actuarial reports, if any, pertaining to such Company Employee Plan, (iv) the summary plan description currently in effect and all material modifications thereto, if any, for such Company Employee Plan, (v) any employee handbook which includes a description of such Company Employee Plan, (vi) the most recent Internal Revenue Service determination letter, if any, for such Company Employee Plan, and (vii) any other written communications to any employee, to the extent that the provisions of such Company Employee Plan described therein differ materially from such provisions as set forth or described in the other information or materials furnished under this subsection (b).
- (c) Each Company Employee Plan which is intended to qualify under Section 401(a) of the Code has received a determination letter from the Internal Revenue Service which states that such plan is so qualified, and on which any employer which has adopted such plan may currently rely.
- (d) Each Company Employee Plan and Foreign Plan has been maintained in accordance with its terms and with all applicable Laws, except to such extent as could not reasonably have a PVH Material Adverse Effect. Neither PVH nor any of its Subsidiaries has any unsatisfied liability, or any unpaid fine, penalty or tax, with respect to any Company Employee Plan, any Foreign Plan or any other Employee Program, which could reasonably have a PVH Material Adverse Effect. To the knowledge of PVH, there has been no "prohibited transaction" (within the meaning of Section 406 of ERISA or Section 4975 of the Code), or any breach of any duty under ERISA, any other applicable Law or any agreement, with respect to any Company Employee Plan or Foreign Plan which could subject PVH or any of its Subsidiaries to material liability either directly or indirectly (including, without limitation, through any obligation of indemnification or contribution) for any damages, penalties, taxes or any other loss or expense which could reasonably have a PVH Material Adverse Effect. PVH and each of its Subsidiaries has made full and timely payment of all contributions required to be made by it to each Company Employee Plan and Foreign Plan by the terms of any such plan or under the applicable Law, except that all contributions which are so required to be made by PVH or any of its Subsidiaries to each Company Employee Plan and Foreign Plan for any period ending prior to the Closing, but which are not due by the date of the Closing, shall be properly reserved or accrued in the appropriate financial statements. There have been no material violations of any reporting or disclosure requirements under ERISA or the Code with respect to any Company Employee Plan, including any requirement to file an annual return.
- (e) No litigation or claim (other than routine claims for benefits), and no governmental administrative proceeding, audit or investigation, is pending or, to the knowledge

of PVH or any of its Subsidiaries, threatened with respect to any Company Employee Plan or Foreign Plan which could reasonably have a PVH Material Adverse Effect.

- (f) With respect to each Company Employee Plan which is a defined benefit pension plan, (i) such Company Employee Plan has not incurred an "accumulated funding deficiency", within the meaning of Section 412 of the Code or Section 302 of ERISA, whether or not waived, (ii) such Company Employee Plan has not been terminated, (iii) no notice of intent to terminate such Company Employee Plan has been issued to participants or filed with the PBGC, (iv) the PBGC has not instituted any proceeding to terminate, or to appoint a trustee or administrator of, such Company Employee Plan, and no circumstances exist that are reasonably likely to constitute grounds under Section 4042 of ERISA which may allow the PBGC to institute any such proceeding, (v) except as set forth on the PVH Disclosure Schedule, if such Company Employee Plan is intended to be a "qualified plan" within the meaning of Section 401(a) of the Code, as of the most recent valuation date for such Company Employee Plan, the present value of benefit liabilities, when computed on a termination basis using actuarial assumptions which are reasonable under the circumstances, does not exceed the value of assets, (vi) neither PVH nor any of its Subsidiaries (x) have incurred any liability to the PBGC or any other Person, or has become subject to any lien, under Title IV of ERISA in connection with such Company Employee Plan (other than PBGC premiums), or (y) has knowledge of any facts or transactions that might reasonably be anticipated to result in the imposition of any liability on, or the imposition of any lien on the assets of, PVH or any of its Subsidiaries to, or in favor of, the PBGC or any other Person under Title IV of ERISA in connection with such Company Employee Plan (other than PBGC premiums), and (vii) no reportable event as defined in Section 4043 of ERISA and the regulations issued under such section, excluding, however, such events as to which the PBGC has by regulation waived the requirement of Section 4043 of ERISA that it be notified within 30 days of the occurrence of such event, has occurred with respect to such Company Employee Plan.
- (g) Except as set forth on the PVH Disclosure Schedule, no Employee Program which has been maintained, administered or contributed to by PVH or any of its Subsidiaries, which has covered any employee of PVH or any of its Subsidiaries, or to which PVH or any of its Subsidiaries has had any obligation to make a contribution, during the six-year period ending on the date of this Agreement, is (i) a "multiemployer plan", as defined in Section 3(37) of ERISA, (ii) a "multiple employer plan", as described in Section 413(c) of the Code, (iii) a "multiple employer welfare arrangement", as defined in Section 3(40) of ERISA, (iv) a "voluntary employees' beneficiary association" within the meaning of Section 501(c)(9) of the Code, or (v) a Foreign Plan other than a plan which does not provide benefits materially greater than any benefits which may be required by applicable Law. Neither PVH nor any of its Subsidiaries has, within the past six years, incurred any withdrawal liability to a Multiemployer Plan.
- (h) Except as set forth on the PVH Disclosure Schedule, all health and medical benefit coverage, and all death benefit coverage, under each Company Employee Plan is provided solely through insurance, and no Company Employee Plan provides health or medical coverage, life insurance coverage, or coverage for any other welfare benefit to any retiree, except

for continuation coverage required by Section 4980B of the Code, Sections 601 to 608 of ERISA or any applicable State Law.

- (i) No employee of PVH or any of its Subsidiaries shall accrue or receive additional benefits, additional credit for service, accelerated vesting or accelerated rights to payment of any benefit under any Company Employee Plan, or become entitled to any severance, termination allowance or similar payments or to the forgiveness of any indebtedness, solely as a result of the execution and delivery of, or the transactions contemplated by this Agreement. Such execution and delivery, or the occurrence of such transactions, shall not result in any increase in the contributions required to be made to any Company Employee Plan. Except as set forth on the PVH Disclosure Schedule, no payment made or contemplated under any Company Employee Plan, or by PVH or any of its Subsidiaries, in connection with the transactions contemplated hereby and by the Apax Transaction, constituted, or would constitute, either (i) an "excess parachute payment" within the meaning of Section 280G of the Code or (ii) a payment which is not deductible by reason of Section 404 of the Code.
- (j) Except as set forth on the PVH Disclosure Schedule, (i) except for the adoption of a plan amendment which is needed to bring the plan documents into conformity with statutory changes enacted in recent years, neither PVH nor any of its Subsidiaries are under any obligation (express or implied) to modify any Company Employee Plan, or to establish any new Employee Program which will cover any employee of PVH or any of its Subsidiaries, (ii) PVH or its Subsidiaries have expressly reserved to itself the right to amend, modify or terminate each Company Employee Plan (and any service or funding agreement or arrangement for each Company Employee Plan), at any time without material liability or penalty to itself (other than routine expenses), and (iii) no Company Employee Plan requires PVH or any of its Subsidiaries to continue to employ or use the services of any employee.
- (k) Except as set forth on the PVH Disclosure Schedule, there has been no amendment, interpretation or announcement by PVH or any of its Subsidiaries relating to any Company Employee Plan which would materially increase the expense of maintaining such plan above the level of expense incurred with respect to that plan, as indicated in the applicable financial statements, for its most recent fiscal year.
- (1) No entity, which at any time during the six-year period ending on the date of this Agreement has been considered a single employer with PVH or any of its Subsidiaries under Section 4001(b) of ERISA or Section 414(b), (c), (m) or (o) of the Code, has any unpaid liability, fine, penalty or tax with respect to any Employee Program for which PVH or any of its Subsidiaries could be liable, and which could reasonably have a PVH Material Adverse Effect.
 - 4.15 Compliance with Laws; Permits.
- (a) PVH and its Subsidiaries are in compliance in all material respects with all material Laws and material Orders promulgated by any Governmental Authority applicable to PVH and its Subsidiaries or to the conduct of the business or operations of PVH and its Subsidiaries or the use of their properties (including any leased properties) and assets. Since January 1, 1999, neither PVH nor any of its Subsidiaries have received any written notice of

violation or alleged material violation of any such Law or Order by any Governmental Authority in any material respect that has not been resolved. Since January 1, 1999, neither PVH nor any of its Subsidiaries have received written notice that it is the subject of an investigation by any Governmental Body which could reasonably be likely to have a PVH Material Adverse Effect.

(b) To the knowledge of PVH, except as set forth in the PVH Disclosure Schedules, PVH and its Subsidiaries have all Permits necessary for the conduct of their business, except where the failure to have such Permits could not reasonably be likely to have a PVH Material Adverse Effect. PVH and its Subsidiaries have complied in all material respects with all conditions of such Permits applicable to it.

4.16 Environmental Protection.

- (a) Each of PVH and its Subsidiaries are in compliance in all material respects with all limitations, restrictions, conditions, standards, prohibitions, requirements, obligations, schedules and timetables contained in any applicable Environmental Laws, or in any plan, Order, notice or demand letter issued, entered, promulgated or approved thereunder.
- (b) Except as set forth in the PVH Disclosure Schedules, to the knowledge of PVH, as a result of activities of PVH or its Subsidiaries or any of their employees, no Hazardous Material has been incorporated in, used on, stored on or under, released from, treated on, transported to or from, or disposed of on or from, any property owned or during the period of PVH's or any of its Subsidiaries' lease(s), or, to the knowledge of PVH without any investigation, any prior period, leased by PVH or any of its Subsidiaries such that, under applicable Environmental Laws (i) any such Hazardous Material would be required to be removed, cleaned-up or remediated before the property could be altered, renovated, demolished or transferred, or (ii) the owner or lessee of the property could be subjected to liability for the removal, clean-up or remediation of such Hazardous Material, except, under circumstances that could not reasonably be likely to have a PVH Material Adverse Effect; and, except as set forth in the PVH Disclosure Schedules, PVH has not received any written notification from any Governmental Authority or other third parties relating to Hazardous Material on or affecting any property owned or leased by PVH or its Subsidiaries or relating to any potential or known liability under applicable Environmental Laws arising from the ownership or leasing of any property, which could reasonably be likely to have a PVH Material Adverse Effect.
- 4.17 No Finder. Except for the engagement of The Blackstone Group L.P., JPMorgan Chase and Lehman Brothers, neither PVH nor any party acting on its behalf has paid or become obligated to pay any fee or commission to any broker, finder or intermediary for or on account of the transactions contemplated hereby or by the Related Agreements.
- 4.18 No Reliance. PVH acknowledges that no written or oral representations or warranties have been made to it or any of its Affiliates or representatives, with regard to any of the CK Companies or otherwise with respect to the transactions contemplated by this Agreement or any Related Agreement, other than as set forth in Article II and Article III and in the Agreement and Assignment. Without limiting the foregoing, PVH acknowledges that the Sellers and the CK Companies are not making any representations and warranties with respect to the

effect on PVH after the Closing of any agreement the CK Companies may have with UNITE, Local 89-22-1 or any other union as a result of the consummation of the transactions contemplated hereby; provided, however, that nothing herein shall relieve the Sellers of any liability hereunder as a result of a breach by the CK Companies of any such agreement or the representations and warranties of the Sellers and the CK Companies with respect thereto.

ARTICLE V.

ACTION PRIOR TO THE CLOSING DATE

From and after the execution of this Agreement until the Closing Time (or earlier termination of this Agreement in accordance with Section 10.1):

5.1 Conduct of Business.

- (a) Each CK Company shall, and each Seller shall cause each CK Company to, (i) continue to conduct the business of such CK Company in the ordinary course thereof and use its commercially reasonable efforts to maintain its business in substantially the same manner as heretofore, carry on its business' practices in substantially the same manner as heretofore and keep the business' books of account, records and files in a manner consistent with past practice, (ii) use its commercially reasonable efforts to preserve the organization of such CK Company intact, to retain the services of the non-administrative and non-union employees of such CK Company and to preserve the goodwill of the licensees, suppliers and customers of such CK Company and others having business relations with such CK Company, (iii) pay and perform all of the debts, obligations and liabilities of such CK Company as and when due consistent with past practice and all Contracts in accordance in all material respects with the terms and provisions thereof; provided, however, the CK Companies shall repay in full and cancel, at or prior to the Closing Time, the CK Debt Obligations and shall cause all Liens granted under the Line Letter to be discharged, (iv) maintain all CK Intellectual Property Rights in a manner consistent with past practices and not take any action adverse to the preservation of such CK Intellectual Property Rights, (v) fully satisfy all obligations, on a timely basis, under each employee benefit plan (as defined in Section 3(3) of ERISA), including, without limitation, all contribution obligations, and to administer, operate and maintain each such employee benefit plan in accordance with its terms and all applicable Law, including with respect to any Qualified Plan, the qualification requirements of the Code, and (vi) use the same efforts as heretofore used to comply with all Laws applicable to such CK Company.
- (b) Notwithstanding Section 5.1(a) hereof, none of the CK Companies shall, and each Seller shall cause the CK Companies to not, without the prior written consent of PVH:
 - (i) make any material change in the businesses or the operations of any CK Company;
 - (ii) (A) incur, pay or discharge any liability or obligation (absolute, accrued, contingent or otherwise), except with respect to the CK Debt Obligations, (B) sell or

transfer any property, or (C) acquire or sell, lease, grant an interest in or dispose of any assets or businesses, in each case, other than in the ordinary course of business consistent with past practice or pursuant to Section 5.11;

- (iii) (A) guarantee or assume any other obligation of any Person (other than another CK Company), or (B) make any loan or advance to any Person (other than business-related advances to employees in the ordinary course of business, consistent with past practice and in an amount not to exceed \$25,000 per employee or \$100,000 in the aggregate);
- (iv) waive any right of value owed to, cancel any debt owed to, or claims held by, any CK Company, except in the ordinary course of business consistent with past practice, or with respect to the CK Debt Obligations;
- (v) settle or compromise any Action, which amount of such settlement or compromise is not paid in full prior to the Closing or which settlement or compromise would have a continuing adverse impact on the business of the CK Companies after the Closing;
- (vi) make any Tax election or change a Tax election or file for any change in any material respect of any method of accounting with the Internal Revenue Service, except as required by any change in Law;
- (vii) revoke any CK Company's election to be taxed as an S corporation within the meaning of Sections 1361 and 1362 of the Code, or the comparable elections or status for applicable state and local Tax purposes, or take or permit to be taken any action that would reasonably be expected to have the effect of terminating any such election or status;
- (viii) make any change in the methods of accounting or accounting principles applied in the preparation of the financial statements of the CK Companies other than a change which is required by reason of a concurrent change in Law or GAAP;
- (ix) (A) adopt, terminate or amend, except as required by this Agreement or applicable Law, or as permitted by Section 6.2(d), in any material respect (other than as may be required to comply with applicable Law), fund or secure any benefit plan or bonus, profit sharing, deferred compensation, incentive, stock option or stock purchase plan, program or commitment, paid time off for sickness or other plan, program or arrangement for the benefit of its employees, consultants or directors, (B) grant any general increase (other than increases required under a Contract) in the compensation of its employees (including any such increase pursuant to any bonus, profit sharing or other compensation or incentive plan, program or commitment) or any increase (other than increases required under a Contract) in the compensation payable or to become payable to any officer or director, or (C) take any action to or permit any member of the Group to take any action to completely or partially withdraw (within the meaning of Section 4201 of ERISA) from any Multiemployer Plan;

- (x) enter into, amend or modify any Contract except in the ordinary course of business and in no event calling for annual payments, if a new Contract, or an increase in annual payments, if an amendment or modification, by, or to, any CK Company in excess of \$100,000, other than (i) with respect to Permitted Arrangements, and (ii) any amendments and modifications pursuant to grace periods customary in the industry;
- (xi) amend or modify any employment agreement or any consulting agreement with respect to an employee of, or consultant to, any CK Company, as applicable, entitled to receive a base salary or consulting fee, as applicable, in excess of \$75,000 per year;
- (xii) enter into, amend or modify any Contract granting to a third party the right to use any CK Intellectual Property Rights to manufacture and sell products, other than Contracts that may be terminated upon less than 90 days' notice by a CK Company granting the right to manufacture and sell products for seasonal purposes or any co-existence agreement relating to any Mark;
- (xiii) enter into, amend or modify any collective bargaining agreements;
- (xiv) authorize, undertake, or enter into any commitment with respect to, capital expenditure projects individually in excess of \$50,000 or in the aggregate in excess of \$250,000;
- (xv) amend the articles or certificates of incorporation, as applicable, or by-laws of any CK Company, or amend the CK Trust Agreement;
- (xvi) issue, deliver, or agree (actually or contingently) to issue or deliver (whether pursuant to any option or otherwise), or grant or modify any option, warrant or other right to purchase or otherwise acquire, any shares of the capital stock or trust interests of any CK Company or the CK Trust, as the case may be, or any security convertible into or exchangeable for, any shares of such capital stock or trust interests, or issue or agree to issue any bonds, notes, or other securities, except borrowings drawn under the Line Letter and ordinary course trade letters of credit and performance bonds;
- (xvii) split, combine or reclassify any shares of the capital stock or trust interests of any CK Company or the CK Trust, as the case may be, retire, redeem or otherwise acquire any shares of the capital stock or trust interests of any CK Company or the CK Trust, as the case may be, or declare, set aside or make any distributions of property other than pursuant to Section 5.11 or other than cash, cash equivalents or readily marketable securities in respect of the capital stock or trust interests of any CK Company or the CK Trust, as the case may be, or agree to do any of the foregoing;
- (xviii) fail to maintain in force, or make any change in (except in the ordinary course of business), the insurance contemplated by Section 3.13 (or substantially equivalent replacement coverage) as being maintained by any CK Company;

- (xix) issue any communication to employees of any CK Company with respect to compensation, benefits or employment continuation or opportunity following the Closing, except as required by Law;
- (xx) enter into any partnership or joint venture agreement or arrangement or any similar agreement or arrangement;
- (xxi) enter into any Contract which would require a consent thereunder with respect to the consummation of the transactions hereby or by any Related Agreement; or
- $(\ensuremath{\mathsf{xxii}})$ agree, whether in writing or otherwise, to do any of the foregoing.
- 5.2 No Breach of Representations and Warranties; Notification of Certain Matters. The Sellers, the CK Companies or PVH, as the case may be, will, in the event of, and promptly after the occurrence of, or promptly after becoming aware of the occurrence of, or the impending or threatened occurrence of, any event or condition which would result in the inability of any condition contained in Articles VII or VIII to be satisfied or would otherwise prevent it from consummating the transactions contemplated hereby or by any Related Agreement, give detailed written notice thereof to the Sellers, the CK Companies or PVH, as the case may be, and each of the Sellers, the CK Companies or PVH, as the case may be, shall use its reasonable best efforts to prevent or promptly to remedy such event, condition or breach. None of the disclosures pursuant to this Section 5.2 will be deemed to qualify, modify, or amend or supplement the representations, warranties or covenants of any party.
- 5.3 Access. Subject to the terms of the Confidentiality Agreement, the Sellers and CKI shall afford PVH's employees, auditors, legal counsel and other authorized representatives and advisors, as well as representatives of PVH's financing sources, all reasonable opportunity and access during normal business hours to inspect, investigate and audit the assets, liabilities, Contracts, operations and business of the CK Companies and the CK Trust and to interview the employees and officers of the CK Companies. The Sellers and CKI shall also permit PVH to meet with the licensees and other business partners of the CK Companies to discuss the business conducted between the CK Companies and such licensees and business partners so long as representatives of the Sellers and/or the CK Companies participate in such meetings and discussions.
- 5.4 Standstill. None of the CK Companies or the Sellers shall, and shall not permit any of their respective officers, directors, employees, agents or Affiliates to, initiate, solicit, encourage, negotiate or enter into any agreement with any Person other than PVH (and the other Sellers) respecting the acquisition of any portion of the business of any CK Company or any of the Purchase Shares.
- 5.5 Notice of Litigation. Promptly after obtaining knowledge of the occurrence of or the written threatened occurrence of any Action against any CK Company, the CK Trust, any Mark or the Purchase Shares, the Sellers or such CK Company shall give detailed written notice thereof to PVH.

- 5.6 Fulfillment of Conditions to PVH's Obligations. Each Seller agrees, and agrees to cause each CK Company, to use reasonable best efforts to effectuate the transactions contemplated hereby and by the Related Agreements and to fulfill the conditions contained in Article VII and Article VIII, as applicable.
- 5.7 Fulfillment of Conditions to Sellers' Obligations. PVH agrees to use reasonable best efforts to effectuate the transactions contemplated hereby and by the Related Agreements and to fulfill the conditions contained in Article VII and Article VIII, as applicable.
- $5.8\,$ Governmental Consents. Each of PVH, the Sellers and the CK Companies shall as promptly as practicable following the execution and delivery of this Agreement make all filings, notices, petitions, statements, registrations, submissions of information, application or submission of other documents required by any Governmental Authority in connection with the transactions contemplated hereby and by the Related Agreements, including without limitation (i) the notification and report forms with the FTC and the DOJ, and any supplemental information requested in connection therewith pursuant to the HSR Act, and (ii) any other comparable notification forms required by any Governmental Authority. Each party will cause all documents that it is responsible for filing with any Governmental Authority under this Section 5.8 to comply in all material respects with all applicable Law. Each such party shall furnish to the other such necessary information and reasonable assistance as the other may request in connection with its preparation of such filings or submissions. Each such party shall keep the other apprised of the status of any communications with, and any inquiries or requests for additional information from, any Governmental Authority and shall comply promptly with any such inquiry or request. Each such party shall use its reasonable best efforts to obtain any clearance required under applicable Law for the consummation of the transactions contemplated hereby and by the Related Agreements. Filing fees attributable to the filings made pursuant to this Section 5.8 shall be borne by the party responsible for making the filing (it being understood that PVH shall pay all filing fees with respect to any filing required under the HSR Act or any other filings with any Governmental Authority in connection with the transactions contemplated hereby).
- 5.9 Third Party Consents. The CK Companies and the Sellers shall use their respective reasonable best efforts (without the obligation to make payment other than obligations when and as required under existing Contracts) to obtain all consents from parties to Contracts which are required by the terms thereof to be obtained in connection with the transactions contemplated hereby and by the Related Agreements. PVH shall use its reasonable efforts to cooperate in obtaining any such consents, so long as PVH is not required to make any payments with respect thereto.
- 5.10 Publicity. No public release or announcement concerning the transactions contemplated hereby or by the Related Agreements shall be issued by any party without the prior consent of the other party (which consent shall not be unreasonably withheld or delayed), except as such release or announcement may be required by Law or the rules or regulations of any United States or foreign securities exchange, in which case the party required to make the release or announcement shall allow the other party reasonable time to comment on such release or

announcement in advance of such issuance and shall make a reasonable effort to take into account such comments.

5.11 Transfer of Certain Assets. (a) To the extent any of the Sellers (or any of their Affiliates, other than the CK Companies or the CK Trust) own any assets primarily used in or necessary to conduct any CK Company's business, such Seller, effective as of the Closing Time, shall (i) sell, assign, deliver and transfer to PVH or the relevant CK Company or the CK Trust all such assets, and (ii) use its best efforts to cause such Affiliate to enter into an agreement with PVH to sell, assign, deliver and transfer to PVH or the relevant CK Company or the CK Trust all of such assets, in each case for no additional consideration; provided, however, that the property set forth in Section 5.11 of the CK Companies Disclosure Schedule may be retained by (or transferred to) the Sellers on or prior to Closing.

ARTICLE VI.

OTHER AGREEMENTS OF THE PARTIES

- 6.1 Cooperation in Litigation; Retention of Employees
- (a) Prior to the third anniversary of the Closing Date, the Sellers shall provide to PVH (at PVH's sole cost and expense) such cooperation as may reasonably be requested (taking into account the business and personal activities of the Sellers) in connection with the defense of any litigation relating to any CK Company whether existing on the Closing Date or arising thereafter out of, or relating to, an occurrence or event happening before the Closing Date, provided, that such cooperation is limited to matters of which the Sellers have Knowledge.
- (b) From and after the date hereof until the Closing, the Sellers agree to use their reasonable best efforts (at PVH's sole cost and expense which shall not include the payment by PVH of any obligations of any CK Company for periods prior to Closing) to assist PVH in retaining, subsequent to the Closing Date, the services of the employees of the CK Companies whom PVH wishes to employ.

6.2 Employees.

- (a) Any "Severance Qualified Employee" (as hereinafter defined) whose employment with PVH or any CK Company is involuntarily terminated without cause (as determined in good faith by PVH) by PVH or any CK Company during the nine-month period beginning on the Closing Date will be entitled to be paid severance by PVH or such CK Company as follows:
 - (i) With respect to any such Severance Qualified Employee who, as of the employment termination in question, has more than 15 full years of service with a CK Company (taking into account such Severance Qualified Employee's service with PVH or a CK Company after the Closing), such severance will be based on two weeks of such Severance Qualified Employee's salary for each full year of such Severance Qualified Employee's service with a CK Company (taking into account such Severance Qualified

Employee's service with PVH or a CK Company after the Closing), up to a maximum severance payment equal to one year's salary; and

(ii) With respect to any such Severance Qualified Employee who is not described in Section 6.2(a)(i) above, such severance will be based on one week of such Severance Qualified Employee's salary for each full year of such Severance Qualified Employee's service with a CK Company (taking into account such Severance Qualified Employee's service with PVH or a CK Company after the Closing), up to a maximum severance payment equal to one year's salary.

Payment of severance pay, if any, to any individual who is not a Severance Qualified Employee and with respect to employment terminations occurring after the expiration of the nine-month period beginning on the Closing Date will be made in accordance with the then-generally applicable policies of PVH. Any severance payments made hereunder shall be net of withholding and other Taxes required under applicable Law. For purposes hereof, a "Severance Qualified Employee" is any individual, other than an International Employee, who was an employee of any CK Company immediately prior to the Closing, who, prior to the employment termination in question, was continuously employed by a CK Company (taking into account such employee's service with PVH or a CK Company after the Closing) for at least one full year and who, as of the employment termination in question, is not covered by a written agreement that provides such Severance Qualified Employee with a severance benefit. The amount of severance, if any, payable to International Employees whose employment terminates after the Closing shall be determined by PVH in accordance with its then-generally applicable policies and applicable Law.

- (b) PVH agrees that each individual who was an employee of the CK Companies immediately prior to the Closing will receive service credit for all periods of employment with the CK Companies or any predecessor thereto prior to the Closing Date for purposes of eligibility and vesting under any employee plan within the meaning of Section 3(3) of ERISA in which such employee participates after the Closing Date (but specifically excluding credit for benefit accrual purposes) and vacation entitlement.
- (c) With respect to any plan maintained by the CK Companies, PVH or any Affiliate at or following the Closing Date that provides welfare benefits to employees of the CK Companies, with respect to such employees:
 - (i) Any pre-existing condition limitations with respect to any employee of the CK Companies shall be waived to the extent the condition of such employees would have been covered under the plan of the CK Companies in which such employee participated immediately prior to the Closing Date; and
 - (ii) Such employee will be provided with credit for any co-payments and deductible paid prior to the Closing Date and during the calendar year in which the Closing occurs under a CK Companies plan in satisfying any applicable deductible or out-of-pocket requirements for such calendar year under any welfare benefit plan in which such employee may be eligible to participate after the Closing Date.

- (d) Immediately prior to the Closing, the CK Companies shall terminate the Calvin Klein, Inc. Retirement Savings Plan (the "CKI Plan"), effective as of immediately prior to the Closing. Each individual who was an employee of the CK Companies immediately prior to the Closing and who satisfies the requirements for eligibility in either the PVH Associates Investment Plan for Salaried Associates or the PVH Associates Investment Plan for Hourly Associates (each a "PVH Savings Plan"), taking into account service credited pursuant to Section 6.2(b) above, shall be eligible to participate in the applicable PVH Savings Plan immediately following the Closing. Immediately prior to the Closing, PVH and the Sellers shall cooperate and use their commercially reasonable efforts to (i) prevent any loan to a participant under the CKI Plan that is outstanding as of the termination of the CKI Plan from accelerating solely on account of such termination of the CKI Plan, and (ii) obtain any participant consents and adopt any amendments to the CKI Plan necessary to prevent such acceleration. As soon as practicable following the Closing, PVH and the Sellers shall cooperate to communicate such termination of the CKI Plan and such eligibility under the applicable PVH Savings Plan to affected employees.
- (e) Immediately prior to the Closing, the Sellers shall take all reasonable steps to terminate the participation of any individual who, immediately prior to the Closing, is not an employee of any of the CK Companies or an eligible dependent of such an employee in any Employee Benefit Plan maintained by any of the CK Companies and such termination of participation shall not be a "qualifying event" within the meaning of Section 4980B of the Code.
- (f) Following the Closing, each of Mr. Klein and Mr. Schwartz, will be entitled under "COBRA" to continue their current group health plan coverage for a period of up to 18 months; continuation of such coverage shall be at Mr. Klein's and Mr. Schwartz's respective sole discretion and at their sole cost.
 - 6.3 Non-Competition, Non-Solicitation, Non-Derogation and Confidentiality.
- (a) From and after the Closing Date (until the fifth anniversary of the Closing Date with respect to Mr. Schwartz), each of Mr. Klein and Mr. Schwartz agrees individually on his own behalf that he will not, directly or indirectly (including, without limitation, through any other Seller), engage in, be employed by, be a consultant to, own, manage, operate, provide financing to, control or participate in the ownership, management or control of, or otherwise have an interest in any other Person in the conduct of, any Restricted Business or any Person with respect to any Restricted Business, provided, that neither Mr. Klein nor Mr. Schwartz shall be precluded from ownership of (i) the PVH Shares, the Warrant or the shares underlying the Warrant, as applicable, (ii) less than 5% of the issued and outstanding stock or other securities of a corporation listed on a national securities exchange, electronic quotation and trading system or traded in the over-the-counter market, and (iii) 10% or less in a fund or similar investment vehicle, as a passive investor, provided, further that, if PVH has breached its obligations under Section 1.5(a), and such breach continues for 60 days or more after a final non-appealable judgment by a court of competent jurisdiction relating to such breach, Mr. Klein shall not be restricted by the provisions of this Section 6.3. For purposes of this Section 6.3, "Restricted Business" means the business of developing, designing, merchandising, licensing, manufacturing or causing the manufacture of, wearing apparel of any type, shoes, accessories, luggage, watches, jewelry, fragrances, eyewear and optical products, home furnishing products and accessories,

table top housewares, silverware, floor and wall coverings, furniture, leather goods, beauty products, and retail stores, internet sites and locations, primarily engaged in the business of selling any of the foregoing, and excludes, (x) with respect to Mr. Schwartz, the business of all activities related to the equine business, and engaging in any activity that is incidental to the equine business including, horse breeding and horse racing or any other business in which he engages that is not a Restricted Business (e.g., selling NYRA t-shirts), and (y) with respect to Mr. Klein, engaging in any activity that is incidental to any other business in which he engages that is not a Restricted Business (e.g. selling t-shirts at a restaurant bearing the name of the restaurant). In no event shall any agreement contained in this Section 6.3 modify the rights and obligations of Mr. Klein and PVH contained in the Agreement and Assignment. No Seller shall be in violation of this Section 6.3(a) if the action of such Seller would otherwise be permitted by another subparagraph of this Section 6.3 or under the Agreement and Assignment.

- (b) From and after the Closing Date, until the third anniversary of the Closing Date, each of Mr. Klein and Mr. Schwartz agrees individually on his own behalf that he will not, without the prior written consent of PVH, directly or indirectly, for or on behalf of himself or any other Person, employ or solicit for employment any officer or executive or key employee of PVH or of any CK Company (other than those staff members listed in Section 6.3(b) of the CK Companies Disclosure Schedule) at the time of such solicitation or hiring or during the immediately preceding six-month period or cause or seek to cause any such individual to leave the employ of PVH or any CK Company, provided, that this Section 6.3(b) will not be deemed to apply to a general solicitation to the public.
- (c) (i) Each Seller agrees that such Seller shall not make any public comment with the intent to hurt or disparage the Marks or make any public comment with the intent to hurt, disparage or be derogatory of PVH or any of its Affiliates known to such Seller and those individuals known to such Seller to be its officers, directors and key employees. Notwithstanding the foregoing, nothing in this Agreement shall prevent Sellers from (A) making public statements including complimentary statements about third parties and their goods and services in the Restricted Business, or (B) providing general commentary, opinions or interviews regarding any goods or services of a Restricted Business, so long as such commentary, opinions or interviews are not intended to (x) provide any comparison with the goods or services Bearing a Mark that is adverse to such goods or services Bearing a Mark, or (y) be negative about any goods or services marked under the Marks or (C) making any comments otherwise restricted by this Section 6.3(c) that are specifically related to asserting his or their claims filed in a court of law or with another body exercising arbitration or similar quasi-judicial authority or power or defending itself against claims made by PVH or any of its Affiliates.
- (ii) PVH shall not, and shall cause its Affiliates not to, issue or make any public comment with the intent to hurt or to be disparaging or derogatory to a Seller or any member of his or her immediate family or any of their respective affiliates known to PVH to be affiliates of such Seller. Notwithstanding the foregoing, nothing in this Agreement shall prevent PVH or its Affiliates from making any such comments that are specifically related to asserting their claims or defending themselves against claims filed in a court of law or with another body exercising arbitration or similar quasi-judicial authority or power or defending itself against claims made by a Seller.

- (d) Each Seller acknowledges and agrees that the covenants and agreements contained in this Section 6.3 have been negotiated in good faith by each of them and reflect the fame, prestige and geographic coverage of the businesses operated by the CK Companies and their licensees and distributors, as well as the fame and prestige of the Sellers. Each Seller further acknowledges and agrees that the covenants and agreements contained in this Section 6.3 are reasonable and not more restrictive or broader than necessary to protect the interests of PVH in purchasing the CK Companies and would not achieve their intended purpose if they were on different terms or for shorter periods of time or applied in more restrictive geographical and business line areas than are provided herein. If any one or more of the provisions contained in this Section 6.3 shall be held to be excessively broad as to scope, territory or period of time, such provisions shall be construed by limiting and reducing them so as to be enforceable to the maximum extent allowed by applicable Law.
- (e) Each Seller agrees to hold confidential in accordance with commercial business practice all Confidential Information obtained in the course of its ownership of shares or participation in the management of any CK Company or otherwise which is either non-public, confidential or proprietary in nature.
- (f) Each Seller agrees that PVH shall be entitled to seek equitable relief, including injunction and specific performance, in the event of any breach of the provisions of this Section 6.3. Such remedies shall not be deemed to be the exclusive remedies for a breach of this Section 6.3 by any Seller but shall be in addition to all other remedies available at Law or equity (subject to the provisions of Article IX). It is further understood and agreed that failure or delay by PVH in exercising any right, power or privilege under this Section 6.3 shall not operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise of any right, power or privilege under this Agreement. Each Seller hereby waives any requirement that PVH post a bond in connection with any claim for equitable relief.

6.4 Tax Matters.

- (a) The Sellers shall prepare or cause to be prepared and file or cause to be filed in a timely manner all Tax Returns for the CK Companies and the CK Trust for all periods ending on or prior to the Closing Date ("Pre-Closing Tax Periods"). Any such Tax Returns filed between the date hereof and the Closing Date for any Pre-Closing Tax Period shall be prepared, and each item thereon treated, in a manner consistent with past practices (including, without limitation, prior Tax elections and accounting methods or conventions made or utilized by the CK Companies and the CK Trust), except as required by a change in the applicable Law.
- (b) PVH shall prepare or cause to be prepared and file or cause to be filed all Tax Returns for the CK Companies and the CK Trust for all periods beginning before and ending after the Closing Date ("Straddle Periods") which are filed after the Closing Date. All such Tax Returns with respect to Pre-Closing Tax Periods with respect to which the Sellers have agreed to indemnify PVH pursuant to Section 9.2 shall be prepared and filed in a manner that is consistent with past practices (including, without limitation, prior Tax elections and accounting methods or conventions made or utilized by the CK Companies and the CK Trust), except (i) as is due to the termination of the Subchapter S status (or analogous status under state or local Law) of the CK

Companies in connection with the transactions contemplated hereby or (ii) as required by a change in the applicable Law. PVH shall submit such Tax Returns to the Sellers' Representative for review and comment at least 30 days prior to filing. However, if the Sellers' Representative disputes any item on such Tax Return, it shall promptly notify PVH of such disputed item (or items) and the basis for its objection. The parties shall act in good faith to resolve any such dispute prior to the date on which the Tax Return is required to be filed. If the parties cannot resolve any disputed item, the item in question shall be resolved by an independent accounting firm mutually acceptable to the Sellers and PVH. The fees and expenses of such accounting firm shall be borne equally by the Sellers and PVH. All disputes must be resolved prior to the date on which the Tax Return is required to be filed and the parties shall resolve all disputes (in the manner described herein) and the applicable CK Company (or the CK Trust) shall sign and file the Tax Return on or before the last day on which such Tax Return is required to be filed.

- (c) The Sellers shall pay or cause to be paid when due and payable all unreserved Taxes of each of the CK Companies and the CK Trust, to the extent allocable to any of the CK Companies, for any Pre-Closing Tax Period and any pre-Closing portion of a Straddle Period; and (ii) PVH shall so pay or cause to be paid such Taxes for any Tax periods commencing on or after the Closing Date and any post-Closing portion of a Straddle Period. If the aggregate reserves for accrued and unpaid Taxes of the CK Companies for any Pre-Closing Tax Period and any pre-Closing portion of a Straddle Period exceeds the aggregate Taxes shown on the applicable Tax Returns later filed for such periods, then PVH shall pay such excess to the Sellers, but only to the extent that such Tax reserves were reflected as liabilities or otherwise reduced the Closing Date Net Book Value.
- (d) PVH and the Sellers agree to furnish or cause to be furnished to each other, upon request, as promptly as practicable, such information and assistance relating to each CK Company and the CK Trust (including, without limitation, access to books and records, employees, contractors and representatives) as is reasonably necessary for the filing of all Tax Returns, the making of any election related to Taxes, the preparation for and conduct of any audit by any Tax Authority, and the prosecution or defense of any claim, suit or proceeding relating to any Tax Return. Such cooperation and information shall include providing copies of relevant Tax Returns or portions thereof, together with accompanying schedules, related work papers and documents relating to rulings or other determinations by Tax Authorities at the sole cost of the requesting party. If reasonably requested by PVH, the Sellers' Representative shall make himself available on a basis mutually convenient to both parties to provide explanations of any documents or information provided hereunder. PVH and the Sellers shall retain all books and records with respect to Taxes pertaining to the CK Companies and the CK Trust in their respective possession (which shall include, in the case of PVH, books and records in the possession of the CK Companies) until the later of (i) the expiration of the statute of limitations of the Tax periods to which such Tax Returns and other documents relate, without regard to extensions except to the extent notified by the other party in writing of such extensions for the respective Tax periods prior to such expirations, and (ii) six years following the due date (without extension) for such Tax Returns. At the end of such period, each party shall provide the others with at least ten days' prior written notice before destroying any such books and records, during which period the parties receiving such notice can elect to take possession, at their own expense, of such books and records. PVH and the Sellers shall cooperate with each other in the

conduct of any audit or other proceeding related to Taxes involving the CK Companies. Any information provided or obtained under this Section 6.4 shall be kept confidential except as may be otherwise necessary in connection with the filing of Tax Returns or in conducting an audit or other proceeding.

- (e) As a result of the transfer of the Purchase Shares, each of the CK Companies and the CK Trust shall be treated for federal income Tax purposes as having two short years for Tax purposes during the calendar year in which the Closing occurs pursuant to Section 1362(e)(1) of the Code: (i) a period from January 1 of the calendar year in which the Closing occurs through the end of the day prior to the Closing Date; and (ii) a period from the beginning of the Closing Date through December 31 of the calendar year in which the Closing occurs. Accordingly, each of the CK Companies and the CK Trust shall close their books as of the Closing Date.
- (f) The Sellers and PVH agree that each will pay one-half of all applicable transfer Taxes, sales and/or use Taxes, real property transfer or excise Taxes, recording, deed, stamp and other similar Taxes, fees and duties under applicable Law incurred in connection with the transfer of the Purchase Shares. The Sellers and PVH agree to jointly prepare or cause to be prepared and file or cause to be filed in a timely manner, all Tax Returns required to be filed with respect to such Taxes.
- (g) For all Tax purposes, PVH, the Sellers, and the CK Companies agree that they will report the transactions contemplated by this Agreement in a manner consistent with the terms of this Agreement, and all parties agree to file their Tax Returns accordingly.
- (h) Mr. Klein shall, and shall cause each holder, if any, of rights with respect to the Design Services Purchase Payments (in either case, a "Rights Holder") to, provide to PVH two properly completed and executed Internal Revenue Service Forms W-9 or other applicable forms, certificates or documents prescribed by the Internal Revenue Service of the United States certifying as to such Rights Holder's entitlement to exemption from United States withholding Tax and back-up withholding Tax with respect to the Design Services Purchase Payments to be made to such Rights Holder hereunder ("Certificates of Exemption"). Mr. Klein shall provide such Certificates of Exemption on or before the Closing Date and thereafter as and when requested to do so by PVH. Mr. Klein shall cause each Person that becomes a Rights Holder after the Closing Date to provide Certificates of Exemption on or before the date such Person becomes a Rights Holder and thereafter as and when requested to do so by PVH. Mr. Klein agrees, and shall cause each Rights Holder to agree, to provide PVH with new Certificates of Exemption (x) upon the obsolescence of any letter, form, certificate or document previously delivered by such Rights Holder to PVH hereunder or (y) promptly after the occurrence of any event requiring a change in the status of the Rights Holder or in any of the other information provided on the most recent letter, form, certificate or document previously delivered by the Rights Holder to PVH hereunder. If PVH shall provide written notice to a Rights Holder that new Certificates of Exemption are required, and if the Rights Holder does not submit them within 30 days following the receipt of such notice, then PVH shall be permitted to deduct all applicable withholding Taxes from the payment due to such Rights Holder.

- 6.5 Access. From and after the Closing Date, PVH shall and shall cause the CK Companies to afford to the Sellers and their agents reasonable access to their properties, books, records, employees and auditors (during normal business hours and upon reasonable prior written notice) to the extent necessary to permit the Sellers to determine any matter relating to their rights and obligations hereunder or to any period ending on or before the Closing Date; provided, that any such access by the Sellers or their agents shall not unreasonably interfere with the conduct of the business of PVH or the CK Companies.
- 6.6 Directors' and Officers' Indemnification and Insurance. To the extent any director, officer or employee of any of the CK Companies has the benefit of any right to indemnification pursuant to the certificate of incorporation or by-laws or similar organizational documents of any of the CK Companies or the CK Trust, PVH agrees to continue such indemnification and not to amend, alter or waive in any adverse respect any such indemnification for a period of not less than six years following the Closing; provided, however, PVH shall not be required to provide such indemnification for any liability or expense of such director, officer or employee arising out of the willful malfeasance, bad faith or fraud of any such director, officer or employee, or for any liability or expense arising from any state of facts, event or condition for which any PVH Group Member would be entitled to indemnification under Article IX, disregarding any limitations set forth in Section 9.4 and the survival periods set forth in Section 9.1. In addition, PVH will provide, or cause the CK Companies to provide, for a period of not less than six years after the Closing, each of the CK Companies' current directors and officers an insurance and indemnification policy that provides coverage for events occurring at or prior to the Closing that is no less favorable than the existing policy pursuant to which such directors and officers are covered; provided, however, that PVH and the CK Companies shall not be required to pay an annual premium for such policy in excess of 200% of the annual premium currently paid by the CK Companies for such insurance, but in such case shall purchase as much of such coverage as possible for such amount.
- 6.7 Offices. PVH agrees that it will maintain and make available, at PVH's sole cost and expense, to Mr. Klein and Mr. Schwartz, their existing office space for the benefit of each of Mr. Klein and Mr. Schwartz on an as-needed basis and for transition purposes for a period of up to 90 days following the Closing.

6.8 Legend.

(a) The certificates evidencing the PVH Shares, the Warrant and the shares of the PVH Common Stock issuable upon conversion of the Warrant will bear a legend substantially similar to the following:

"THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR THE SECURITIES LAWS OF ANY JURISDICTION. SUCH SECURITIES MAY NOT BE OFFERED, SOLD OR TRANSFERRED EXCEPT PURSUANT TO (I) A REGISTRATION STATEMENT WITH RESPECT TO SUCH SECURITIES THAT IS EFFECTIVE UNDER SAID ACT AND APPLICABLE STATE SECURITIES LAW, OR (II) ANY

EXEMPTION FROM REGISTRATION UNDER SAID ACT, AND APPLICABLE STATE SECURITIES LAW, RELATING TO THE DISPOSITION OF SECURITIES."

- (b) The legend endorsed on the certificates pursuant to Section 6.8(a) shall be removed and the Company shall issue a certificate without such legend to the holder thereof at such time as the securities evidenced thereby cease to be restricted securities upon the earliest to occur of (i) a registration statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities shall have been disposed of in accordance with such registration statement, (ii) the securities shall have been sold to the public pursuant to Rule 144 (or any successor provision) under the Securities Act, and (iii) such securities may be sold by the holder without restriction or registration under Rule 144(k) under the Securities Act (or any successor provision).
- $6.9\,$ Certain Matters. Each party agrees to comply with Section 6.9 of the CK Companies Disclosure Schedule.
- 6.10 Further Assurances. From and after the Closing Date, each party shall, at any time and from time to time, make, execute and deliver, or cause to be made, executed and delivered, for no additional consideration but at the cost and expense of the requesting party (excluding any internal costs incurred, such as having any of the following reviewed by counsel) such assignments, deeds, drafts, checks, stock certificates, returns, filings and other instruments, agreements, consents and assurances and take or cause to be taken all such actions as the other party or its counsel may reasonably request for the effectual consummation and confirmation of this Agreement and the Related Agreements and the transactions contemplated hereby and by the Related Agreements.

ARTICLE VII.

CONDITIONS PRECEDENT TO OBLIGATIONS OF PVH

The obligation of PVH to consummate the transactions contemplated under this Agreement are subject to the fulfillment of each of the following conditions, any or all of which may be waived in whole or in part by PVH, in its sole discretion:

- 7.1 Representations and Warranties. The representations and warranties contained in Articles II and III shall be materially true and correct when taken as a whole as of the Closing Time.
- 7.2 Performance. The CK Companies and the Sellers shall each have performed and complied in all material respects with all covenants and agreements required by this Agreement to be performed or complied with by any CK Company or any Seller prior to or at the Closing Time.

- 7.3 No Material Adverse Effect. Between the date of the execution of this Agreement and the Closing Date, none of the CK Companies shall have suffered or experienced a CK Companies Material Adverse Effect.
- 7.4 Certificates. PVH shall have received (a) a certificate of an executive officer of the CK Companies dated the Closing Date (or the HSR Date, if applicable) and certifying as to the knowledge of such officer (taking into account the Knowledge of those persons identified in the definition of "Knowledge" contained in Appendix A) to the fulfillment on the part of the CK Companies of the conditions specified in Sections 7.1, 7.2, and 7.3; (b) a certificate of the Secretary or Assistant Secretary of the CK Companies dated the Closing Date, setting forth the resolutions of the Board of Directors of the CK Companies adopting and approving this Agreement, the Related Agreements (to the extent a party thereto) and all other documents contemplated hereby and thereby and authorizing the transactions hereby and thereby contemplated; and (c) such other evidence with respect to the fulfillment of said conditions as PVH may reasonably request.
- $7.5\,$ No Injunction. There shall not be in effect any injunction or restraining order issued by a court of competent jurisdiction in an Action against the consummation of the transactions contemplated hereby or by any Related Agreement.
- 7.6 CK Debt Obligations; Liens Discharged. All of the CK Debt Obligations shall have been repaid in full and cancelled by the CK Companies and all Liens granted under the Line Letter shall have been discharged, and the Sellers shall have delivered reasonably satisfactory evidence to PVH thereof.
- $7.7\,$ HSR Act Clearance. The transactions contemplated hereby and by the Related Agreements shall have received HSR Clearance.
- 7.8 Government Approvals. In addition to the HSR Clearance, the parties hereto shall have received all other approvals from any applicable Governmental Authority necessary to consummate the transactions contemplated hereby and by the Related Agreements, other than non-material approvals of a local governmental authority.
- 7.9 Third Party Consents. The CK Companies and the Sellers shall have obtained any written consents necessary to consummate the transactions contemplated hereby and by the Related Agreements, in form and substance reasonably satisfactory to PVH, relating to any Contracts to which any CK Company or any Seller is a party.
- 7.10 Consulting Agreement. Mr. Klein shall have entered into an agreement (the "Consulting Agreement") with PVH in the form attached hereto as Exhibit D.
- 7.11 Agreement and Assignment. Mr. Klein shall have entered into an agreement (the "Agreement and Assignment") with PVH and CKI in the form attached hereto as Exhibit E.
- 7.12 Notarial Deed. The Sellers shall have executed the notarial deed relating to CK Italy (the "Notarial Deed") in substantially the form attached hereto as Exhibit F and such other documents including any proxies as are necessary to effect the sale of all of the parts of capital of

CK Italy to PVH (and its designee) in accordance with the terms of this Agreement and applicable Law.

- 7.13 Stock Certificates. PVH shall have received the Stock Certificates.
- 7.14 Good Standing. PVH shall have received long-form good standing certificates, where recognized (or the equivalent thereto in the relevant jurisdiction) relating to each CK Company and the CK Trust, dated within five Business Days of the Closing Date (or the HSR Date, if applicable), issued by the Secretary of State (or other similar official) of their respective states of incorporation or organization, as the case may be.
- 7.15 Resignations. PVH shall have received the written resignations of all directors of the CK Companies (and in the case of CK Italy, any internal auditors), effective as of the Closing Time.
- 7.16 Stockholder Loans. Any loan from any CK Company to a stockholder of any CK Company shall have been paid in full by such stockholder as of the Closing Time; it being understood that any such loan of a Seller may be paid simultaneously at Closing with the proceeds of the Cash Purchase Price, and the Sellers shall have delivered reasonably satisfactory evidence to PVH thereof.
- 7.17 Termination under Design Services Agreement. Mr. Schwartz shall have entered into an agreement in substantially the form attached hereto as Exhibit G (the "Schwartz Design Services Form") evidencing the termination of the rights and obligations of Mr. Schwartz under the Design Services Agreement.
- 7.18 Escrow Agreement. The Sellers shall have entered into the Escrow Agreement.

ARTICLE VIII.

CONDITIONS PRECEDENT TO OBLIGATIONS OF THE SELLERS

The obligation of the Sellers to consummate the transactions contemplated by this Agreement are subject to the fulfillment of each of the following conditions, any or all of which may be waived in whole or in part by the Sellers:

- $8.1\,$ Representations and Warranties. The representations and warranties contained in Article IV shall be materially true and correct when taken as a whole as of the Closing Time.
- 8.2 Performance. PVH shall have performed and complied in all material respects with all covenants and agreements required by this Agreement to be performed or complied with by it prior to or at the Closing Time.
- 8.3 Certificates. The Sellers shall have received (a) a certificate of an executive officer of PVH, dated the Closing Date, certifying to the fulfillment of the conditions specified in Sections 8.1 and 8.2; (b) a certificate of the Secretary of PVH, dated the Closing Date, setting

forth the resolutions of the Board of Directors of PVH approving this Agreement, the Related Agreements and all other documents contemplated hereby and thereby, and authorizing the transactions hereby and thereby contemplated; and (c) such other evidence with respect to the fulfillment of any of said conditions as the Sellers may reasonably request.

- 8.4 No Injunction. There shall not be in effect any injunction or restraining order issued by a court of competent jurisdiction in an Action against the consummation of the transactions contemplated hereby or by any Related Agreement.
- $8.5\,$ HSR Act Clearance. The transactions contemplated hereby and by the Related Agreements shall have received HSR Clearance.
- 8.6 Government Approvals. In addition to the HSR Clearance, the parties hereto shall have received all other approvals from any applicable Governmental Authority necessary to consummate the transactions contemplated hereby and by the Related Agreements.
- 8.7 Consulting Agreement. PVH shall have entered into the Consulting Agreement.
- $8.8\,$ Agreement and Assignment. PVH shall have entered into the Agreement and Assignment.
- 8.9 Equity Matters. PVH shall have delivered the PVH Shares (subject to the Holdback Shares delivered in accordance with the Escrow Agreement) to the Sellers in the proportions as set forth on Exhibit A and shall have executed and delivered to Mr. Klein the Warrant.
- 8.10 Notarial Deed. PVH shall have executed the Notarial Deed and such other documents including proxies as are necessary to effect the sale of all of the parts of capital of CK Italy to PVH (and its designee) in accordance with the terms of this Agreement and applicable Law.
- 8.11 Termination under Design Services Agreement. PVH shall have entered into the Schwartz Design Services Form.
- 8.12 Security Agreements. PVH shall have entered into the Design Services Security Agreement and the Intercreditor Agreement and shall have caused the Guarantee to be entered into by CKI.
- 8.13 Registration Rights Agreement. PVH shall have entered into the registration rights agreement (the "Registrations Rights Agreement") with the Sellers and the other parties thereto in the form attached hereto as Exhibit H.
 - 8.14 Escrow Agreement. PVH shall have entered into the Escrow Agreement.
- 8.15 Opinion. Mr. Klein shall have received the opinion of Katten Muchin Zavis Rosenman in form and substance reasonably satisfactory to Mr. Klein and his counsel, but such

opinion shall not relate to the PVH Shares, the Warrant, or the shares of PVH Common Stock underlying the Warrant, or the purchase and sale of the Purchase Shares.

ARTICLE IX.

INDEMNIFICATION

- 9.1 Survival. All representations and warranties made herein (or in the certificates to be delivered pursuant to Sections 7.4 or 8.3 hereof) by the parties to this Agreement and their respective obligations, covenants and agreements to be performed pursuant to the terms hereof, shall survive the Closing Time, provided, that, the representations and warranties made herein (or in such certificates) by the parties shall terminate on the second anniversary of the Closing Date, except that (i) the representations and warranties set forth in Sections 3.15 (Tax Matters), 3.19 (Environmental Matters) and 3.20 (Employee Plans) shall survive the Closing Time for the applicable statute of limitations, and (ii) the representations and warranties set forth in Sections 2.1 (Authority; Execution and Delivery; Enforceability), 2.2 (Non-Contravention), 2.3 (Title to Purchase Shares), 3.1 (Organization; Good Standing), 3.3 (Authority; Execution and Delivery; Enforceability) and 3.6 (Capitalization; Options; Trust Interests) shall survive the Closing Time indefinitely. Notwithstanding the foregoing, if written notice of any matter setting forth in reasonable detail a claim for a breach of any representation or warranty is given to PVH or the applicable Sellers, as the case may be, in writing pursuant to this Agreement prior to the end of the applicable survival period, any such representation or warranty that would otherwise terminate shall be deemed to survive solely with respect to such matter until such matter is resolved.
 - 9.2 Indemnification by the Sellers.
- (a) Mr. Klein, jointly and severally as to Mr. Klein and the other Klein Group Seller, and the other Klein Group Seller, severally as to such Klein Group Seller, shall indemnify and hold harmless any PVH Group Member from and against any and all Damages incurred by such PVH Group Member arising from:
 - (i) any failure by any Klein Group Seller to perform any covenants or other obligations of any Klein Group Seller contained in Article I or Section 6.3;
 - (ii) any breach of any representation or warranty of any Klein Group Seller contained in Article II (as of the date hereof for purposes of indemnification hereunder), provided, that for purposes of this Section 9.2 (a) (ii), with respect to any representation or warranty qualified by materiality, a breach of such representation or warranty will be deemed to occur if there would have been a breach of such representation or warranty absent such qualification; or
 - (iii) the allegation in writing by any third party of the existence of any liability, obligation, lease, agreement, contract, other commitment or state of facts which, if such allegation were true, would have constituted a breach by any Klein Group Seller of any

representation or warranty of such Klein Group Seller contained in Article II (as of the date hereof for purposes of indemnification hereunder), provided, that for purposes of this Section 9.2(a)(iii), with respect to any representation or warranty qualified by materiality, a breach of such representation or warranty will be deemed to occur if there would have been a breach of such representation or warranty absent such qualification.

- (b) Mr. Schwartz, jointly and severally as to Mr. Schwartz and the other Schwartz Group Sellers, and the other Schwartz Group Sellers, severally as to such Schwartz Group Seller, shall indemnify and hold harmless any PVH Group Member from and against any and all Damages incurred by such PVH Group Member arising from:
 - (i) any failure by any Schwartz Group Seller to perform any covenants or other obligations of any Schwartz Group Seller contained in Article I or Section 6.3;
 - (ii) any breach of any representation or warranty of any Schwartz Group Seller contained in Article II (as of the date hereof for purposes of indemnification hereunder), provided, that for purposes of this Section 9.2(b)(ii), with respect to any representation or warranty qualified by materiality, a breach of such representation or warranty will be deemed to occur if there would have been a breach of such representation or warranty absent such qualification; or
 - (iii) the allegation in writing by any third party of the existence of any liability, obligation, lease, agreement, contract, other commitment or state of facts which, if such allegation were true, would have constituted a breach by any Schwartz Group Seller of any representation or warranty of such Schwartz Group Seller contained in Article II (as of the date hereof for purposes of indemnification hereunder), provided, that for purposes of this Section 9.2(b)(iii), with respect to any representation or warranty qualified by materiality, a breach of such representation or warranty will be deemed to occur if there would have been a breach of such representation or warranty absent such qualification.
- (c) Each of Mr. Klein and Mr. Schwartz shall indemnify (to the extent set forth in Section 9.4(c)) and hold harmless any PVH Group Member from and against any and all Damages incurred by such PVH Group Member arising from:
 - (i) any failure by any CK Company (as constituted prior to the Closing) to perform any covenants or other obligations of any CK Company contained in this Agreement, or any failure by any Seller to perform any covenants or other obligations of any Seller contained in this Agreement (other than in Article I or Section 6.3);
 - (ii) any breach of any representation or warranty of any Seller or any CK Company (as constituted prior to the Closing) contained in Article III (as of the date hereof for purposes of indemnification hereunder);
 - (iii) the allegation in writing by any third party of the existence of any liability, obligation, lease, agreement, contract, other commitment or state of facts which, if such allegation were true, would have constituted a breach by any Seller or any CK Company (as constituted prior to the Closing) of any representation or warranty of such Seller or

such CK Company (as constituted prior to the Closing) contained in Article III (as of the date of this Agreement for purposes of indemnification hereunder);

- (iv) any and all Taxes imposed upon or assessed against any of the CK Companies by reason of liability under Treasury Regulation Section 1.1502-6 (or any similar provision of state, local or foreign Law), for periods or portions thereof ending on or prior to the Closing Date, or pursuant to any Tax sharing agreement, indemnification agreement, partnership or trust agreement or other contract in effect prior to the Closing Date, to the extent that such Taxes exceed Taxes which are included as current liabilities (excluding any reserve for deferred Taxes established to reflect timing differences between book and Tax income) in the Closing Date Balance Sheet;
- (v) Taxes of any of the CK Companies or the CK Trust for periods or portions thereof ending on or before the Closing Date, to the extent that such Taxes exceed Taxes which are included as current liabilities (excluding any reserve for deferred Taxes established to reflect timing differences between book and Tax income) in the Closing Date Balance Sheet, including without limitation, Taxes required to have been withheld;
- (vi) any and all Taxes imposed upon or assessed against any of PVH, the CK Companies, the CK Trust or their respective assets as a result of the transfer of the Purchase Shares or rights under the Design Services Agreement as contemplated by this Agreement;
- (vii) the matter of Sydney Bachman v. Calvin Klein, Inc. (provided, however, to the extent that the amount reserved for such litigation on the Closing Date Balance Sheet, exceeds the Damages arising from such litigation, after the final judicial determination or settlement of such litigation, then PVH shall pay such excess to the Sellers); or
- (viii) the notice from Union of Needletrades Industrial and Textile Employees regarding potential claims relating to the production of goods in the Commonwealth of the Northern Marianas Islands (to the extent the amount of such Damages is in excess of amounts paid prior to Closing in respect thereof).
- 9.3 Indemnification by PVH. PVH will indemnify and hold harmless any Seller Group Member from and against any and all Damages incurred by such Seller Group Member arising from:
- (a) any failure by PVH to perform any of its covenants or other obligations in this Agreement;
- (b) any breach of any representation or warranty of PVH contained in Article IV (as of the date hereof for purposes of indemnification hereunder), provided, that for purposes of this Section 9.3(b), with respect to any representation or warranty that is qualified by materiality, a breach of such representation or warranty will be deemed to occur if there would have been a breach of such representation or warranty absent such qualification;

- (c) the allegation in writing by any third party of the existence of any liability, obligation, lease, agreement, contract, other commitment or state of facts which, if such allegation were true, would have constituted a breach by PVH of any representation or warranty of PVH contained in this Agreement (as of the date hereof for purposes of indemnification hereunder), provided, that for purposes of this Section 9.3(c), with respect to any representation or warranty that is qualified by materiality, a breach of such representation or warranty will be deemed to occur if there would have been a breach of such representation or warranty absent such qualification; or
- (d) the ownership or operation of PVH, the CK Companies and the CK Trust with respect to all periods and the allegation in writing by any third party of the existence of any liability, obligation, lease, agreement, contract, other commitment or state of facts which, if such allegation were true, would give rise to indemnification under this Section 9.3(d); provided, however, PVH shall not be required to provide such indemnification to the extent such Damages are incurred as a result of a state of facts, event or condition for which any PVH Group Member would be entitled to indemnification under this Article IX, disregarding any limitations set forth in Section 9.4 and the survival periods set forth in Section 9.1 or as a result of fraud by a Seller Group Member.

9.4 Limitations on Indemnification.

- (a) No PVH Group Member shall be entitled to be indemnified pursuant to Sections 9.2(a)(ii) or (iii), 9.2(b)(ii) or (iii), 9.2(c)(ii) or (iii), or for any breach of Section 5.2, as the case may be, unless and until the aggregate of all Damages incurred by PVH Group Members shall exceed \$5,000,000 (the "Basket"), and the maximum aggregate amount of indemnification pursuant to Sections 9.2(a)(ii) or (iii), 9.2(b)(ii) or (iii), 9.2(c)(ii) or (iii) or for breaches of Section 5.2 that may be received by PVH Group Members shall not exceed \$100,000,000 (the "Cap"); provided, however, PVH Group Members shall be entitled to be indemnified for all Damages on a dollar-for-dollar basis from the first dollar of Damages, without regard to the Basket or the Cap, incurred as a result of any breach of the representations and warranties set forth in Sections 2.1, 2.3, 3.1(b), 3.3, 3.6 or 3.15. Notwithstanding anything contained herein to the contrary, nothing in this Section 9.4 shall limit PVH's or CKI's (as constituted after the Closing) rights under the Agreement and Assignment or the Consulting Agreement.
- (b) The maximum aggregate amount of indemnification for breaches of the representations and warranties contained in Sections 4.9, 4.10, 4.11, 4.12, 4.13, 4.14, 4.15 and 4.16 that may be received by Seller Group Members shall not exceed \$30,000,000.
- (c) Without limitation to the foregoing, for purposes of Sections 9.2(c) (ii) and 9.2(c) (iii), in determining the amount of any Damages in connection with any inaccuracy of a representation and warranty (but not for purposes of determining whether any such inaccuracy has occurred), any materiality or CK Companies Material Adverse Effect qualifier in such representation or warranty will be disregarded.
- (d) Notwithstanding the preamble to Article III, in no event will Mr. Klein or Mr. Schwartz be liable for more than one-half of any Damages arising under Section 9.2(c). PVH

agrees to proceed against Mr. Klein and Mr. Schwartz at the same time in connection with any action arising under Section 9.2(c).

- (e) The amount of any Damages for which indemnification is provided under this Article IX shall be net of (i) any amount accrued or reserved for on the Closing Date Balance Sheet in respect of Damages, and (ii) any insurance proceeds available under any insurance policies as then in effect to an Indemnitee hereunder in connection with the events or circumstances giving rise to the indemnification, but only to the extent that the Indemnitee actually receives any such insurance proceeds or any benefits thereof and only in an amount not to exceed the amount of any such insurance proceeds actually received by the Indemnitee. The Indemnitee will use commercially reasonable efforts to claim and recover under such insurance policies.
- (f) The parties acknowledge and agree that after the Closing, the indemnification provisions contained in Sections 9.2 and 9.3 shall be the sole and exclusive remedy for Damages arising out of or caused by the breach of any of the representations, warranties, covenants or agreements of the parties contained in this Agreement or in any certificate delivered in connection herewith, except for any remedies that may be available under Section 6.3, for a breach of Section 1.5, or with respect to claims arising out of fraud.
- 9.5 Tax Treatment of Indemnity Payments. It is the intention of the parties to treat any indemnity payment made under this Agreement as an adjustment to the Purchase Price for all federal, state, local and foreign Tax purposes, and the parties agree to file their Tax Returns accordingly.
- 9.6 Tax Refunds. PVH shall pay to Sellers or credit against their indemnification obligations (if applicable), all refunds or credits of Taxes received by PVH or any CK Company after the Closing Date and attributable to Taxes paid by or on behalf of any CK Company (unless paid by PVH) with respect to Pre-Closing Tax Periods net of any Taxes imposed on such refund amount; provided, however, that the rights and obligations described in this sentence shall not apply (i) if such refunds or credits of Taxes were reflected as assets or otherwise increased the Closing Date Net Book Value, (ii) to any net operating loss of any CK Company arising in any Pre-Closing Tax Period that is utilized by PVH in any period beginning after the Closing Date, or (iii) to any refund or credit that is attributable to the carry-back of losses, deductions, or credits from a taxable period other than a Pre-Closing Tax Period. The Sellers agree that they will promptly pay PVH the amount of any refund or credit received by them that results from PVH's carry-back of any losses or other Tax attributes to a Pre-Closing Tax Period. Each such payment (which shall include any interest received or credited with respect to such refund or credit) shall be made to the other party promptly after receipt of any such refund from, or allowance of such credit by, the relevant Tax Authority. The Sellers and PVH shall cooperate (at the sole cost and expense of the party seeking the applicable refund or credit) in obtaining any refund or credit of Taxes available from the relevant Tax Authority. In the event that any refund or credit of Taxes for which a payment has been made pursuant to this Section 9.6 is subsequently reduced or disallowed, the party that ultimately received the benefit of the refund or credit pursuant to this Section 9.6 shall promptly pay to the other party the amount of any refund or credit that has been disallowed or the amount by which such refund or credit has been

reduced and shall indemnify, defend, and hold harmless the other parties for any Tax liability, including interest and penalties, assessed against such indemnified person by reason of the reduction or disallowance.

- 9.7 Notice of Claims. Any PVH Group Member or Seller Group Member seeking indemnification hereunder (an "Indemnitee") shall give to the party or parties obligated to provide indemnification to such Indemnitee (an "Indemnitor") a notice ("Claim Notice") describing in reasonable detail the facts giving rise to any claim for indemnification hereunder and shall include in such Claim Notice (if then known) the amount or the method of computation of the amount of such claim, and a reference to the provision of this Agreement or any other agreement, document or instrument executed hereunder or in connection herewith upon which such claim is based.
- 9.8 Third Party Claims. In the case of any third party Action as to which indemnification is sought by an Indemnitee, the Indemnitor shall have 15 Business Days after receipt of a Claim Notice to notify the Indemnitee that it elects to conduct and control such Action. If the Indemnitor elects to conduct and control such Action, the Indemnitor shall agree promptly to reimburse the Indemnitee for the full amount of any Damages resulting from such Action, except fees and expenses of counsel for the Indemnitee incurred after the assumption of the conduct and control of such Action by the Indemnitor. If the Indemnitor does not give the foregoing notice, the Indemnitee shall have the right, at the sole expense of the Indemnitor, to conduct and control such Action, provided, that (x) the Indemnitee shall permit the Indemnitor to participate in such conduct or settlement through counsel chosen by the Indemnitor, but the fees and expenses of such counsel shall be borne by the Indemnitor, and (y) the Indemnitee may not compromise or settle such Action without the consent of the Indemnitor (which consent will not be unreasonably withheld or delayed), unless (i) there is no finding or admission of any violation of Law by the Indemnitor or any violation by the Indemnitor of the rights of any Person and no effect on any other claims that may be made against the Indemnitor, (ii) the sole relief provided is monetary Damages that are paid in full by the Indemnitee, (iii) the Indemnitor shall have no liability with respect to any compromise or settlement and (iv) such settlement includes an unconditional release in favor of the Indemnitor by the third-party claimant from all liability with respect to such claim, or (v) the claim is solely for other than money Damages which would not affect the business of the Indemnitor and the Indemnitee waives any right to indemnification for any claim of the type identified in this clause (v), other than with respect to reimbursement of the reasonable fees and expenses incurred in the defense of such Action. If the Indemnitor gives the foregoing notice, subject to the first and second sentences of this Section 9.8, the Indemnitor shall have the right, at the sole expense of the Indemnitor, to conduct and control, such Action with counsel reasonably acceptable to the Indemnitee, and the Indemnitee shall cooperate with the Indemnitor in connection therewith, provided, that (x) the Indemnitor shall permit the Indemnitee to participate in such conduct or settlement through counsel chosen by the Indemnitee, but the fees and expenses of such counsel shall be borne by the Indemnitee, and (y) the Indemnitor may not compromise or settle any such Action without the consent of the Indemnitee (which consent will not be unreasonably withheld or delayed) unless (i) there is no finding or admission of any violation of Law by the Indemnitee or any violation by the Indemnitee of the rights of any Person and no effect on any other claims that may be made against the Indemnitee, (ii) the sole relief provided is money Damages that are paid in full by the

Indemnitor, (iii) the Indemnitee shall have no liability with respect to any compromise or settlement and (iv) such settlement includes an unconditional release in favor of the Indemnitee by the third-party claimant from all liability with respect to such claim. In the case of any third party Action as to which indemnification is sought by the Indemnitee which involves a claim for Damages other than solely for money Damages which could have a continuing effect on the business of the Indemnitee, the Indemnitee and the Indemnitor shall jointly control the conduct of such Action. The parties hereto shall use their reasonable best efforts to minimize any Damages from claims by third parties and shall act in good faith in responding to, defending against, settling or otherwise dealing with such claims, notwithstanding any dispute as to liability under this Article IX.

ARTICLE X.

TERMINATION

- 10.1 Termination. This Agreement may be terminated at any time prior to the Closing Date:
 - (a) by mutual consent of PVH and all of the Sellers; or
- (b) by PVH, if there has been (i) a material breach by any CK Company or any Seller of the representations and warranties contained in this Agreement when taken as a whole or (ii) a material violation by any CK Company or any Seller of any covenant or agreement contained in this Agreement, provided, that written notice of such violation or breach shall have been given to such CK Company or such Seller, as applicable, and such violation or breach shall not have been cured within ten days of receipt of such notice (provided, further, that, to exercise its rights under this Section 10.1(b) for any particular violation or breach, PVH must terminate this Agreement within 15 Business Days after the cure period provided for in this Section 10.1(b)); or
- (c) by all of the Sellers, if there has been (i) a material breach by PVH of the representations and warranties contained in this Agreement when taken as a whole or (ii) a material violation by PVH of any covenant or agreement contained in this Agreement, provided, that written notice of such violation or breach shall have been given to PVH and such violation or breach shall not have been cured within ten days of receipt of such notice (provided, further, that, to exercise their rights under this Section 10.1(c) for any particular violation or breach, the Sellers must terminate this Agreement within 15 Business Days after the end of the cure period provided for in this Section 10.1(c)); or
- (d) by either PVH, on one hand, or all of the Sellers, on the other hand, if the Closing shall not have occurred by March 31, 2003 (the "Termination Date").
- 10.2 Effects of Termination. (a) In the event of a termination of this Agreement pursuant to this Article X (i) all further obligations of the parties under this Agreement shall terminate, (ii) no party shall have any right under or in connection with this Agreement or the transactions contemplated hereby against any other party except as set forth in Sections 11.1, 11.8 (to the

extent such rights were pursued through legal process prior to such termination) and the Confidentiality Agreement, and (iii) each party shall bear its own costs and expenses, except as provided in Section 11.1; provided, however, that any termination of this Agreement under this Article X shall not relieve any party of liability for any material breach of this Agreement prior to the date of termination that gives rise to such termination, or constitute a waiver of any claim with respect thereto.

ARTICLE XI.

MISCELLANEOUS

- 11.1 Expenses of the Transaction. Each of the parties hereto agrees to pay such party's own fees and expenses in connection with this Agreement and the Related Agreements and the transactions contemplated hereby and by the Related Agreements including, without limitation, legal and accounting fees and expenses (it being understood that PVH shall pay all filings fees in connection with any filings to be made by either party under the HSR Act or any other filings with any Governmental Authority).
- 11.2 Notices. All notices or other communications required or permitted hereunder shall be in writing and shall be deemed given or delivered (i) when delivered personally or by private courier, (ii) when actually delivered by registered or certified United States mail, return receipt requested, or (iii) when sent by facsimile transmission (provided, that it is confirmed by a means specified in clause (i) or (ii)), addressed as follows:

If to PVH or to any CK Company (subsequent to the Closing) to:

Phillips-Van Heusen Corporation 200 Madison Avenue New York, New York 10016

Attention: Vice President, General Counsel and Secretary

Facsimile: (212) 381-3970 Telephone: (212) 381-3509

with a copy to:

Katten Muchin Zavis Rosenman 575 Madison Avenue New York, New York 10022

Attention: David H. Landau, Esq.

Facsimile: (212) 940-8776 Telephone: (212) 940-8800 If to any Seller to the addresses set forth in Section 11.2 of the Sellers Disclosure Schedule and if to any CK Company (prior to the Closing) to:

> Calvin Klein, Inc. 205 West 39th Street New York, New York 10018

Attention: Mr. Barry Schwartz, Chairman and CEO Facsimile: (212) 719-4591
Telephone: (212) 719-2600

in each case, with copies to:

Grubman, Indursky & Schindler, P.C.

152 West 57th Street New York, New York 10019

Attention: Arthur Indursky, Esq.

Facsimile: (212) 554-0444 Telephone: (212) 554-0400

and

Paul, Weiss, Rifkind, Wharton & Garrison 1285 Avenue of the Americas New York, New York 10019 Attention: James H. Schwab, Esq.

Facsimile: (212) 757-3990 Telephone: (212) 373-3000

or to such other address as such party may indicate by a notice delivered to the other parties hereto.

- 11.3 No Modification Except in Writing. This Agreement shall not be changed, modified, or amended except by a writing signed by the party to be affected by such change, modification or amendment, and this Agreement may not be discharged except by performance in accordance with its terms or by a writing signed by the party to which performance is to be rendered.
- 11.4 Entire Agreement. This Agreement together with the Schedules, Appendices and Exhibits hereto and the Confidentiality Agreement, sets forth the entire agreement and understanding among the parties as to the subject matter hereof and merges and supersedes all prior discussions, agreements and understandings of every kind and nature among them with respect to such subject matter.
- 11.5 Severability. If any provision of this Agreement or the application of any provision hereof to any person or circumstances is held invalid, the remainder of this Agreement

and the application of such provision to other persons or circumstances shall not be affected unless the provision held invalid shall substantially impair the benefits of the remaining portions of this Agreement.

- 11.6 Assignment. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective heirs, executors, administrators, successors and permitted assigns. This Agreement may not be assigned by any Seller or any CK Company except with the prior written consent of PVH; provided, however, Mr. Klein may assign all or part of his rights to receive the Design Services Purchase Payments. PVH may assign its rights, but not its obligations, under this Agreement to any Affiliate of PVH so long as said Affiliate agrees in writing provided to the Sellers that it will also be bound by and perform all obligations of PVH in this Agreement. Notwithstanding the foregoing, PVH may assign its rights under Section 6.3 to any Person in connection with the sale of all or substantially all of the assets of PVH or all or substantially all of the assets comprising the business of the CK Companies (whether conducted directly, or indirectly, by PVH) to such Person, so long as such Person agrees in writing reasonably acceptable to the Sellers that it will also be bound by and will perform all of the obligations of PVH in this Agreement and the Related Agreements.
 - 11.7 Governing Law; Jurisdiction.
- (a) This Agreement shall be governed by, and construed in accordance with, the Laws of the State of New York applicable to contracts made and to be performed wholly within said State, without giving effect to the conflict of laws principles thereof.
- (b) Each party to this Agreement irrevocably submits to the exclusive jurisdiction of (i) the courts of the State of New York located in New York County and (ii) the United States District Court for the Southern District of New York, for the purposes of any Action arising out of this Agreement or the transactions contemplated hereby.
- (c) Notwithstanding Sections 11.7(a) and (b), in the event of any dispute between PVH, on the one hand, and any Seller, on the other hand, prior to the commencement of any litigation proceeding, such parties may (but are not obligated to) select an independent mediator to assist in the mediation and resolution of such disputes. PVH and the Sellers with the assistance of such mediator, shall use reasonable best efforts to mediate their disputes within 60 days after the submission of the dispute to such mediator. If the parties are not able to successfully mediate such disputes within such 60-day period, the parties agree that the resolution of such dispute will be governed by Sections 11.7(a) and (b). Unless provided otherwise in this Agreement, the cost of such mediator shall be borne equally by the parties.
- 11.8 Specific Performance. The parties agree that if any of the provisions of this Agreement were not performed by PVH, on the one hand, or the Sellers and the CK Companies, on the other hand, in accordance with their specific terms or were otherwise breached by such parties, irreparable damage would occur, no adequate remedy at Law would exist and damages would be difficult to determine, and that the non-breaching party will be entitled to specific performance of the terms hereof. The parties waive any requirement for the posting of a bond in connection with any Action seeking specific performance; provided, however, that nothing

herein will affect the right of any of the parties to seek recovery against any party hereto, at Law, in equity or otherwise, with respect to any covenants, agreements or obligations to be performed by such party or parties after the Closing Date.

- 11.9 Headings; References. The headings appearing in this Agreement are inserted only as a matter of convenience and for reference and in no way define, limit or describe the scope and intent of this Agreement or any of the provisions hereof. Any reference in this Agreement (including in any Exhibit, Appendix or Schedule hereto) to a "Section," "Article," or "Exhibit" shall mean a Section, Article or Exhibit of or to this Agreement unless expressly stated otherwise.
- 11.10 Interpretation. In this Agreement, (a) words used herein regardless of the gender specifically used shall be deemed and construed to include any other gender, masculine, feminine or neuter, as the context shall require, and (b) all terms defined in the singular shall have the same meanings when used in the plural and vice versa. Any statute defined or referred to herein or in any agreement or instrument that is referred to herein means such statute as from time to time amended, modified or supplemented, including (in the case of statutes) by succession of comparable successor statutes. References to a Person are also its predecessors and permitted successors and assigns.
- 11.11 Schedules. Any item that is disclosed in any part of the CK Companies Disclosure Schedule, the Sellers Disclosure Schedule or of the PVH Disclosure Schedule with sufficient specificity so that it is apparent on the face of such Schedule that such disclosure is also applicable to one or more other parts of such Schedule shall also be deemed disclosed for purposes of such other parts of such Schedule to which such disclosure is applicable. Terms used in any Schedule to this Agreement and not otherwise defined in such Schedule shall have the meaning ascribed to such term in this Agreement.
- 11.12 Third Parties. Except as expressly provided herein or as otherwise agreed by the parties hereto, the provisions of this Agreement are solely for the benefit of the parties hereto and shall not inure to the benefit of any third party.
- 11.13 Counterparts. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original and all of which taken together shall constitute a single agreement. This Agreement will become binding when one or more counterparts taken together will have been executed and delivered (including by facsimile) by the parties.

IN WITNESS WHEREOF, each of the parties hereto has executed this Agreement on the day and year first above written.

PHILLIPS-VAN HEUSEN CORPORATION

By: /s/ Bruce J. Klatsky

Name: Bruce J. Klatsky

Title: Chairman and Chief Exec. Officer

CALVIN KLEIN, INC.

By: /s/ Barry Schwartz

Name: Barry Schwartz

Title: Chairman and Chief Exec. Officer

CALVIN KLEIN (EUROPE), INC.

By: /s/ Barry Schwartz

Name: Barry Schwartz

Title: Chairman and Chief Exec. Officer

CALVIN KLEIN (EUROPE II) CORP.

By: /s/ Barry Schwartz

Name: Barry Schwartz

Title: Chairman and Chief Exec. Officer

By: /s/ Barry Schwartz _____

Name: Barry Schwartz

Title: Chairman and Chief Exec. Officer

CK SERVICE CORP.

By: /s/ Barry Schwartz

Name: Barry Schwartz Title: Chairman and Chief Exec. Officer

TRUST F/B/O ISSUE OF CALVIN KLEIN

By: /s/ Flore Klein

Flore Klein, as trustee

By: /s/ Deirdre Miles-Graeter

Deirdre Miles-Graeter, as trustee

TRUST F/B/O ISSUE OF BARRY SCHWARTZ

By: /s/ Sheryl Rona Schwartz

Sheryl Rona Schwartz, as trustee

| /s/ Calvin Klein |
|--------------------------------|
| Calvin Klein |
| /s/ Barry Schwartz |
| Barry Schwartz |
| /s/ Stephanie Schwartz-Ferdman |
| Stephanie Schwartz-Ferdman |
| /s/ Jonathan Schwartz |
| Jonathan Schwartz |

APPENDIX A

DEFINITIONS

Definitions. The following terms when used in the Agreement shall have the respective meanings ascribed to them below:

"Accounting Firm" has the meaning ascribed to such term in Section 1.4(b).

"Action" shall mean any action, suit, claim, litigation, proceeding, arbitration, audit, investigation, or hearing (whether civil, criminal, administrative, investigative or informal) commenced, brought, conducted or heard by or before, any Governmental Authority.

"Adjustment Payment" has the meaning ascribed to such term in Section 1.4(c).

"Affiliate" shall mean, with respect to a specified Person, any other Person who, directly or indirectly, controls, is controlled by, or is under common control with such specified Person. As used in this definition, the term "control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through ownership of voting securities, by contract or otherwise.

"Agreement" has the meaning ascribed to such term in the Preamble hereto.

"Agreement and Assignment" has the meaning ascribed to such term in Section 7.11.

"Apax Transaction" shall mean the transaction contemplated by the Securities Purchase Agreement among PVH, Lehman Brothers Inc. and the Investors named therein dated as of December 16, 2002.

"Audited Financial Statements" has the meaning ascribed to such term in Section 3.14(a).

"Balance Sheet Rules" shall mean the rules identified on Schedule IV.

"Base Amount" shall mean \$20,000,000.

"Basket" has the meaning ascribed to such term in Section $9.4\,(a)$.

"Bearing the Marks" has the meaning ascribed to such term in Section $1.5\,\mathrm{(a)}$.

"Business Day" shall mean a day (other than a Saturday or Sunday), on which commercial banks are open for business in New York, New York.

"CKI" has the meaning ascribed to such term in the Preamble hereto.

"CK Companies" has the meaning ascribed to such term in the Preamble hereto and, for purposes of this Agreement, shall include any division of any CK Company, whether or not such division is a separate entity including, without limitation, CRK Advertising.

"CK Companies Disclosure Schedule" shall mean that certain schedule attached hereto as Schedule I qualifying the representations and warranties contained in Article III.

"CK Companies Material Adverse Effect" shall mean any event, condition or contingency that has had, or is reasonably likely to have, a material adverse effect on the business, assets, liabilities, results of operations or financial condition of the CK Companies, taken as a whole. For the purposes of this Agreement, a CK Companies Material Adverse Effect shall not include any event, condition or contingency, or the effect thereof, resulting from (i) the entering into of any transaction consented to by PVH pursuant to Section 5.1(b), (ii) the announcement, negotiation, contemplation of Closing or actions taken in furtherance, of the transactions contemplated herein, (iii) events of war impacting the economy in general, and (iv) changes in general economic conditions or in the fashion, apparel, accessories, home furnishings or fragrance industries, in general.

"CK Debt Obligations" has the meaning ascribed to such term in Section 1.6.

"CK Europe I" has the meaning ascribed to such term in the Preamble hereto.

"CK Europe II" has the meaning ascribed to such term in the Preamble hereto.

"CK Intellectual Property Rights" has the meaning ascribed to such term in Section 3.12(a).

"CK Italy" has the meaning ascribed to such term in the Preamble hereto.

"CK Trust" shall mean the Calvin Klein Trademark Trust, provided, that all references herein to the CK Trust shall only be with respect to the interests in such owned by CKI and shall exclude for all purposes of this Agreement (including Article III), the Class B and C ownership interests of Warnaco Inc. or any action taken by or on behalf of Warnaco Inc. with respect to the CK Trust or its interests therein.

"CK Trust Agreement" shall mean the trust agreement established on March 14, 1994, as amended, by and between CKI, as settlor, and Wilmington Trust Company, as trustee, which trust agreement governs the CK Trust.

"CK Umbrella" has the meaning ascribed to such term in Section $1.5\,\mathrm{(a)}$.

"CKI Plan" has the meaning ascribed to such term in Section 6.2(d).

"Cap" has the meaning ascribed to such term in Section 9.4(a).

"Capital Stock" shall mean, with respect to any Person, any and all shares, interests, participations, rights in or other equivalents (however designated and whether voting or non-voting) of such Person's capital stock or any form of membership, ownership or participation interests, as applicable, including partnership interests, whether now outstanding or hereafter issued and any and all securities, debt instruments, rights, warrants or options exercisable or exchangeable for or convertible into such capital stock.

"Cash Purchase Price" has the meaning ascribed to such term in Section 1.2.

"Certificates of Exemption" has the meaning ascribed to such term in Section $6.4\,(\mathrm{h})$.

"Channel" has the meaning ascribed to such term in Section 1.5(a).

"Claim Notice" has the meaning ascribed to such term in Section 9.7.

"Closing" has the meaning ascribed to such term in Section 1.1.

"Closing Date" has the meaning ascribed to such term in Section 1.1.

"Closing Date Balance Sheet" has the meaning ascribed to such term in Section $1.4\,\mathrm{(a)}$.

"Closing Date Net Book Value" shall mean the number obtained by subtracting (a) the sum of the liabilities, including deferred revenues, of the CK Companies set forth on the Closing Date Balance Sheet, from (b) the sum of the total assets of the CK Companies set forth on the Closing Date Balance Sheet. The Closing Date Net Book Value shall be calculated in accordance with GAAP.

"Closing Time" has the meaning ascribed to such term in Section 1.1.

"Code" shall mean the Internal Revenue Code of 1986, as amended from time to time, and any successor statute thereto and all final or temporary regulations promulgated thereunder and published.

"Company Employee Plan" has the meaning ascribed to such term in Section $4.14\,(a)$.

"Company Plans" has the meaning ascribed to such term in Section 3.20(d).

"Confidential Information" shall mean trade secrets, confidential or proprietary information, knowledge, or know-how pertaining primarily to the business of any CK Company or any confidential or proprietary information concerning any supplier or customer of any CK Company, including, without limitation, customer lists, research and development information and materials, inventions, formulas, methods, techniques, processes, plans, product designs, material designs, fabric designs, procedures, contracts, financial information and computer models. The term Confidential Information shall not include (i) information that is generally available to the public or within the apparel and fashion industry, other than as a result of a disclosure by the receiving party or its directors, officers, stockholders, partners, Affiliates, employees, agents or advisors in violation of this Agreement; (ii) information which, prior to disclosure to the receiving party by or on behalf of the disclosing party, was already in the receiving party's possession on a non-confidential basis; (iii) information that was developed without the use of Confidential Information; (iv) information that becomes available to the receiving party on a non-confidential basis from a source other than any CK Company or any of its advisors, agents or Affiliates, provided, that such source is not known by the receiving party to be bound by a confidentiality agreement with or other obligation of secrecy to such CK Company or any other party; (v) information which is reasonably necessary for the purpose of

the disclosing party asserting its rights in a dispute among the parties hereunder or under any Related Agreement; or (vi) information reasonably related to any Tax Returns or similar matters required to be prepared by the disclosing party or any of their representatives and filed with any Governmental Authority, provided, that, with respect to Confidential Information disclosed as a result of or in connection with clause (vi) herein, the disclosing party shall request confidential treatment of such Confidential Information.

"Confidentiality Agreement" shall mean that certain confidentiality agreement between PVH and CKI dated March 6, 2002.

"Consulting Agreement" has the meaning ascribed to such term in Section 7.10.

"Contracts" shall mean all legally binding leases, including, without limitation, Real Property Leases, licenses, contracts, agreements, indentures, promissory notes, guarantees, arrangements, commitments and understandings of any kind, whether written or oral, to which any CK Company or the CK Trust is a party or by which any CK Company or the CK Trust or any of the assets of any CK Company or the CK Trust may be bound, and all rights arising under any of them.

"Damages" shall mean losses, obligations, liabilities, settlement payments, awards, judgments, fines, penalties, damages, deficiencies, Taxes and reasonable expenses and costs, including reasonable attorneys' and auditors' fees (and any reasonable experts' fees) and court costs, in each case as determined by an agreement of the parties or a court of competent jurisdiction.

"Design Services Agreement" shall mean the letter agreement dated July 25, 1994 relating to the design services provided by Mr. Klein to CKI.

"Design Services Purchase Payment Default" has the meaning ascribed to such term in Section $1.5\left(f\right)$.

"Design Services Purchase Payments" has the meaning ascribed to such term in Section $1.5\,(\mathrm{a})$.

"Design Services Rate" has the meaning ascribed to such term in Section 1.5 of the Sellers Disclosure Schedule.

"Design Services Security Agreement" shall mean the pledge and security agreement to be entered into at Closing between PVH, the CK Companies (other than CK Italy) (each as constituted after the Closing) and Mr. Klein in the form attached hereto as Exhibit I.

"DOJ" shall mean the United States Department of Justice.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended from time to time, and any successor statute thereto and all final or temporary regulations promulgated thereunder.

"Employee Benefit Plans" has the meaning ascribed to such term in Section $3.20\,(a)$.

"Employee Program" shall mean any "employee benefit plan", within the meaning of Section 3(3) of ERISA, whether or not it is subject to ERISA.

"Environment" shall mean soil, surface waters, ground waters, land, stream, sediments, surface or subsurface strata and ambient air.

"Environmental Condition" shall mean any condition with respect to the Environment on or off any Facility caused by a Release of Hazardous Substances or violation of Environmental Laws, whether or not yet discovered, which could or does result in any Damages, including, without limitation, any condition resulting from the operation of the business of any CK Company or the operation of the business of any subtenant or occupant of any Facility or that of other property owners or operators in the vicinity of any Facility or any activity or operation formerly conducted by any person or entity on or off such Facility.

"Environmental Laws" shall mean all Laws relating to the pollution of or protection of the Environment, from contamination by, or relating to injury to, or the protection of, real or personal property or human health or the Environment, including, without limitation, all valid and lawful requirements of courts and other Governmental Authorities pertaining to reporting, licensing, permitting, investigation, remediation and removal of, emissions, discharges, releases or threatened releases of Hazardous Substances, chemical substances, pesticides, petroleum or petroleum products, pollutants, contaminants or hazardous or toxic substances, materials or wastes, into the Environment, or relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of Hazardous Substances, pollutants, contaminants or hazardous or toxic substances, materials or wastes.

"Environmental Report" shall mean any report, study, assessment, audit, or other similar document that addresses any issue of actual or potential noncompliance with, actual or potential liability under or cost arising out of, or actual or potential impact on business in connection with, any Environmental Law or any proposed or anticipated change in or addition to any Environmental Law.

"Escrow Agreement" has the meaning ascribed to such term in Section $1.3(b)\,.$

"Estimated Closing Date Net Book Value" has the meaning ascribed to such term in Section 1.4.

"Exchange Act" shall mean the U.S. Securities Exchange Act of 1934, as amended, or any successor law, and regulations and rules issued under that Act or any successor law.

"FTC" shall mean the United States Federal Trade Commission. ---

"Facility" shall mean any facility that is now or has heretofore been owned, leased or used in connection with the business of any CK Company.

"Final Quarter" has the meaning ascribed to such term in Section $1.5\,\mbox{(a)}\,.$

"Financial Statements" shall mean the Audited Financial Statements and the Unaudited Financial Statements.

"GAAP" shall mean United States generally accepted accounting principles, and for purposes of Section 1.4 shall include the Balance Sheet Rules.

"Governmental Authority" shall mean any (i) federal, state, local, provincial, municipal, foreign, or other government, (ii) governmental or quasi-governmental authority of any nature or (iii) other body exercising any statutory, administrative, judicial, arbitrative, legislative, police, regulatory, or taxing authority or power.

"Governmental Permits" shall mean all licenses, franchises, registrations, permits, privileges, immunities, approvals and other authorizations from a Governmental Authority.

"Group" has the meaning ascribed to such term in Section 3.20(a).

"Guarantee" shall mean the pledgor guarantee in favor of Mr. Klein executed by the CK Companies (other than CK Italy) (each as constituted after the Closing) as of the Closing Time in the form attached hereto as Exhibit L.

- (i) the presence of which requires notification, investigation, or remediation under any Environmental Law;
- (ii) which is or becomes defined as "toxic", a "hazardous waste",
 "hazardous material" or "hazardous substance" or "pollutant" or
 "contaminant" under any present Environmental Laws;
- (iii) which is toxic, explosive, corrosive, flammable, infectious, radioactive, carcinogenic, mutagenic or otherwise hazardous and is regulated by any Governmental Authority;
- (iv) which contains gasoline, diesel fuel or other petroleum hydrocarbons or volatile organic compounds;
- (v) which contains polychlorinated byphenyls (PCBs) or asbestos or urea formaldehyde foam insulation; or
- $% \left(v_{1}\right) =0$ which contains or emits radioactive particles, waves or materials, including radon gas.

"Foreign Plan" shall mean any plan which is described in Section 4(b)(4) of ERISA, and which is maintained, sponsored or contributed to by, or covers any employee of, PVH or any of its Subsidiaries

"HSR Act" shall mean the Hart Scott Rodino Antitrust Improvements Act of 1976, as amended.

"HSR Clearance" shall mean the expiration or early termination of any applicable waiting period under the HSR Act, without any suit to enjoin the transactions contemplated herein having been filed by the DOJ or FTC under Title 15 of the United States Code.

"HSR Date" has the meaning ascribed to such term in Section 1.1.

"Holdback Shares" shall mean such number of shares of PVH Common Stock equal to the quotient of (i) \$15,000,000\$ divided by (ii) the PVH Common Stock Closing Price (rounded up to the nearest whole number).

- "Indemnitee" has the meaning ascribed to such term in Section 9.7.
- "Indemnitor" has the meaning ascribed to such term in Section 9.7.
- "Initial Quarter" has the meaning ascribed to such term in Section 1.5(a).
- "Intellectual Property Rights" has the meaning ascribed to such term in Section 3.12(a).

"Intercreditor Agreement" shall mean the intercreditor and subordination agreement by and among the Working Capital Agent, Term Lender (each term as defined therein) and Mr. Klein to be entered into at Closing in the form attached hereto as Exhibit K.

"International Employees" has the meaning ascribed to such term in Section 3.10(d).

- "International Plans" has the meaning ascribed in Section 3.20(a).
- "J. Schwartz" has the meaning ascribed to such term in the Preamble hereto.
- "Klein Group Sellers" shall mean Mr. Klein and the Klein Trust.
- "Klein Heirs" has the meaning ascribed to such term in Section 1.5(a).
- "Klein Trust" has the meaning ascribed to such term in the Preamble hereto.

"Knowledge" shall mean, with respect to the Sellers, the actual knowledge, or the knowledge that would be expected to have been obtained upon reasonable inquiry by any of the Sellers, and with respect to the CK Companies, the actual knowledge, or the knowledge that would be expected to have been obtained upon reasonable inquiry of Mr. Klein, Mr. Schwartz, Tom Murry, Len LaSalandra, Kathy Melita, Pamela Bradford, Deirdre Miles-Graeter, Max Weinstein, Kim Vernon, William Baer, William Voight and Mark Katz.

"Law" shall mean any constitution, law, treaty, compact, directive, ordinance, principal of common law, permit, authorization, variance, regulation, rule, or statute, including, without limitation, all federal, foreign, international, state and local laws related to Taxes, ERISA, Hazardous Substances and the Environment, zoning and land use, intellectual property, privacy,

occupational safety and health, consumer protection, product quality, safety, employment and labor matters.

"Licensee Sales" has the meaning ascribed to such term in Section 1.5(a).

"Liens" shall mean all mortgages, pledges, liens, security interests, conditional sale agreements, encumbrances or similar restrictions.

"Line Letter" shall mean the Line Letter between CKI and JPMorgan Chase Bank dated March 14, 2002.

"Liquidated Damages Amount" shall mean the amount of liquidated damages for the year in which any Design Services Purchase Payment Default occurs determined pursuant to the formula set forth on Schedule V.

"Marks" shall mean the trademarks and service marks CALVIN KLEIN, CK/CALVIN KLEIN and CK, and all variations and combinations thereof throughout the entire world in connection with all products and services other than to the extent a trademark or service mark is subject to the Class B and Class C interest in the CK Trust, together with all related recordations, applications for registration and registrations, and all goodwill associated therewith. Notwithstanding the foregoing, for purposes of Section 1.5, "Marks" shall have the meaning ascribed thereto in the Agreement and Assignment.

"Mr. Klein" has the meaning ascribed to such term in the Preamble hereto.

"Mr. Schwartz" has the meaning ascribed to such term in the Preamble hereto.

"Multiemployer Plan" has the meaning ascribed to such term in Section 3.20(e).

"Net Book Value Overage" has the meaning ascribed to such term in Section 1.4(a).

"Net Book Value Underage" has the meaning ascribed to such term in Section 1.4(a).

"New Corporate Name" has the meaning ascribed to such term in Section $1.5\,(\mathrm{a})$.

"Notarial Deed" has the meaning ascribed to such term in Section 7.12.

"Notice of Disagreement" has the meaning ascribed to such term in Section $1.4\,\mathrm{(b)}$.

"NYSE" shall mean the New York Stock Exchange.

"Off-Balance Sheet Arrangement" shall mean any transaction, agreement or other contractual arrangement to which an entity that is not consolidated with PVH is a party, under which PVH, whether or not a party to the arrangement, has, or in the future may have: (i) any obligation under a direct or indirect guarantee or similar arrangement; (ii) a retained or contingent interest in assets transferred to an unconsolidated entity or similar arrangement; (iii) derivatives, to the extent that the fair value thereof is not fully reflected as a liability or asset in the financial statements of PVH; or (iv) any obligation or liability in excess of \$5,000,000,

including a contingent obligation or liability, not incurred in the ordinary course of business to the extent that such obligation or liability is not fully reflected in the financial statements of PVH (excluding the footnotes thereto).

"Order" shall mean any award, decision, injunction, decree, stipulation, determination, writ, judgment, order, ruling, or verdict ordered, issued, made or rendered by any court, administrative agency or other Governmental Authority.

"Payment Period" has the meaning ascribed to such term in Section $1.5\ \mathrm{of}$ the Sellers Disclosure Schedule.

"Payment Statement" has the meaning ascribed to such term in Section $1.5\,\mathrm{(b)}$.

"PBGC" shall mean the Pension Benefit Guaranty Corporation.

"Permitted Arrangements" has the meaning ascribed to such term in Section $3.11\,(a)$.

"Permitted Liens" shall mean Liens (i) for any current taxes or assessments not yet delinquent, (ii) created by statute of carriers, warehousemen, vendors, mechanics, laborers or materialmen incurred in the ordinary course of business for sums not yet due or (iii) real and personal property leases and licenses for the use of CK Intellectual Property Rights.

"Person" shall mean any individual, firm, unincorporated organization, corporation (including any not-for-profit corporation), general or limited partnership, limited liability company, cooperative marketing association, joint venture, estate, trust, association or other entity as well as any syndicate or group that would be deemed to be a person under Section 13(a)(3) of the Exchange Act.

"Pre-Closing Tax Periods" has the meaning ascribed to such term in Section $6.4\,(\mathrm{a})$.

"Purchase Price" has the meaning ascribed to such term in Section 1.2.

"Purchase Shares" has the meaning ascribed to such term in the Recitals hereto.

"PVH" has the meaning ascribed to such term in the Preamble hereto.

"PVH Balance Sheet" has the meaning ascribed to such term in Section 4.11.

"PVH Common Stock Closing Price" shall mean the average closing sales price on the New York Stock Exchange Composite Tape (as reported in The Wall Street Journal) for a share of PVH Common Stock for the ten Business Day period ending on the Business Day immediately prior to the Closing Date.

"PVH Channel" has the meaning ascribed to such term in Section 1.5(a).

"PVH Disclosure Schedule" shall mean that certain schedule attached hereto as Schedule II qualifying the representations and warranties contained in $Article\ IV$.

"PVH Group Member" shall mean each of PVH and its Affiliates (including, each CK Company as constituted after the Closing) and their respective directors, officers, employees, agents and attorneys and their respective successors and assigns.

"PVH Intellectual Property" shall have the meaning ascribed to such term in Section $4.13\,(a)$.

"PVH Marks" shall mean all of the trademarks, service marks and trade names throughout the entire world licensed to or used by PVH or any of its Subsidiaries and/or owned by PVH or any of its Subsidiaries including, but not limited to, Van Heusen, Bass, G.H., Bass & Co., Izod, cKCalvin Klein, Geoffrey Beene, Arrow, DKNY, Kenneth Cole New York and Reaction by Kenneth Cole.

"PVH Material Adverse Effect" shall mean any event, condition or contingency that has had, or is reasonably likely to have, a material adverse effect on the business, assets, liabilities (including contingent liabilities), results of operations or financial condition or, to the knowledge of PVH, prospects of PVH and its Subsidiaries, taken as a whole. For the purposes of this Agreement, a PVH Material Adverse Effect shall not be deemed to arise by reason of (i) the transactions contemplated hereby and by the Apax Transaction, (ii) events of war impacting the economy in general, and (iii) changes in general economic conditions or in the fashion, apparel, accessories, home furnishings or fragrance industries, in general.

"PVH Savings Plan" has the meaning ascribed to such term in Section 6.2(d).

"PVH Shares" has the meaning ascribed to such term in Section 1.2.

"Qualified Plan" has the meaning ascribed to such term in Section 3.20(d).

"Real Property Leases" shall mean all leases for real property to which any CK Company is a party or by which it is bound.

"Registration Rights Agreement" has the meaning ascribed to such term in Section 8.13.

"Related Agreements" shall mean (i) the Consulting Agreement, the Agreement and Assignment and the Registration Rights Agreement, and (ii) those other agreements and documents entered into or delivered between PVH, the CK Companies and the Sellers related to, ancillary to, or in connection with this Agreement or the documents listed in clause (i) hereof.

"Restricted Business" has the meaning ascribed to such term in Section $6.3\,(a)$.

"Retail Direct Sales" has the meaning ascribed to such term in Section $1.5\,\mathrm{(a)}$.

"Right" shall have the meaning set forth in the Rights Agreement.

"Rights Agreement" shall mean the Rights Agreement, dated as of June 10, 1986, as amended, by and between the Company and The Bank of New York (successor to The Chase Manhattan Bank, N.A.), as Rights Agent, and each amendment and extension thereof.

- "Rights Holder" has the meaning ascribed to such term in Section 6.4(h).
- "Schwartz Design Services Form" has the meaning ascribed to such term in Section 7.17.
- "Schwartz Group Sellers" shall mean Mr. Schwartz, the Schwartz Trust, Schwartz-Ferdman and J. Schwartz.
- "Schwartz Trust" has the meaning ascribed to such term in the Preamble hereto.
- "Schwartz-Ferdman" has the meaning ascribed to such term in the Preamble hereto.
 - "SEC Reports" has the meaning ascribed to such term in Section 4.7.
- "Securities Act" shall mean the Securities Act of 1933, as amended, or any successor law, and regulations and rules issued under that Act or any successor law.
- "Seller Group Member" shall mean the Sellers and each of their respective agents and attorneys and their respective successors and permitted assigns.
 - "Sellers" has the meaning ascribed to such term in the Preamble hereto.
- "Sellers Disclosure Schedule" shall mean that certain schedule attached hereto as Schedule III qualifying the representations and warranties contained in Article II.
- "Sellers' Representative" shall mean Mr. Klein, for the Klein Group Sellers, and Mr. Schwartz for the Schwartz Group Sellers, or such other persons as the Sellers from time to time designated by written notice to PVH.
- "September 28 Balance Sheet" shall mean the combined balance sheets of CKI and its affiliates for the period ended September 28, 2002.
- "Service Corp." has the meaning ascribed to such term in the Preamble hereto.
- "Servicing Agreement" shall mean the servicing agreement dated as of March 14, 1994 between CKI and the CK Trust.
- "Severance Qualified Employee" has the meaning ascribed to such term in Section $6.2\,\mathrm{(a)}$.
 - "Stock Certificates" has the meaning ascribed to such term in Section 1.2.
 - "Straddle Periods" has the meaning ascribed to such term in Section 6.4(b).
- "2001 Balance Sheet" shall mean the combined balance sheets of CKI and its affiliates for the fiscal year ended December 29, 2001.
- "Subsidiary" shall mean, as to any Person, any other Person more than 50% of the shares of the voting stock or other voting interests of which are owned or controlled, or the ability to select or elect more than 50% of the directors or similar managers is held, directly or indirectly,

by such first Person or one or more of its Subsidiaries or by such first Person and one or more of its Subsidiaries.

"Taxes" shall mean any and all federal, state, local and foreign taxes, assessments, governmental charges, duties, impositions and liabilities, including, but not limited to, income (whether net or gross), excise, property, sales, transfer, gains, gross receipts, occupation, privilege, payroll, wage, unemployment, workers' compensation, social security, escheat, use, value added, capital, gross receipts, franchise, license, severance, stamp, premium, windfall profits, environmental (including taxes under Code Sec. 59A), capital stock, profits, withholding, disability, real property, personal property, registration, customs duties, alternative or add-on minimum, estimated or other tax of any kind whatsoever (whether disputed or not) imposed by any Tax Authority, including any related charges, fees, interest, penalties, additions to tax or other assessments.

"Tax Authority" shall mean any foreign or domestic government, any subdivision, agency, commission or authority thereof, or any quasi-governmental body or other authority exercising any taxing or Tax regulatory authority.

"Tax Returns" shall mean all returns, reports, declarations, estimates, information returns and statements (including any related or supporting information) filed or to be filed with any Tax Authority in connection with the determination, assessment, collection or administration of any Taxes.

"Termination Date" has the meaning ascribed to such term in Section $10.1(\mbox{d})$.

"Unaudited Financial Statements" has the meaning ascribed to such term in Section $3.14\,(\mathrm{a})$.

"Union Plans" has the meaning ascribed to such term in Section 3.20(d).

"WARN" shall mean the Worker Adjustment and Retraining Notification ${\tt Act}$ of 1988, as amended.

"Warrant" shall mean the warrant exercisable for an aggregate of 320,000 shares of PVH's common stock, par value \$1.00 per share (the "PVH Common Stock") at an exercise price equal to \$28 per share. The Warrant shall be issued to Mr. Klein, exercisable for a nine-year period from the Closing Date, and shall be in the form attached hereto as Exhibit J.

"Wholesale Direct Sales" has the meaning ascribed to such term in Section $1.5\,\mathrm{(a)}$.

SECURITIES PURCHASE AGREEMENT

AMONG

PHILLIPS-VAN HEUSEN CORPORATION,

LEHMAN BROTHERS INC.

AND

THE INVESTORS

DATED AS OF DECEMBER 16, 2002

TABLE OF CONTENTS

| | | PAGE |
|---|--|----------------|
| | | |
| ARTICLE I Defini | tions | 1 |
| Section 1.01. Section 1.02. | Definitions Rules of Construction | 1 10 |
| ARTICLE II issua | nce, Sale and Purchase of the Series B Stock | 11 |
| Section 2.01. Section 2.02. Section 2.03. | Sale and Purchase of the Series B Stock | 11 11 12 |
| ARTICLE III Repr | esentations and Warranties of the Company | 12 |
| Section 3.01. Section 3.02. Section 3.03. | Organization and Good Standing Authority; Binding Effect Organization and Good Standing of Company | 12 12 |
| Section 3.04. Section 3.05. | Subsidiaries Capitalization No Violations; Consents | 13 13 14 |
| Section 3.06. Section 3.07. | Listing Financial Statements | 15 15 |
| Section 3.08. Section 3.09. | Commission Filings | 16 16 |
| Section 3.10. Section 3.11. | Litigation Tax Matters | 18 19 |
| Section 3.12. Section 3.13. | Real Property Inventory | 19 20 |
| Section 3.14. Section 3.15. | Intellectual Property | 20 20 |
| Section 3.16. Section 3.17. | Company Employee Plans | 21 24 |
| Section 3.18. Section 3.19. | Compliance with Laws; Permits Preferred Stock Exemption | 25 25 |
| Section 3.20. Section 3.21. | Environmental Protection | 26 26 |
| Section 3.21. Section 3.22. Section 3.23. | Transactions with Affiliates | 26 26 |
| Section 3.24. | Customers | 27 |
| Section 3.25. Section 3.26. | Financial Advisors DGCL Section 203 and Rights Agreement | 27 27 |
| Section 3.27. Section 3.28. | D&O Insurance | 27 27 |

| Section | 3.29. | Full Disclosure | 27 |
|------------|-----------|--|----|
| Section | 3.30. | No General Solicitation | 27 |
| Section | 3.31. | No Integration | 28 |
| | | | |
| ARTICLE IV | Repre | sentations and Warranties of the | |
| | - | initial purchaser and the Investors | 28 |
| | | • | |
| Section | 4.01. | Initial Purchaser Representations | 28 |
| Section | 4.02. | Investors Representations | 28 |
| | | - | |
| ARTICLE V | Covena | nts of the Company | 30 |
| | | | |
| Section | 5.01. | Accounting System | 30 |
| Section | 5.02. | Taxes | 31 |
| Section | 5.03. | Corporate Existence | 31 |
| Section | | Maintenance of Properties | 31 |
| Section | | Insurance | 31 |
| Section | | Compliance with Law | 31 |
| Section | | Maintain Listing | 31 |
| Section | | Secure Listing | 32 |
| Section | | Transfer Taxes | 32 |
| Section | 3.03. | italistet taxes | 22 |
| ARTICLE VI | 7 CTT T (| NS PRIOR TO CLOSING | 32 |
| AKIICLE VI | ACTIO | NS PRIOR TO CLOSING | 32 |
| Section | 6 01 | Access to Information | 32 |
| Section | | Conduct of Business | 33 |
| | | | |
| Section | | No Solicitation | 34 |
| Section | | Consent | 34 |
| Section | | HSR | 34 |
| Section | | Notice of Breach | 35 |
| Section | | Other Transaction Documents | 35 |
| Section | 6.08. | Publicity | 35 |
| | | | |
| ARTICLE VI | I cond | itions to Closing | 36 |
| _ | | | |
| Section | 7.01. | Conditions to Obligations of the Initial | |
| | | Purchaser and the Investors | 36 |
| Section | 7.02. | Conditions to Obligations of the Company | 39 |
| | | | |
| ARTICLE VI | II sur | vival | 39 |
| | | | |
| Section | 8.01. | Survival | 39 |
| | | | |
| ARTICLE IX | Indem | nification | 40 |
| | | | |
| Section | 9.01. | Generally | 40 |
| Section | 9.02. | Indemnification Procedure | 41 |
| Section | 9.03. | Certain Qualifications | 42 |
| Section | 9.04. | Limitations on Indemnification | 42 |

| ARTICLE X FE | ES, Expenses and Costs 43 |
|--|----------------------------|
| Section 10 | .01. Reimbursement |
| ARTICLE XI T | ERMINATION |
| Section 11 Section 11 Section 11 Section 11 | .02. Effect Of Termination |
| ARTICLE XII | Miscellaneous45 |
| Section 12 Section 12 Section 12 Section 12 Section 12 Section 12 | .02. Captions |
| Section 12 Section 12 Section 12 Section 12 Section 12 Section 12 Section 12 | Parties in Interest |

SECURITIES PURCHASE AGREEMENT

THIS SECURITIES PURCHASE AGREEMENT, dated as of December 16, 2002 (this "Agreement"), is by and among Phillips-Van Heusen Corporation, a Delaware corporation (the "Company"), Lehman Brothers Inc., as initial purchaser (the "Initial Purchaser") and each investor executing a signature page hereto (each an "Investor" and collectively, the "Investors").

RECITALS:

- A. The Company, on or prior to closing will, authorize a new series of Preferred Stock designated the "Series B Convertible Preferred Stock" (the "Series B Stock"), which will be convertible into shares of Common Stock in accordance with the terms of the Certificate of Designations governing the Series B Stock, in the form attached hereto as Exhibit A (the "Certificate of Designations").
- B. The Initial Purchaser desires to purchase from the Company, and the Company desires to sell to the Investors, upon the terms and subject to the conditions of this Agreement, the Series B Shares.
- C. The Investors have agreed to purchase from the Initial Purchaser, and the Initial Purchaser has agreed to sell to the Investors, subject to the terms and conditions of this Agreement, the Series B Shares.
- $\ensuremath{\text{D.}}$ The Company, the Initial Purchaser and the Investors desire to set forth certain agreements herein.

AGREEMENT:

NOW, THEREFORE, in consideration of the foregoing and of the respective covenants and undertakings hereunder and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, intending to be legally bound, the parties hereto do hereby agree as follows:

ARTICLE I

Section 1.01. Definitions. As used in this Agreement, the following terms have the meanings set forth below.

"AAA" shall have the meaning set forth in Section 9.02(b).

"Accredited Investor" shall mean any Person that is an "accredited investor" within the definition contained in Rule 501(a) under the Securities Act.

"Affiliate" shall mean (a) with respect to an individual, any member of such individual's family residing in the same household; (b) with respect to an entity: (i) any executive officer, director, partner or Person that owns ten percent (10%) or more of the outstanding beneficial interest of or in such entity, or (ii) any brother, sister, brother-in-law, sister-in-law, lineal descendant or ancestor of any executive officer, director, partner or Person that owns ten percent (10%) or more of the outstanding beneficial interest of or in such entity; and (c) with respect to a Person, any Person which directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with such Person or entity; provided, however, that for purposes of the definition of "Affiliate," no Investor shall be deemed an "Affiliate" of the Company.

"Agreement" shall have the meaning set forth in the preamble.

"Audited Financial Statements" shall have the meaning set forth in Section 3.07.

"Balance Sheet" shall have the meaning set forth in Section 3.07.

"Basket" shall have the meaning set forth in Section 9.04(a).

"Break-up Fee" shall mean any "break-up fee," "termination fee," or similar fee that the Company will receive pursuant to the CK Purchase Agreement or that certain letter agreement, dated September 26, 2002, between the Company and CKI reduced by the amount that the Company's Expenses exceed the amount by which the Company has been, or will be, reimbursed by CKI or any of its Affiliates for such expenses.

"Board of Directors" shall mean the Board of Directors of the Company.

"Business Day" shall mean any day other than (i) a Saturday, (ii) a Sunday or (iii) any other day on which banks in the City of New York are authorized or required to close.

"By-Laws" shall mean, when used with respect to a specified Person, the by-laws of a Person, as the same may be amended from time to time.

"Capital Stock" shall mean, with respect to any Person, any and all shares, interests, participations, rights in or other equivalents (however designated and whether voting or non-voting) of such Person's capital stock or any form of membership, ownership or participation interests, as applicable, including partnership interests, whether now outstanding or hereafter issued and any and all securities, debt instruments, rights, warrants or options exercisable or exchangeable for or convertible into such capital stock.

"Certificate of Designations" shall have the meaning set forth in the recitals.

"Certificate of Incorporation" shall mean, when used with respect to a specified Person, the Articles or Certificate of Incorporation or other applicable organizational document of such Person, as currently in effect.

"CK Acquisition" shall mean the acquisition by the Company of all of the Capital Stock of the CK Companies pursuant to the CK Purchase Agreement.

"CK Recovery" shall mean any amounts received by the Company or any Company Subsidiary paid by the CK Companies or Sellers (or any insurer thereof) by reason of the failure of the closing pursuant to the CK Purchase Agreement in excess of (i) the Company Expenses, (ii) Fees and Expenses of the Investors paid by the Company, and (iii) any out-of-pocket third party expense incurred by the Company in obtaining the CK Recovery (i.e., legal fees in any action by the Company against Sellers).

"CK Companies" shall have the meaning set forth in the CK Purchase Agreement.

"CK Companies Material Adverse Effect" shall mean any event, condition or contingency that has had, or is reasonably likely to have, a material adverse effect on the business, assets, liabilities, results of operations or financial condition of the CK Companies, taken as a whole. For the purposes of this Agreement, a CK Companies Material Adverse Effect shall not include any event, condition or contingency, or the effect thereof, resulting from (i) the entering into of any transaction consented to by PVH pursuant to Section 5.1(b) of the CK Purchase Agreement, (ii) the announcement, negotiation, or contemplation of the closing of the transactions contemplated by the CK Purchase Agreement or any actions taken in furtherance of the transactions contemplated by the CK Purchase Agreement, (iii) events of war impacting the economy in general, and (iv) changes in general economic conditions or in the fashion, apparel, accessories, home furnishings or fragrance industries, in general.

"CK Purchase Agreement" shall mean that certain Stock Purchase Agreement, by the Company, each CK Company, Mr. Calvin Klein, Mr. Barry Schwartz, Trust for the Benefit of the Issue of Calvin Klein, Trust for the Benefit of the Issue of Barry Schwartz, Stephanie Schwartz-Ferdman and Jonathan Schwartz.

"CK Transaction Documents" shall mean the CK Purchase Agreement and the Related Agreements (as such term is defined in the CK Purchase Agreement).

"CKI" shall mean Calvin Klein, Inc., a New York corporation.

"Closing" shall have the meaning set forth in Section 2.02(a).

"Closing Date" shall have the meaning set forth in Section $2.02\,(a)$.

"Code" shall mean the Internal Revenue Code of 1986, as amended, and the rules and regulations promulgated thereunder.

"Commission Filings" shall have the meaning set forth in Section 3.08.

"Common Stock" shall mean the common stock, \$1 par value per share, of the Company, including any associated Right.

"Company" shall have the meaning set forth in the preamble.

"Company Employee Plan" shall have the meaning set forth in Section $3.16\,(a)$.

"Company's Expenses" shall mean all of the fees and out-of-pocket expenses incurred by the Company in connection with the Contemplated Transactions.

"Company Group Member" shall mean each of the Company and its Affiliates and their respective directors, officers, employees, agents and attorneys and their respective successors and assigns.

"Company Subsidiary" shall have the meaning set forth in Section 3.03.

"Competing Offer" shall mean any offer or proposal for, or indication of interest in, acquiring more than 10% of the Capital Stock or other equity securities or securities convertible into or exchangeable for equity securities of the Company or any of the Company Subsidiaries or any debt financing (other than as contemplated by the CK Transaction Documents) intended to replace the amounts committed pursuant to this Agreement.

"Consents" shall mean all governmental and third party consents, approvals, authorizations, qualifications and waivers necessary to be received by a Person for the consummation of the Contemplated Transactions.

"Contemplated Transactions" shall mean the transactions contemplated by each of this Agreement, the other Transaction Documents, the CK Purchase Agreement and the other CK Transactions Documents.

"Contract" shall mean any legally binding contract, agreement, mortgage, deed of trust, bond, loan, indenture, lease, license, note, option, warrant, right, instrument, commitment or other similar document, arrangement or agreement, whether written or oral.

"Credit Facility" shall mean the Revolving Credit Agreement dated as of October 17, 2002 among the Company, certain of the Company Subsidiaries, as co-borrowers, the lenders identified therein (or that may thereafter become party thereto), and JPMorgan Chase Bank, as agent for such lenders.

"DGCL Section 203" shall have the meaning set forth in Section 3.26.

"Employee" shall mean any current, former or retired officers, directors, consultants, employees, independent contractors, agents and other Persons who render services to the Company or any Company Subsidiary.

"Employee Program" shall mean any "employee benefit plan", within the meaning of Section 3(3) of ERISA, whether or not it is subject to ERISA.

"Environment" shall mean soil, surface waters, ground waters, land, stream, sediments, surface or subsurface strata and ambient air.

"Environmental Laws" shall mean all Laws relating to the pollution of or protection of the Environment, from contamination by, or relating to injury to, or the protection of, real or personal property or human health or the Environment, including, without limitation, all valid and lawful requirements of courts and other Governmental Bodies pertaining to reporting, licensing, permitting, investigation, remediation and removal of, emissions, discharges, releases or threatened releases of Hazardous Materials, chemical substances, pesticides, petroleum or petroleum products, pollutants, contaminants or hazardous or toxic substances, materials or wastes, into the Environment, or relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of Hazardous Materials, pollutants, contaminants or hazardous or toxic substances, materials or wastes.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" shall mean any entity which at any time during the six-year period ending on the date of this Agreement has been considered a single employer with the Company or any Company Subsidiary under Section 4001(b) of ERISA or Section 414(b), (c), (m) or (o) of the Code.

"Excess Amount" shall have the meaning set forth in Section 10.01(d).

"Exchange Act" shall mean the Securities Exchange Act of 1934, as amended, and the rules and regulations of the SEC promulgated thereunder.

"Fees and Expenses" shall have the meaning set forth in Section 10.01(a).

"Fifth Amendment" shall have the meaning set forth in Section $7.01\,(\text{m})$.

"Foreign Plan" shall mean any plan which is described in Section 4(b)(4) of ERISA, and which is maintained, sponsored or contributed to by, or covers any employee of, the Company or any Company Subsidiary.

"GAAP" shall mean generally accepted accounting principles applied on a consistent basis as used in the United States of America.

"Governmental Body" shall mean any government or governmental or quasi-governmental authority including, without limitation, any federal, state, territorial, county, municipal or other governmental or quasi-governmental agency, board, branch, bureau, commission, court, arbitral body (public or private), department or other instrumentality or political unit or subdivision, whether located in the United States or abroad, the National Association of Securities Dealers, Inc., the New York Stock Exchange, the Nasdaq National Market, the Nasdaq SmallCap Market or the American Stock Exchange.

"Hazardous Materials" shall mean any substance whether solid, liquid or gaseous in nature: (i) the presence of which requires or may hereafter require notification, investigation, or remediation under any Environmental Law; (ii) which is or becomes defined as "toxic", a "hazardous waste", "hazardous material" or "hazardous substance" or "pollutant" or "contaminant" under any present or future Environmental Laws; (ii) the presence of which adversely affects or is injurious to human health or the Environment; (iv) which is toxic, explosive, corrosive, flammable, infectious, radioactive, carcinogenic, mutagenic or otherwise hazardous and is or becomes regulated by any Governmental Body; (v) which contains gasoline, diesel fuel or other petroleum hydrocarbons or volatile organic compounds; (vi) which contains polychlorinated byphenyls (PCBs) or asbestos or urea formaldehyde foam insulation; or (vii) which contains or emits radioactive particles, waves or materials, including radon gas.

"HSR Act" shall mean the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder.

- "Indemnitee" shall have the meaning set forth in Section 9.01.
- "Indemnitor" shall have the meaning set forth in Section 9.01.
- "Initial Purchase Price" shall have the meaning set forth in Section 2.01.
- "Initial Purchaser" shall have the meaning set forth in the preamble.

"Initial Purchaser Group Members" shall mean the Initial Purchaser and its Affiliates and their respective directors, officers, employees, agents and attorneys and their respective successors and assigns.

"Intellectual Property" shall have the meaning set forth in Section 3.14(a).

- "Investors" shall have the meaning set forth in the preamble.
- "Investor Group Member" shall mean the Investors and each of their respective partners, officers, employees, agents and attorneys and their respective successors and permitted assigns.
- "Investor Representatives" shall mean each of Apax Managers, Inc. and Apax Partners Europe Managers Ltd.
- "Investors' Rights Agreement" shall have the meaning set forth in Section 7.01(1).
 - "IRS" shall mean the Internal Revenue Service.
- "Law" shall mean any treaty, statute, ordinance, code, rule, regulation, Order or other legal requirement enacted, adopted, promulgated, applied or followed by any Governmental Body.

"Leased Real Property" shall have the meaning set forth in Section 3.12(b).

"Legal Proceeding" shall mean any judicial, administrative or arbitral actions, suits, proceedings (public or private) or governmental proceedings.

"Legend" shall mean the Legend set forth in Section 4.02(e).

"Liability" shall mean any debt, liability or obligation, whether known or unknown, asserted or unasserted, accrued, absolute, contingent or otherwise, whether due or to become due.

"Lien" shall mean any mortgage, pledge, lien (statutory or otherwise), security interest, hypothecation, conditional sale agreement, encumbrance or similar restriction or agreement.

"Loan Documents" shall mean the Term Loan Agreement between the Company, as borrower, and the Investors, as lenders, of even date herewith and the various instruments and agreements described therein.

"Loss" shall have the meaning set forth in Section 9.01.

"Marks" shall mean all of the trademarks, service marks and trade names throughout the entire world licensed to or used by the Company or any Company Subsidiary and/or owned by the Company or any Company Subsidiary including, but not limited to, Van Heusen, Bass,G.H. Bass & Co., Izod, cKCalvin Klein, Geoffrey Beene, Arrow, DKNY, Kenneth Cole New York and Reaction by Kenneth Cole.

"Material Adverse Effect" shall mean any event, condition or contingency that has had, or is reasonably likely to have, a material adverse effect on the business, assets, liabilities (including contingent liabilities), results of operations, financial condition or, to the knowledge of the Company, prospects of the Company and the Company Subsidiaries, taken as a whole. For the purposes of this Agreement, a Material Adverse Effect shall not be deemed to arise by reason of (i) the entering into of any transaction consented to by the Investor Representatives pursuant to Section 6.02, (ii) the Contemplated Transactions, (iii) events of war impacting the economy in general, and (iv) changes in general economic conditions or in the fashion, apparel, accessories, home furnishings or fragrance industries, in general.

"Material Employee" shall mean any Person, including any employee, officer or director of the Company or any Company Subsidiary, that earned or was paid in any twelve (12) month period two hundred and fifty thousand dollars (\$250,000) or more, for services rendered in any capacity to the Company or any Company Subsidiary during the past two (2) years.

"Maximum Amount" shall mean an amount equal to all of the Fees and Expenses plus all of the Monitor Expenses.

"Minimum Amount" shall mean an amount equal to 50% of the Fees and Expenses plus all of the Monitor Expenses.

- "Monitor Expenses" shall have the meaning set forth in Section 10.01(a).
- "Multiemployer Plan" shall have the meaning set forth in Section 3.16(a).
- "Notice" shall have the meaning set forth in Section 9.02(a).
- "NYSE" shall mean the New York Stock Exchange.

"Off-Balance Sheet Arrangement" shall mean any transaction, agreement or other contractual arrangement to which an entity that is not consolidated with the Company is a party, under which the Company, whether or not a party to the arrangement, has, or in the future may have: (i) any obligation under a direct or indirect guarantee or similar arrangement; (ii) a retained or contingent interest in assets transferred to an unconsolidated entity or similar arrangement; (iii) derivatives, to the extent that the fair value thereof is not fully reflected as a liability or asset in the financial statements of the Company; or (iv) any obligation or liability in excess of \$5,000,000, including a contingent obligation or liability, not incurred in the ordinary course of business to the extent that such obligation or liability is not fully reflected in the financial statements of the Company (excluding the footnotes thereto).

"Order" shall mean any order, injunction, judgment, decree, ruling, writ, assessment or arbitration award.

"Owned Real Property" shall have the meaning set forth in Section 3.12(a).

"PBGC" shall mean the Pension Benefit Guaranty Corporation.

"Partial Reimbursement" shall have the meaning set forth in Section 10.01(d).

"Percentage" shall mean that percentage of the Company's Expenses paid by CKI or any of its Affiliates.

"Permits" shall mean any approvals, authorizations, licenses, permits or certificates by or of any Governmental Body.

"Permitted Liens" shall mean (a) easements, restrictions, covenants, rights of way or minor irregularities of title currently of record against any of the Real Property, (b) real and personal property leases, (c) Liens for Taxes not yet due and payable, or for Taxes being contested in good faith, provided that in each such case, adequate reserves are maintained in accordance with GAAP on the Balance Sheet, and (d) any Lien created by statute of carriers, warehousemen, vendors, mechanics, laborers or materialmen incurred in the ordinary course of business for sums not yet due and payable.

"Person" shall mean any individual, corporation, partnership, firm, limited liability company, joint venture, trust, association, unincorporated organization, group, joint-stock company, Governmental Body or other entity.

"Preferred Stock" shall mean the preferred stock, \$100 par value per share, of the Company.

"Purchase Price" shall mean \$250,000,000 in the aggregate, payable as set forth in Section 2.02.

"Real Property" shall have the meaning set forth in Section 3.12(b).

"Registration Rights Agreement" shall have the meaning set forth in Section 7.01(k) .

"Reportable Event" shall mean a reportable event as defined in Section 4043 of ERISA and the regulations issued under such section, excluding, however, such events as to which the PBGC has by regulation waived the requirement of Section 4043 of ERISA that it be notified within 30 days of the occurrence of such event.

"Retiree" shall mean any Employee who has retired or terminated employment from the Company or any Company Subsidiary.

"Right" shall have the meaning set forth in the Rights Agreement.

"Rights Agreement" shall mean the Rights Agreement, dated as of June 10, 1986, as amended, by and between the Company and The Bank of New York (successor to The Chase Manhattan Bank, N.A.), as Rights Agent, and each amendment and extension thereof (including, without limitation, the Fifth Amendment).

"SEC" shall mean the U.S. Securities and Exchange Commission. ---

"Securities Act" shall mean the Securities Act of 1933, as amended, and the rules and regulations of the SEC promulgated thereunder.

"Sellers" shall have the meaning set forth in the CK Purchase Agreement.

"Series A Stock" shall mean the Series A Cumulative Participating Preferred Stock, \$100 par value per share, of the Company.

"Series B Designees" shall mean the directors elected by the Investors to the Board of Directors pursuant to the Certificate of Designations.

"Series B Shares" shall have the meaning set forth in Section 2.01.

"Series B Stock" shall have the meaning set forth in the recitals.

"Specified Purchase Price" shall have the meaning set forth in Section 2.01.

"Subsidiary" shall mean, as to any Person, any other Person more than 50% of the shares of the voting stock or other voting interests of which are owned or controlled, or the ability to select or elect more than 50% of the directors or similar managers is held, directly or indirectly, by such first Person or one or more of its Subsidiaries or by such first Person and one or more of its Subsidiaries.

"Tax Return" shall have the meaning set forth in Section 3.11(a).

"Taxes" shall mean all U.S. federal, state, local and foreign income, gross income, corporation, advance corporation, gross receipts, estimated, import, customs, duties, transfer, excise, property, sales, use, value-added, license, payroll, pay as you earn, withholding, social security and franchise or other governmental taxes, imposed by any Governmental Body and any interest, penalties or additions to tax with respect thereto.

"Termination Fee" shall have the meaning set forth in Section 11.03.

"Termination Warrants" shall mean ten-year warrants to purchase an aggregate amount of shares of Common Stock equal to ten percent (10%) of the Common Stock of the Company determined on a fully-diluted basis at an exercise price per share equal to the volume weighted average price of the shares Common Stock as reported by Bloomberg (or if such information is not available from Bloomberg, from another nationally recognized independent pricing source) on the date of execution of this Agreement, in the form annexed hereto as Exhibit B.

"Transaction Documents" shall mean this Agreement, the schedules and exhibits hereto, the Certificate of Designations, the Registrations Rights Agreement, the Investors' Rights Agreement, the Fifth Amendment and the Loan Documents and any certificate or other document delivered by or on behalf of the Company or the Investors pursuant to this Agreement or in connection with the transactions contemplated by this Agreement, but does not include the CK Transaction Documents.

"Unaudited Financial Statements" shall have the meaning specified in Section 3.07.

"Warrants" shall have the meaning set forth in the CK Purchase Agreement.

"2002 Budget" shall mean the annual operating budget for the Company, for the fiscal year commencing on February 4, 2002 and ending on February 2, 2003, previously delivered to each of the Investor Representatives.

Section 1.02. Rules of Construction. Unless the context otherwise requires:

- (a) an accounting term defined by GAAP that is not otherwise defined herein has the meaning assigned to it in accordance with GAAP;
 - (b) "or" is not exclusive;

- (c) words in the singular include the plural, and words in the plural include the singular;
- (d) the words "include" and "including" shall be deemed to mean "include, without limitation," and "including, without limitation";
- (e) "herein," "hereof," "hereto," "hereunder" and other words of similar import refer to this Agreement as a whole and not to any particular article, section, paragraph or clause where such terms may appear;
- (f) references to sections mean references to such section in this Agreement, unless stated otherwise; and
 - (g) the use of any gender shall be applicable to all genders.

ARTICLE II ISSUANCE, SALE AND PURCHASE OF THE SERIES B STOCK

Section 2.01. Sale and Purchase of the Series B Stock. Upon the terms and subject to the conditions of this Agreement, the Company will sell to the Initial Purchaser, and the Initial Purchaser will purchase from the Company, an aggregate of 10,000 shares of the Series B Stock (the "Series B Shares") for an aggregate purchase price of \$249,250,000 (the "Initial Purchase Price"). The Initial Purchaser will then immediately sell the Shares to the Investors, and the Investors will purchase from the Initial Purchaser, an aggregate of 10,000 shares of the Series B Shares for an aggregate purchase price of \$250,000,000 (the "Purchase Price"). The number of Series B Shares to be purchased by each Investor at the Closing and the portion of the aggregate purchase price to be paid by each Investor at the Closing in the exchange therefor, shall be as specified in Schedule 2.01 (with respect to each such Investor, such Investor's "Specified Purchase Price").

Section 2.02. Closing.

- (a) Subject to the satisfaction or waiver of the conditions set forth in this Agreement, the closing of the transactions contemplated by Section 2.01 (the "Closing") shall take place immediately prior to or concurrently with the closing of the CK Acquisition, or at such other time as may be mutually agreed upon by the Investor Representatives and the Company (the "Closing Date"). The Closing shall occur on the Closing Date at the offices of Katten Muchin Zavis Rosenman, 575 Madison Avenue, New York, New York.
- (b) At the Closing: (i) the Company will deliver to the Initial Purchaser certificates for the Series B Shares to be sold in accordance with the provisions of Section 2.01 registered in the name of the Initial Purchaser; (ii) the Initial Purchaser, in full payment for the Series B Shares, will deliver to the Company immediately available funds, by wire transfer to such account as the Company shall specify, the Initial Purchase Price; and (iii) each party shall

take or cause to happen such other actions, and shall execute and deliver such other instruments or documents, as shall be required under Article VII.

(c) At the Closing: (i) the Initial Purchaser will deliver to the Investor Representatives certificates for the Series B Shares to be sold in accordance with the provisions of Section 2.01 registered in the respective names of the Investors and proportions set forth in Schedule 2.01; (ii) each Investor, in full payment for the Series B Shares, will deliver to the Initial Purchaser immediately available funds, by wire transfer to such account as the Initial Purchaser shall specify, such Investor's Specified Purchase Price; and (iii) each party shall take or cause to happen such other actions, and shall execute and deliver such other instruments or documents, as shall be required under Article VII.

Section 2.03. Use of Proceeds. The Company shall use the proceeds from the sale of the Series B Stock solely to (i) pay a portion of the purchase price for the Capital Stock of the CK Companies and (ii) pay the fees and out-of-pocket expenses relating to the Contemplated Transactions.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company represents and warrants to the Initial Purchaser and to each of the Investors as follows:

Section 3.01. Organization and Good Standing. The Company is a corporation duly incorporated, validly existing and in good standing under the Laws of the jurisdiction of its incorporation, has full corporate power and authority to own, lease and operate its properties, and carry on its business as presently conducted. The Company is duly qualified, registered or licensed as a foreign corporation to do business and is in good standing in each jurisdiction in which the ownership or leasing of its properties or the character of its present operations makes such qualification, registration or licensing necessary, except where the failure to so qualify or be in good standing could not reasonably have a Material Adverse Effect. The Company has heretofore delivered or made available to the Initial Purchaser and to the Investors complete and correct copies of the Certificate of Incorporation and By-Laws of the Company, as in effect as of the date of this Agreement.

Section 3.02. Authority; Binding Effect. The Company has full corporate power and authority to execute and deliver this Agreement, the other Transaction Documents and the CK Transaction Documents and to consummate the Contemplated Transactions. The execution and delivery of this Agreement and the other Transaction Documents and the consummation by the Company of the transactions contemplated hereby and by the other Transaction Documents have been duly and validly approved by all necessary corporate action on the part of the Company. This Agreement has been duly executed and delivered by the Company and constitutes the legal, valid and binding obligation of the Company, enforceable in accordance with its terms, except as such enforceability may be subject to the effects of any applicable bankruptcy, insolvency,

fraudulent conveyance, reorganization, moratorium or similar Laws affecting creditors' rights generally and subject to the effects of general equitable principles. The other Transaction Documents and the CK Transaction Documents, when executed and delivered by the Company, will be duly executed and delivered by the Company and constitute legal, valid and binding obligations of the Company, enforceable in accordance with their respective terms, except as such enforceability may be subject to the effects of any applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or similar Laws affecting creditors' rights generally and subject to the effects of general equitable principles.

Section 3.03. Organization and Good Standing of Company Subsidiaries. Schedule 3.03 lists all Subsidiaries of the Company and their respective jurisdictions of incorporation (collectively, the "Company Subsidiaries" and each, a "Company Subsidiary"). Except as set forth in Schedule 3.03, the Company owns, directly or indirectly, all the shares of outstanding Capital Stock of each Company Subsidiary. There are no outstanding securities or rights convertible into or exchangeable for shares of any Capital Stock of any Company Subsidiary and there are no Contracts by which any Company Subsidiary is bound to issue additional shares of Capital Stock. All of the shares of Capital Stock of each of the Company Subsidiaries are duly and validly authorized, fully paid and non-assessable and, except for the Liens created by the security agreement entered into in connection with the Credit Facility, are owned by the Company free and clear of any Lien with respect thereto. Each Company Subsidiary is a corporation duly organized, validly existing and, where recognized, in good standing under the Laws of its jurisdiction of organization, and has all requisite corporate power and authority to own, operate and lease its properties and to carry on its business as it is now being conducted, and is duly licensed or qualified to do business in each other jurisdiction in which it owns or leases properties, or conducts any business, so as to require such qualification, except where the failure to be so licensed or qualified in any such jurisdiction could not reasonably have a Material Adverse Effect.

Section 3.04. Capitalization.

(a) Schedule 3.04(a) sets forth, in each case as of the date hereof, (i) the authorized Capital Stock of the Company, the number of shares of each class of Capital Stock issued and outstanding and the number of shares of Common Stock reserved for issuance in connection with the Company's stock option plans, and (ii) all options, warrants, rights to subscribe to, calls, contracts, undertakings, arrangements and commitments to issue which may result in the issuance of Capital Stock of the Company, other than (x) options to purchase 615,887, 1,886,878 and 2,387,564 shares of Common Stock issued and outstanding under the Company's 1987, 1997 and 2000 stock option plans, respectively and (y) the Rights. All of the issued and outstanding shares of the Company's Capital Stock have been duly and validly authorized and issued and are fully paid and non-assessable and are not subject to any preemptive rights. Except pursuant to this Agreement, the Company's stock option plans, the Rights Agreement and the Certificate of Designations, (i) no equity securities of the Company are or may be required to be issued by reason of any options, warrants, rights to subscribe to, calls or commitments of any character whatsoever, (ii) there are outstanding no securities or

rights convertible into or exchangeable for shares of any Capital Stock of the Company, and (iii) there are no contracts, commitments, understandings or arrangements by which the Company is bound to issue additional shares of its Capital Stock or securities or rights convertible into or exchangeable for shares of any Capital Stock of the Company, or options, warrants or rights to purchase or acquire any additional shares of its Capital Stock. Neither the Company nor any Company Subsidiary is subject to any obligation (contingent or otherwise) to repurchase or otherwise acquire or retire any of its Capital Stock. Except as contemplated by the Registration Rights Agreement, there are no Contracts between the Company and any Person granting such Person the right to require the Company to file a registration statement under the Securities Act with respect to any securities of the Company owned or to be owned by such Person or to require the Company to include such securities in any other registration statement filed by the Company under the Securities Act.

- (b) The Series B Shares will have the voting powers, designations, preferences and rights, and the qualifications, limitations and restrictions thereof, set forth in the Certificate of Designations which will be filed with the Secretary of State of the State of Delaware on or prior to the Closing. The Company has reserved, or will on or before Closing reserve, for issuance the shares of Common Stock issuable upon conversion of the Series B Shares. When paid for by, and issued to, the Initial Purchaser, the Series B Shares will be duly authorized, validly issued, fully paid and non-assessable and will be free and clear of any Liens. When paid for by, and sold to, each of the Investors the Series B Shares will be duly authorized, validly issued, fully paid and non-assessable and will be free and clear of any Liens. The issuance and sale of the Series B Shares is not subject to any preemptive rights. Except for the restrictions set forth, or referred to, in the Legend, the Series B Shares when issued and sold will not be subject to any restriction on use, voting or transfer; and the shares of Common Stock issuable to each such Investor upon conversion of the Series B Shares, when issued in accordance with the Company's Certificate of Incorporation, will be duly authorized, validly issued, fully paid and non-assessable, and will be free and clear of any Liens and except for the restrictions set forth, or referred to, in the Legend, will not be subject to any restriction on use, voting or transfer or to any preemptive rights.
- (c) Without limiting the generality of subsection (b) above, the terms and conditions of the Rights to be associated with the Series B Shares will be set forth in the Rights Agreement, substantially in the form provided to the Initial Purchaser and to each of the Investors. The Company has reserved, or will on or before Closing reserve, for issuance the shares of Series A Stock issuable upon exercise of the Rights. The shares of Series A Stock issuable to each such Investor upon exercise of the Rights, when issued in accordance with the Rights Agreement, will be duly authorized, validly issued, fully paid, non-assessable, and free of preemptive rights and will be free and clear of any Liens and except for the restrictions set forth, or referred to, in the Legend, will not be subject to any restriction on use, voting or transfer or to any preemptive rights.

Section 3.05. No Violations; Consents. Except as set forth on Schedule 3.05, neither the execution, delivery or performance by the Company of this Agreement, the other Transaction

Documents or the CK Transaction Documents nor the consummation of the Contemplated Transactions, will (a) conflict with, or result in the breach of, any provision of the Certificate of Incorporation or By-Laws of the Company or any Company Subsidiary, (b) conflict with, violate, result in the breach or termination of, or constitute a default or give rise to any right of termination or acceleration or right to increase the obligations or otherwise modify the terms thereof under any Contract, Permit or Order to which the Company or any Company Subsidiary is a party or by which the Company or any Company Subsidiary or any of the properties or assets of the Company or any Company Subsidiary is bound, (c) constitute a violation of any Law applicable to the Company or any Company Subsidiary; or (d) result in the creation of any Lien upon the properties or assets of the Company or any Company Subsidiary, other than with respect to the foregoing clauses (b), (c) and (d), such requirements, conflicts, violations, breaches or rights which could not reasonably have a Material Adverse Effect. Except as set forth on Schedule 3.05, other than those which have been obtained or made or which could not reasonably have a Material Adverse Effect, no Consent is required on the part of the Company or the Company Subsidiaries in connection with the execution and delivery of this Agreement or the Transaction Documents, or the compliance by the Company with any of the provisions hereof or thereof.

Section 3.06. Listing. The Company is not in violation of the listing requirements of the NYSE in any material respect. The Company has not received any written notice from the NYSE that the Common Stock is to be delisted by the NYSE. The transactions contemplated under this Agreement constitute a bona fide private financing under paragraph 312.03 of the NYSE Listed Company Manual and stockholder approval is not required under the rules and regulations of the NYSE in order to authorize the issuance of the Series B Shares pursuant to this Agreement or the listing of the Common Stock into which the Series B Stock is convertible.

Section 3.07. Financial Statements. The Company has previously delivered to the Initial Purchaser and to the Investors copies of (i) the consolidated balance sheet of the Company and the Company Subsidiaries as of February 3, 2002 and February 4, 2001 and the related consolidated income statements, changes in stockholders' equity and cash flows for the fiscal years ended February 3, 2002, February 4, 2001 and January 30, 2000, as reported in the Company's Annual Report on Form 10-K for the fiscal year ended February 3, 2002, filed by the Company with the SEC under the Exchange Act, and accompanied by the audit report of Ernst & Young LLP, independent public accountants, (collectively, the "Audited Financial Statements"), and (ii) the unaudited consolidated balance sheet of the Company and the Company Subsidiaries as of August 4, 2002 (the "Balance Sheet") and the related unaudited consolidated income statements and cash flows for the thirteen weeks and twenty-six weeks ended August 4, 2002, as reported in the Company's Quarterly Report on Form 10-Q for the quarterly period ended August 4, 2002, filed with the SEC under the Exchange Act (the "Unaudited Financial Statements"). The Audited Financial Statements accurately reflect the books and records of the Company and

present fairly, in all material respects, the consolidated financial position of the Company and the Company Subsidiaries and the consolidated results of their operations and their cash flows for the periods and dates covered thereby, in conformity with GAAP. The Unaudited Financial Statements accurately reflect the books and records of the Company and present fairly, in all material respects, the consolidated financial position of the Company and the Company Subsidiaries and the consolidated results of their operations and their cash flows for the period and date covered thereby, in conformity with GAAP, except for changes resulting from year-end adjustments (none of which will be material in amount) and the absence of footnote disclosures thereto. The Company and the Company Subsidiaries have no liabilities or obligations of a type that GAAP would require to be on the Balance Sheet (absolute, accrued, contingent or otherwise) which are not fully reflected or reserved against in the Balance Sheet, except for liabilities and obligations that may have arisen in the ordinary and usual course of business and consistent with past practice since August 4, 2002 and that, individually or in the aggregate, could not reasonably have a Material Adverse Effect. Neither the Company nor any Company Subsidiary is currently a party to any Off-Balance Sheet Arrangement, which is not reflected in the financial statements (or the footnotes thereto) referred to in this Section 3.07. During the past three years, the Company has not restated any of its published financial results and the Company is not aware of any facts which may require such restatement.

Section 3.08. Commission Filings. The Company has filed all reports, registration statements, proxy statements and other materials, together with any amendments required to be made with respect thereto, that were required to be filed with the SEC under the Securities Act or the Exchange Act from and after January 30, 2000 (all such reports and statements are collectively referred to herein as the "Commission Filings"). As of their respective dates, the Commission Filings, including the financial statements contained therein, complied in all material respects with all of the statutes and published rules and regulations enforced or promulgated by the regulatory authority with which the Commission Filings were filed, and, except to the extent the information in any Commission Filing has been revised or superseded by a later filed Commission Filing, did not and do not as of the date hereof contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading.

Section 3.09. Absence of Certain Developments.

- (a) Except as set forth on Schedule 3.09(a), since August 4, 2002 and through the date hereof:
 - (i) no event occurred which could reasonably have a Material Adverse Effect;
 - (ii) there has not been any declaration, setting a record date, setting aside or authorizing the payment of, any dividend or other distribution in respect of any shares of Capital Stock of the Company or any repurchase, redemption or other acquisition by the Company, of any of the outstanding shares of Capital Stock of the Company other than the Company's regular quarterly dividends to its stockholders paid on September 13, 2002 and December 16, 2002;

- (iii) neither the Company nor any Company Subsidiary has transferred, issued, sold or disposed of any shares of their Capital Stock or granted any options, warrants, calls or other rights to purchase or otherwise acquire shares of their Capital Stock other than under the Company's employee stock option plans and the dissolution of former Company Subsidiaries;
- (iv) neither the Company nor any Company Subsidiary, except in the ordinary course of business and consistent with past practice, has (x) awarded or paid any material bonuses to any senior executive, or (y) entered into any Plan, material employment, deferred compensation, severance or similar agreement (nor amended or terminated any such agreement) or agreed to increase materially the compensation payable or to become payable to any senior executive or agreed to increase materially the coverage or benefits available under any material severance pay, deferred compensation, bonus or other incentive compensation, pension or other employee benefit plan, payment or arrangement made to, for or with such senior executive;
- (v) except in connection with the CK Acquisition, neither the Company nor any Company Subsidiary has made, or agreed to, make any material acquisition of any business or assets other than in the ordinary course of business;
- (vi) neither the Company nor any Company Subsidiary has made, or agreed to make, any loans or investments in any business of any Affiliate of the Company other than a Company Subsidiary;
- (vii) there has not been any damage, destruction or loss, whether or not covered by insurance, with respect to the property of the Company or any Company Subsidiary having a material adverse impact on the business of the Company or the Company Subsidiaries, taken as a whole;
- (viii) except as granted under the Credit Facility or the Transaction Documents, neither the Company nor any Company Subsidiary has mortgaged, pledged or subjected to any Lien (other than Permitted Liens) any of its assets, or sold, assigned, transferred, conveyed or otherwise disposed of any material assets of the Company or any Company Subsidiary, except for assets sold, assigned, transferred, conveyed or otherwise disposed of in the ordinary course of business;
- (ix) except for the CK Purchase Agreement, neither the Company nor any Company Subsidiary has entered into any Contract to make any capital expenditures obligating the Company or any Company Subsidiary to pay an amount which together with amounts already expended or committed during its 2002 fiscal year would exceed the amount budgeted for capital expenditures in the 2002 Budget;
- (\mathtt{x}) except in connection with the Credit Facility and the CK Acquisition, neither the Company nor any Company Subsidiary has created, incurred, assumed

or guaranteed any debt for borrowed money, whether due or to become due, other than in the ordinary course of business and consistent with past practice;

- (xi) except in connection with the Contemplated Transactions, neither the Company nor any Company Subsidiary has entered into any material transaction other than in the ordinary course of business consistent with past practice;
- (xii) neither the Company nor any Company Subsidiary has made any change in the accounting principles, methods or practices followed by it (including, without limitation, its method of accounting for stock options) other than a change which was required by reason of a concurrent change in Law or GAAP;
- (xiii) neither the Company nor any Company Subsidiary has amended its Certificate of Incorporation or By-Laws except as contemplated by this Agreement:
- (xiv) neither the Company nor any Company Subsidiary has had any material disagreements with its independent public accountants regarding any matter of accounting principles or practices, financial statement disclosure, or auditing scope or procedure which has not been resolved;
- (xv) neither the Company nor any Company Subsidiary has sold, assigned or transferred, or allowed any rights to lapse with respect to, Intellectual Property, other than in the ordinary course of business and consistent with past practice and which could reasonably have a Material Adverse Effect;
- (xvi) neither the Company nor any Company Subsidiary has entered into, modified, amended or terminated any material Contract, other than in the ordinary course of business and consistent with past practice and which could reasonably have a Material Adverse Effect; and
- (xvii) neither the Company nor any Company Subsidiary has agreed, whether in writing or otherwise, to do any of the foregoing.

Section 3.10. Litigation. There are no Legal Proceedings pending or, to the knowledge of the Company, threatened, that question the validity of this Agreement or the Transaction Documents or any action taken or to be taken by the Company or any Company Subsidiary in connection with the consummation of the Contemplated Transactions. Except as otherwise disclosed herein or on Schedule 3.10, there are no Legal Proceedings pending or, to the knowledge of the Company, threatened, against or involving the Company or any Company Subsidiary or any of their respective properties or assets, at Law or in equity, involving claims of more than \$1,000,000 by stockholders of the Company, or claims involving more than \$1,000,000 for product liability, infringement of trademark, patent or intellectual property, violations of health and safety laws covering Employees, violations of Environmental Laws, violations of customs laws, sexual harassment or discrimination, or racial discrimination. There

is no outstanding or, to the knowledge of the Company, threatened, Order of any Governmental Body against the Company or any Company Subsidiary or any of their respective properties or assets, which Order could reasonably have a Material Adverse Effect.

Section 3.11. Tax Matters.

- (a) For the periods since January 1, 1999, the Company and each Company Subsidiary has timely filed or caused to be timely filed any and all returns, declarations, reports (including any consolidated, combined or unitary returns), claims for refund, information returns, or other documents or statements relating to Taxes, including any schedule or attachment thereto and any amendment or supplement thereof (each, a "Tax Return") required to be filed by it under applicable federal, state, local or foreign Law, except to the extent that any failure to do so could not reasonably have a Material Adverse Effect. The reserves for Taxes contained in the financial statements of the Company or carried on the books and records of the Company and the Company Subsidiaries, as applicable, are in the aggregate adequate to cover all Tax liabilities and deferred Taxes of the Company and the Company Subsidiaries as of the date of this Agreement, except to the extent that any inadequacy could not reasonably have a Material Adverse Effect.
- (b) For the periods since January 1, 1999, all Taxes shown as being due and owing by the Company or any Company Subsidiary on any Tax Return have been paid.
- (c) The Company and each Company Subsidiary has timely withheld and paid all Taxes required to have been withheld and paid in connection with any amounts paid or owing to any employee, independent contractor, creditor, stockholder, or third party, except to the extent that any failure to do so could not reasonably have a Material Adverse Effect.
- (d) None of the Company nor any Company Subsidiary has waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency, in each case except as to Taxes that are disclosed on Schedule 3.11(d) or in the financial statements of the Company and the Company Subsidiaries, as applicable, or that, if assessed could not reasonably have a Material Adverse Effect.
- (e) The Company has not been a "United States real property holding corporation" within the meaning of Section 897 of the Code during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code.

Section 3.12. Real Property.

(a) Except as disclosed in Schedule 3.12(a), each of the Company and the Company Subsidiaries has fee simple title to the material real property that it owns (the "Owned Real Property"), free and clear of all Liens (other than Permitted Liens).

- (b) Schedule 3.12(b) contains a description of all real property leased by the Company and the Company Subsidiaries as lessee (the "Leased Real Property;" the Leased Real Property and the Owned Real Property are collectively referred to as the "Real Property"), other than properties leased for retail store operations. Except as disclosed in Schedule 3.12(b), each of the Company and the Company Subsidiaries has a subsisting leasehold estate in the Leased Real Property identified on Schedule 3.12(b), free and clear of all Liens (other than Permitted Liens), and is in sole possession of each parcel of Leased Real Property that it leases.
- (c) There are no condemnation or appropriation proceedings pending or, to the knowledge of the Company, threatened, against the Real Property.
- (d) The buildings, structures, facilities, fixtures and other improvements constituting a portion of the Owned Real Property are in good operating condition and in a state of good maintenance and repair, ordinary wear and tear excepted, and are adequate and suitable for the purposes for which they are presently being used.
- Section 3.13. Inventory. The Company does not now intend to write-down more than \$5,000,000 in inventory.

Section 3.14. Intellectual Property.

- (a) For the purposes of this Agreement, "Intellectual Property" shall mean (i) all Marks, trade dress, logos, trade names, domain names, web-sites, brand names and corporate names (and all licenses or other rights relating to any of the foregoing), together with all goodwill associated therewith, and all applications, registrations and renewals in connection therewith, (ii) all inventions and designs (whether patentable or unpatentable and whether or not reduced to practice), all improvements thereto, and all patents, patent applications, and patent disclosures, together with all reissuances, continuations, continuations-in-part, revisions, extensions and reexaminations thereof, and (iii) all copyrightable works, all copyrights and all applications, registrations and renewals in connection therewith, in each case which is owned by or licensed to the Company and the Company Subsidiaries.
- (b) Each of the Company and the Company Subsidiaries owns, or is licensed to use, all Intellectual Property material to its business, and to the knowledge of the Company, the use thereof by the Company and the Company Subsidiaries does not infringe upon the trademark, copyright or proprietary rights of any other Person, except for any such infringements that, individually or in the aggregate, could not reasonably have a Material Adverse Effect.
- (c) Except as set forth on Schedule 3.14(c), no Mark licensed to or used by the Company or any Company Subsidiary is scheduled to expire (without the right of extension on the part of the Company) within 12 months from the date hereof, which such expiration could reasonably have a Material Adverse Effect.

Section 3.15. Material Contracts.

- (a) Except as set forth on Schedule 3.15(a) or in the Commission Filings or for the CK Transaction Documents, neither the Company, the Company Subsidiaries nor any of their properties or assets is a party to or bound by any (i) Contract involving a commitment or payment by the Company or the Company Subsidiaries in excess of \$5,000,000 in any twelve-month period (whether or not in the ordinary course of business); (ii) Contract among stockholders that beneficially own five percent (5%) or more of the Common Stock; (iii) Contract (other than a contract for any Leased Real Property) which contains any provision that may require payments to be made by the Company or any Company Subsidiary upon or following a "change of control" (as such term is defined by such Contract) or similar event. The Company has delivered or otherwise made available to the Initial Purchaser and the Investors complete and correct copies of the Contracts listed on Schedule 3.15(a), together with all amendments, modifications, supplements or side letters to such Contracts.
- (b) There is no default and the Company has received no written notice of default under any Contract listed on Schedule 3.15(a) or a Contract listed in the Commission Filings by the Company, the Company Subsidiaries or, to the knowledge of the Company, by any other party thereto, in each case which could reasonably have a Material Adverse Effect, and no event has occurred that with the lapse of time or the giving of notice or both would constitute a default thereunder by the Company, the Company Subsidiaries or, to the knowledge of the Company, by any other party thereto, that could reasonably have a Material Adverse Effect.

Section 3.16. Company Employee Plans.

- (a) Schedule 3.16(a) sets forth a complete and correct list of every Employee Program which is maintained, administered, sponsored or contributed to by the Company or any Company Subsidiary, which covers any Employee of the Company or any Company Subsidiary or with respect to which an obligation of the Company or any Company Subsidiary to make any contribution exists, other than a Foreign Plan. Any Employee Program listed in such schedule (other than any employee Program that is a multiemployer plan, as defined in Section 3(37) of ERISA (a "Multiemployer Plan") is referred to below as a "Company Employee Plan"
- (b) The Company has made available to the Initial Purchaser and the Investors with respect to each Company Employee Plan complete and correct copies of (i) all written documents comprising such Company Employee Plan (including amendments, individual agreements, service agreements, trusts and other funding agreements), (ii) the three most recent annual returns in the Federal Form 5500 series (including all schedules thereto) filed with respect to such Company Employee Plan, (iii) the three most recent audited financial statements and actuarial reports, if any, pertaining to such Company Employee Plan, (iv) the summary plan description currently in effect and all material modifications thereto, if any, for such Company Employee Plan, (v) any employee handbook which includes a description of such Company Employee Plan, (vi) the most recent IRS determination letter, if any, for such Company Employee Plan, and (vii) any other written communications to any Employee, to the extent that the provisions of such Company Employee Plan described therein differ materially from such

provisions as set forth or described in the other information or materials furnished under this subsection (b).

- (c) Each Company Employee Plan which is intended to qualify under Section 401(a) of the Code has received a determination letter from the IRS which states that such plan is so qualified, and on which any employer which has adopted such plan may currently rely.
- (d) Each Company Employee Plan and Foreign Plan has been maintained in accordance with its terms and with all applicable Laws, except to such extent as could not reasonably have a Material Adverse Effect. Neither the Company nor any Company Subsidiary has any unsatisfied liability, or any unpaid fine, penalty or tax, with respect to any Company Employee Plan, any Foreign Plan or any other Employee Program, which could reasonably have a Material Adverse Effect. To the knowledge of the Company, there has been no "prohibited transaction" (within the meaning of Section 406 of ERISA or Section 4975 of the Code), or any breach of any duty under ERISA, any other applicable Law or any agreement, with respect to any Company Employee Plan or Foreign Plan which could subject the Company or any Company Subsidiary to material liability either directly or indirectly (including, without limitation, through any obligation of indemnification or contribution) for any damages, penalties, taxes or any other loss or expense which could reasonably have a Material Adverse Effect. The Company and each Company Subsidiary has made full and timely payment of all contributions required to be made by it to each Company Employee Plan and Foreign Plan by the terms of any such plan or under the applicable Law, except that all contributions which are so required to be made by the Company or any Company Subsidiary to each Company Employee Plan and Foreign Plan for any period ending prior to the Closing, but which are not due by the date of the Closing, shall be properly reserved or accrued in the appropriate financial statements. There have been no material violations of any reporting or disclosure requirements under ERISA or the Code with respect to any Company Employee Plan, including any requirement to file an annual return.
- (e) No litigation or claim (other than routine claims for benefits), and no governmental administrative proceeding, audit or investigation, is pending or, to the knowledge of the Company or any of the Company Subsidiaries, threatened with respect to any Company Employee Plan or Foreign Plan which could reasonably have a Material Adverse Effect.
- (f) With respect to each Company Employee Plan which is a defined benefit pension plan, (i) such Company Employee Plan has not incurred an "accumulated funding deficiency", within the meaning of Section 412 of the Code or Section 302 of ERISA, whether or not waived, (ii) such Company Employee Plan has not been terminated, (iii) no notice of intent to terminate such Company Employee Plan has been issued to participants or filed with the PBGC, (iv) the PBGC has not instituted any proceeding to terminate, or to appoint a trustee or administrator of, such Company Employee Plan, and no circumstances exist that are reasonably likely to constitute grounds under Section 4042 of ERISA which may allow the PBGC to institute any such proceeding, (v) except as set forth on Schedule 3.16(f), if such Company Employee Plan is intended to be a "qualified plan" within the meaning of Section 401(a) of the

Code, as of the most recent valuation date for such Company Employee Plan, the present value of benefit liabilities, when computed on a termination basis using actuarial assumptions which are reasonable under the circumstances, does not exceed the value of assets, (vi) neither the Company nor any Company Subsidiary (x) has incurred any liability to the PBGC or any other Person, or has become subject to any lien, under Title IV of ERISA in connection with such Company Employee Plan (other than PBGC premiums), or (y) has knowledge of any facts or transactions that might reasonably be anticipated to result in the imposition of any liability on, or the imposition of any lien on the assets of, the Company or any of the Company Subsidiaries to, or in favor of, the PBGC or any other Person under Title IV of ERISA in connection with such Company Employee Plan (other than PBGC premiums), and (vii) no Reportable Event has occurred with respect to such Company Employee Plan.

- (g) Except as set forth on Schedule 3.16(g), no Employee Program which has been maintained, administered or contributed to by the Company or any Company Subsidiary, which has covered any Employee of the Company or any Company Subsidiary, or to which the Company or any Company Subsidiary has had any obligation to make a contribution, during the six-year period ending on the date of this Agreement, is (i) a "multiemployer plan", as defined in Section 3(37) of ERISA, (ii) a "multiple employer plan", as described in Section 413(c) of the Code, (iii) a "multiple employer welfare arrangement", as defined in Section 3(40) of ERISA, (iv) a "voluntary employees' beneficiary association" within the meaning of Section 501(c)(9) of the Code, or (v) a Foreign Plan other than a plan which does not provide benefits materially greater than any benefits which may be required by applicable law. Neither the Company nor any Company Subsidiary has, within the past six years, incurred any withdrawal liability to a Multiemployer Plan.
- (h) Except as set forth on Schedule 3.16(h), all health and medical benefit coverage, and all death benefit coverage, under each Company Employee Plan is provided solely through insurance, and no Company Employee Plan provides health or medical coverage, life insurance coverage, or coverage for any other welfare benefit to any Retiree, except for continuation coverage required by Section 4980B of the Code, Sections 601 to 608 of ERISA or any applicable State Law.
- (i) No Employee of the Company or any Company Subsidiary shall accrue or receive additional benefits, additional credit for service, accelerated vesting or accelerated rights to payment of any benefit under any Company Employee Plan, or become entitled to any severance, termination allowance or similar payments or to the forgiveness of any indebtedness, solely as a result of the execution and delivery of, or the transactions contemplated by this Agreement. Such execution and delivery, or the occurrence of such transactions, shall not result in any increase in the contributions required to be made to any Company Employee Plan. Except as set forth on Schedule 3.16(i), no payment made or contemplated under any Company Employee Plan, or by the Company or any Company Subsidiary, in connection with the Contemplated Transactions, constituted, or would constitute, either (i) an "excess parachute payment" within the meaning of Section 280G of the Code or (ii) a payment which is not deductible by reason of Section 404 of the Code.

- (j) Except as set forth on Schedule 3.16(j), (i) except for the adoption of a plan amendment which is needed to bring the plan documents into conformity with statutory changes enacted in recent years, neither the Company nor any Company Subsidiary is under any obligation (express or implied) to modify any Company Employee Plan, or to establish any new Employee Program which will cover any Employee of the Company or any Company Subsidiary, (ii) the Company or a Company Subsidiary has expressly reserved to itself the right to amend, modify or terminate each Company Employee Plan (and any service or funding agreement or arrangement for each Company Employee Plan), at any time without material liability or penalty to itself (other than routine expenses), and (iii) no Company Employee Plan requires the Company or any of the Company Subsidiaries to continue to employ or use the services of any Employee.
- (k) Except as set forth on Schedule 3.16(k), there has been no amendment, interpretation or announcement by the Company or any of the Company Subsidiaries relating to any Company Employee Plan which would materially increase the expense of maintaining such plan above the level of expense incurred with respect to that plan, as indicated in the applicable financial statements, for its most recent fiscal year.
- (1) No ERISA Affiliate has any unpaid liability, fine, penalty or tax with respect to any Employee Program for which the Company or any of the Company Subsidiaries could be liable, and which could reasonably have a Material Adverse Effect.
 - Section 3.17. Labor Relations. Except as set forth on Schedule 3.17(a):
- (a) the Company and the Company Subsidiaries are in compliance in all material respects with all applicable Laws respecting employment and employment practices, terms and conditions of employment, wages, hours, collective bargaining, safety and health, work authorization, equal employment opportunity, immigration, withholding, unemployment compensation, worker's compensation and employee privacy and right to know, other than any such failure to comply that could not reasonably have a Material Adverse Effect;
- (b) there is no pending, or to the knowledge of the Company, any threatened, charge or complaint, against the Company or any of the Company Subsidiaries before the National Labor Relations Board or any other comparable Governmental Body, which could reasonably have a Material Adverse Effect;
- (c) there is, and for the past three years have been, (i) no labor strike, union-related slowdown, work stoppage or other union-related labor controversy pending, or to the knowledge of the Company, threatened in writing, against or otherwise affecting or involving the Company or the Company Subsidiaries or (ii) no lawsuits (other than grievance proceedings) pending, or to the knowledge of the Company, threatened, between any of the Company or the Company Subsidiaries and any employee of the Company or the Company Subsidiaries or any union or other collective bargaining unit representing any employees of the Company or the Company Subsidiaries, which could reasonably have a Material Adverse Effect;

- (d) none of the employees of the Company or the Company Subsidiaries are covered by any collective bargaining agreements (other than as disclosed in the Commission Filings) and, to the knowledge of the Company, no solicitation of the employees of the Company or any of the Company Subsidiaries is currently being made by any union to organize any of such employees; and
- (e) no Material Employee has given notice to the Company or any Company Subsidiary of his intent to terminate his employment with the Company or any Company Subsidiary.

Section 3.18. Compliance with Laws; Permits.

- (a) The Company and the Company Subsidiaries are in compliance in all material respects with all material Laws (including, without limitation, the Sarbanes-Oxley Act of 2002) and material Orders promulgated by any Governmental Body applicable to the Company and the Company Subsidiaries or to the conduct of the business or operations of the Company and the Company Subsidiaries or the use of their properties (including any leased properties) and assets. Since January 1, 1999, neither the Company nor any Company Subsidiary has received any written notice of violation or alleged material violation of any such Law or Order by any Governmental Body in any material respect that has not been resolved. Since January 1, 1999, neither the Company nor any Company Subsidiary has received written notice that it is the subject of an investigation by any Governmental Body which could reasonably have a Material Adverse Effect.
- (b) To the knowledge of the Company, except as set forth on Schedule 3.18(b), the Company and the Company Subsidiaries have all Permits necessary for the conduct of their business, except where the failure to have such Permits could not reasonably have a Material Adverse Effect. The Company and the Company Subsidiaries have complied in all material respects with all conditions of such Permits applicable to it.

Section 3.19. Preferred Stock Exemption.

- (a) Assuming the representations and warranties of the Initial Purchaser and of the Investors contained in Article IV are true, the offer and sale of the Series B Shares (and the issuance of the Common Stock to such Investors upon the conversion of such Series B Shares) are exempt from the registration requirements of the Securities Act. The Company has not taken and will not take any actions which would cause the offers and sales contemplated hereunder to become ineligible for exemption under the Securities Act.
- (b) Neither the Company nor any Person acting on its behalf has offered the Series B Stock to any Person by means of general or public solicitation or general or public advertising, such as by newspaper or magazine advertisements, by broadcast media, or at any seminar or meeting whose attendees were solicited by such means.

- (a) Each of the Company and the Company Subsidiaries is in compliance in all material respects with all limitations, restrictions, conditions, standards, prohibitions, requirements, obligations, schedules and timetables contained in any applicable Environmental Laws, or in any plan, Order, notice or demand letter issued, entered, promulgated or approved thereunder.
- (b) Except as set forth on Schedule 3.20(b), to the knowledge of the Company, as a result of activities of the Company or the Company Subsidiaries or any of their employees, no Hazardous Material has been incorporated in, used on, stored on or under, released from, treated on, transported to or from, or disposed of on or from, any property owned or during the period of the Company's or any of the Company Subsidiaries' lease(s), or to the knowledge of the Company without any investigation, any prior period, leased by the Company or any Company Subsidiary such that, under applicable Environmental Laws (i) any such Hazardous Material would be required to be removed, cleaned-up or remediated before the property could be altered, renovated, demolished or transferred, or (ii) the owner or lessee of the property could be subjected to liability for the removal, clean-up or remediation of such Hazardous Material, except, under circumstances that could not reasonably have a Material Adverse Effect; and, except as set forth on Schedule 3.20(b), the Company has not received any written notification from any Governmental Body or other third parties relating to Hazardous Material on or affecting any property owned or leased by the Company or the Company Subsidiaries or relating to any potential or known liability under applicable Environmental Laws arising from the ownership or leasing of any property, which could reasonably have a Material Adverse Effect
- Section 3.21. Investment Company Act. The Company and the Company Subsidiaries are not, nor are they directly or indirectly controlled by or acting on behalf of any Person that is, an investment company within the meaning of the Investment Company Act of 1940, as amended.
- Section 3.22. Transactions with Affiliates. Except: (i) for transactions between the Company and any wholly-owned Company Subsidiary, (ii) for expenses in the ordinary course of business, and (iii) as disclosed in the Company's reports filed with the SEC pursuant to the Exchange Act (including transactions related to the Gant Company), neither the Company nor any Company Subsidiary has made any payment to, or received any payment from, or made or received any investment in, or entered into any transaction with, any Affiliate, including without limitation, the purchase, sale or exchange of property or the rendering of any service, where the amount involved is material to the business of the Company.
- Section 3.23. Insurance. There is in full force and effect certain policies of insurance issued by insurers of recognized responsibility, including, without limitation, fire and casualty insurance, insuring the Company and each Company Subsidiary and their properties against such

losses and risks, and in such amounts, as are usual and customary in the case of companies engaged in the same or similar business and similarly situated.

Section 3.24. Customers. Except as set forth in Schedule 3.24, neither the Company nor any Company Subsidiary has received any written notice that any customer has ceased, or will cease, or has substantially reduced, or will substantially reduce, their overall purchases of the products or goods of the Company or Company Subsidiary, in each case which could reasonably have a Material Adverse Effect.

Section 3.25. Financial Advisors. Except as set forth on Schedule 3.25, no agent, broker, investment banker, finder, financial advisor or other Person is or will be entitled to any broker's or finder's fee or any other commission or similar fee from the Company, directly or indirectly, in connection with the Contemplated Transactions.

Section 3.26. DGCL Section 203 and Rights Agreement. The Board of Directors has taken all action necessary to exempt from the provisions of Section 203 of the Delaware General Corporation Law ("DGCL Section 203") and from being deemed an "Acquiring Person" under the Rights Agreement, to the extent applicable, this Agreement, any acquisition by the Investors or their Affiliates of the Series B Stock pursuant to this Agreement, and the Certificate of Designations and any conversion or exercise by the Investors or their Affiliates of the Series B Stock into Common Stock. No agreement or instrument to which the Company or any of the Company Subsidiaries is a party or by which any of them is bound, and no state statute similar to DGCL Section 203 that is applicable to the Company or any of the Company Subsidiaries imposes any restrictions on business combinations or similar transactions with interested stockholders of a nature similar to those set forth in DGCL Section 203.

Section 3.27. D&O Insurance. The Company maintains director and officer liability insurance coverage in the aggregate amount of \$20\$ million, which policy has a \$5,000 per individual deductible up to \$50,000 in the aggregate.

Section 3.28. Solvency. The Company is not, and after giving effect to the Contemplated Transactions, will not be, insolvent within the meaning of Title 11 of the United States Code, the General Corporation Law of the State of Delaware, or the General Laws of the State of New York.

Section 3.29. Full Disclosure. No representation or warranty made by the Company in this Agreement or any other Transaction Document contains or will contain any untrue statement of a material fact, or omits to state a material fact necessary to make the statements contained herein or therein not misleading.

Section 3.30. No General Solicitation. None of the Company or any of its "affiliates" (as defined in Rule 501(b) of Regulation D under the Securities Act ("Regulation D")), has, directly or through an agent, engaged in any form of general solicitation or general advertising in connection with the offering of the Series B Shares (as those terms are used in Regulation D)

under the Securities Act or in any manner involving a public offering within the meaning of Section 4(2) of the Securities Act; and the Company has not entered into any contractual arrangement with respect to the distribution of the Series B Shares except for this Agreement and the Registration Rights Agreement, and the Company will not enter into any such arrangement.

Section 3.31. No Integration. None of the Company or any of its Affiliates (other than the Initial Purchaser in connection with the transactions contemplated by this Agreement about which no representation is made by the Company) has, directly or through any agent, sold, offered for sale, solicited offers to buy or otherwise negotiated in respect of, any "security" (as defined in the Securities Act) which is or will be integrated with the sale of the Series B Shares or the Common Stock in a manner that would require the registration under the Securities Act of the Series B Shares or the Common Stock. The Company has not sold or issued any Series B Shares, shares of Common Stock, any security convertible into Series B Shares or shares of Common Stock, or any security of the same class as the Series B Shares or Common Stock during the six-month period prior to the Closing Date, including any sales pursuant to Rule 144A, Regulation D or Regulation S under the Securities Act, other than shares issued pursuant to employee benefit plans, qualified stock options plans or other employee compensation plans or pursuant to outstanding options, rights or warrants.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF THE INITIAL PURCHASER AND THE INVESTORS

Section 4.01. Initial Purchaser Representations.

- (a) The Initial Purchaser represents and warrants to and agrees with the Company that it is an institutional accredited investor, as defined in Rule 501(a)(1), (2), (3) or (7) under the Securities Act, as the case may be, with such knowledge and experience in financial and business matters as is necessary in order to evaluate the merits and risks of an investment in the Series B Shares, and (ii) it is not acquiring the Series B Shares with any present intention of offering or selling any of the Series B Shares in a transaction that would violate the Securities Act or the securities laws of any state of the United States or any other applicable jurisdiction
- (b) The Initial Purchaser hereby represents and warrants to the Company that it (i) has not solicited offers for, or offered or sold, the Series B Shares by means of any form of general solicitation or general advertising or in any manner involving a public offering within the meaning of Section 4(2) of the Securities Act and (ii) has only solicited offers for the Series B Shares from persons whom the Initial Purchaser reasonably believed to be accredited investors.

Section 4.02. Investors Representations. Each of the Investors represents and warrants, severally and not jointly, to the Company as follows:

- (a) Authorization. Such Investor is a limited partnership duly organized and validly existing under the Laws of the state or country of its jurisdiction of formation. Such Investor has the full power and authority to enter into this Agreement and the other Transaction Documents and to consummate the transactions contemplated hereby and thereby. The execution and delivery of this Agreement and the other Transaction Documents and the consummation by the Investors of the transactions contemplated hereby and thereby have been duly and authorized by all necessary action on the part of the Investors. This Agreement and the other Transaction Documents have been and will be, as the case may be, duly executed and delivered by the Investors and constitute legal, valid and binding obligations of the Investors, enforceable in accordance with their respective terms, except as such enforceability may be subject to the effects of any applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or similar Laws affecting creditors' rights generally and subject to the effects of general equitable principles.
- (b) Investment Representations. Such Investor is an Accredited Investor and is acquiring the Series B Shares allocated to such Investor for such Investor's own account, for investment, and not with a view to, or for sale in connection with, the distribution thereof or of any interest therein. Such Investor has adequate net worth and means of providing for its current needs and contingencies and is able to sustain a complete loss of the investment in such Series B Shares, and has no need for liquidity in such investment. Such Investor, itself or through its officers, employees or agents, has sufficient knowledge and experience in financial and business matters to be capable of evaluating the merits and risks of an investment such as an investment in the Series B Shares, and such Investor, either alone or through its officers, employees or agents, has evaluated the merits and risks of the investment in such Series B Shares. Such Investor understands that the Series B Stock has not been registered under the Securities Act by reason of its issuance in a transaction exempt from the registration requirements of the Securities Act pursuant to the exemption provided in Section 4(2), and that the Series B Stock may not be sold or otherwise disposed of unless registered under the Securities Act or exempted from such registration.
- (c) Investors' Acknowledgment. Each Investor has had the opportunity, directly or through its representatives, to ask questions of and receive answers from Persons acting on behalf of the Company concerning the transactions contemplated by this Agreement.
- (d) Financial Advisors. No agent, broker, investment banker, finder, financial advisor or other Person is or will be entitled to any broker's or finder's fee or any other commission or similar fee from any of the Investors, directly or indirectly, in connection with any of the transactions contemplated by this Agreement or any of the Transaction Documents.

(e) Legend.

(i) The certificates evidencing the Series B Stock and the Common Stock issuable upon conversion of the Series B Stock will bear a legend (the "Legend") substantially similar to the following:

"THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE BEEN ACQUIRED FOR INVESTMENT AND HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED. NO INTEREST IN THESE SECURITIES MAY BE PLEDGED, HYPOTHECATED, SOLD, TRANSFERRED OR OTHERWISE DISPOSED OF EXCEPT WHILE A REGISTRATION STATEMENT IS IN EFFECT UNDER SAID ACT OR PURSUANT TO AN EXEMPTION FROM REGISTRATION UNDER SAID ACT. THIS CERTIFICATE IS ISSUED PURSUANT TO AND SUBJECT TO THE RESTRICTIONS ON TRANSFER AND OTHER PROVISIONS OF THE INVESTORS' RIGHTS AGREEMENT BETWEEN THE COMPANY, AND THE INVESTORS REFERRED TO THEREIN, A COPY OF WHICH IS ON FILE WITH THE COMPANY. EXCEPT AS PROVIDED IN SUCH AGREEMENT, THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE NOT TRANSFERABLE AND ANY PURPORTED TRANSFER IN VIOLATION OF THE PROVISIONS OF SUCH AGREEMENT SHALL BE VOID AND OF NO FORCE AND EFFECT.

- (ii) The legend endorsed on the certificates pursuant to Section 4.02(e) hereof shall be removed and the Company shall issue a certificate without such legend to the holder thereof at such time as the securities evidenced thereby cease to be restricted securities upon the earliest to occur of (i) a registration statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities shall have been disposed of in accordance with such registration statement, (ii) the securities shall have been sold to the public pursuant to Rule 144 (or any successor provision) under the Securities Act, and (iii) such securities may be sold by the holder without restriction or registration under Rule 144(k) under the Securities Act (or any successor provision).
- (f) No Reliance. Each Investor acknowledges that the Company is not making any representation, warranty, covenant or agreement, other than as set forth in this Agreement and the other Transaction Documents.

ARTICLE V COVENANTS OF THE COMPANY

The Company covenants and agrees that for so long as the Series B Shares are outstanding:

Section 5.01. Accounting System. The Company will maintain a system of accounting and proper books of record and account, in accordance with GAAP.

Section 5.02. Taxes. The Company will, and will cause each Company Subsidiary to, pay when due (a) all Taxes imposed upon it or any of its properties or income, other than Taxes which could not reasonably have a Material Adverse Effect, Taxes which are being contested in good faith by appropriate proceedings and Taxes which are subject to filings that are on valid extensions, and (b) all claims or demands of materialmen, mechanics, carriers, warehousemen, landlords and other like Persons which, if unpaid, might result in the creation of a Lien upon any of its properties other than claims or demands which are being contested in good faith or could not reasonably have a Material Adverse Effect.

Section 5.03. Corporate Existence. The Company will preserve and keep in full force and effect its corporate existence and the corporate existence of each Company Subsidiary (unless merged into the Company) and all rights and franchises of the Company and the Company Subsidiaries unless, in the good faith judgment of the Company, the termination of or failure to preserve or keep in full force and effect such corporate existence, right or franchise could not reasonably have a Material Adverse Effect.

Section 5.04. Maintenance of Properties. The Company will, and will cause each Company Subsidiary to, maintain all of its properties necessary for the conduct of its business in good condition, repair and working order (normal wear and tear excepted) and cause to be made all necessary repairs, renewals, replacements, betterments and improvements thereof, all as in the judgment of the Company may be necessary so that the business carried on in connection therewith may be conducted at all times.

Section 5.05. Insurance. Each of the Company and the Company Subsidiaries will maintain insurance with insurers of recognized responsibility, insuring the Company, the Company Subsidiaries and their respective properties, assets, business and projects against such losses and risks, and in such amounts, as the Company reasonably determines. Section 5.06. Compliance with Law. The Company will, and will cause each Company Subsidiary to, comply, in all material respects, with all material Laws which it reasonably believes are applicable with respect to the conduct of their respective businesses and the ownership of their respective properties, provided that the Company shall not be deemed to be in violation of this Section 5.06 as a result of any failure to comply with any provisions of such Laws, the noncompliance with which would not result in enforceable fines, penalties, injunctive relief or other civil or criminal liabilities which, in the aggregate, could reasonably have a Material Adverse Effect.

Section 5.07. Maintain Listing. The Company will use commercially reasonable efforts to (x) maintain the listing and trading of its Common Stock on the NYSE, for so long as the Company qualifies for such listing under the rules and regulations of the NYSE and (y) comply in all material respects with the Company's reporting, filing, and other obligations, under the rules and regulations of the NYSE. In the event that the Common Stock is no longer eligible for listing and trading on the NYSE, the Company will use commercially reasonable efforts to secure the listing or quotation of the Common Stock on the Nasdaq National Market, the Nasdaq

SmallCap Market or the American Stock Exchange (if such listing is permitted by the bylaws, rules or regulations of any of the foregoing) and to comply in all material respects with the Company's reporting, filing and other obligations under the bylaws or rules of such exchanges or the National Association of Securities Dealers, Inc., as applicable. The Company will promptly provide to the Initial Purchaser and each of the Investor Representatives copies of any notices it receives from the NYSE and any other exchange or quotation system on which the Common Stock is then listed regarding the continued eligibility of the Common Stock for listing on such exchanges or quotation systems.

Section 5.08. Secure Listing. The shares of Common Stock issuable upon conversion of the Series B Stock shall have been duly listed, pending notice of issuance, on the NYSE and the Company shall maintain such listing in accordance with Section 5.07.

Section 5.09. Transfer Taxes. The Company shall be responsible for any Liability with respect to any transfer, stamp or similar non-income Taxes that may be payable in connection with the execution, delivery and performance of this Agreement including, without limitation, any such Taxes with respect to the issuance of the Series B Shares or shares of Common Stock issuable upon conversion thereof.

ARTICLE VI ACTIONS PRIOR TO CLOSING

Section 6.01. Access to Information. Until the Closing Date or the earlier termination of this Agreement:

- (a) The Company shall provide the Initial Purchaser and each of the Investor Representatives with (i) monthly reports, as soon as practicable and in any event within 20 calendar days after the close of each such month, (ii) copies of: (A) the unaudited balance sheet of the Company as of the end of such month, (B) the unaudited statements of operations of the Company for such month, and (C) the unaudited statements of changes in cash flows of the Company for such month setting forth in each case in comparative form the corresponding figures for the preceding month and, with respect to statements relating to the Company's 2002 fiscal year, for the 2002 Budget, for the year to date and for the comparable periods in the preceding year; and (iii) such other financial and operating data as the Initial Purchaser and any Investor Representative may reasonably request.
- (b) The Company will permit the Initial Purchaser and representatives of the Investors to visit and inspect any of the properties of the Company or any of the Company Subsidiaries, to examine the corporate books, records, agreements and files of the Company and make copies or extracts therefrom and to request information at reasonable times and intervals concerning the general status of the Company's financial condition and operations, all upon reasonable notice and at such reasonable times and as often as the Initial Purchaser and such Investor may reasonably request. The Investors will, and will instruct each of their respective Affiliates and advisors to, hold in confidence all "Information" as provided in the letter

agreement between Apax Partners, Inc., an Affiliate of the Investors, and the Company, dated January 17, 2002.

Section 6.02. Conduct of Business.

- (a) Except as set forth in Schedule 6.02(a), or as otherwise agreed to in writing by the Initial Purchaser and the Investors, from the date hereof to the Closing Date, the Company shall, and shall cause each Company Subsidiary to (i) use its commercially reasonable efforts to maintain all of the material assets it owns or uses in the ordinary course of business consistent with past practice (subject to the good faith judgement of management of the Company); (ii) use its commercially reasonable efforts to preserve the goodwill and ongoing operations of its business; (iii) maintain its books and records in the usual, regular and ordinary manner, on a basis consistent with past practice; (iv) use its commercially reasonable efforts to maintain insurance in full force and effect with respect to its business with responsible companies, comparable in amount, scope and coverage to that in effect on the date of this Agreement; and (v) comply in all material respects with applicable material Laws.
- (b) Without limiting the generality of the foregoing, except as expressly contemplated by this Agreement or as set forth on Schedule 6.02(b), between the date hereof and the Closing Date, the Company shall not, and shall cause each Company Subsidiary not to, do any of the following without the prior written consent of the Investors:
 - (i) engage in any act, other than in the ordinary course of business and consistent with past practice, which could reasonably have a Material Adverse Effect or in any way prevent the consummation of the Contemplated Transactions;
 - (ii) declare, set a record date, set aside or authorize the payment of, any dividend or other distribution in respect of any shares of Capital Stock of the Company or repurchase, redeem or acquire any of the outstanding shares of Capital Stock of the Company, except for the Company's regular dividends;
 - (iii) transfer, issue, sell or dispose of any shares of its Capital Stock or grant any option, warrant, call or other right to purchase or otherwise acquire shares of its Capital Stock, except for (x) grants of options (and the issuance of any shares of its Capital Stock upon exercise of such options) under the Company's employee option plans, and (y) the issuance of the Warrants;
 - (iv) except with respect to the CK Acquisition, mortgage or pledge (other than pursuant to the Credit Agreement or with respect to Permitted Liens) any of its assets, or acquire any material assets or sell, assign, transfer, convey or otherwise dispose of any material assets of the Company or the Company Subsidiaries, except in the ordinary course of business;

- (v) except for the CK Purchase Agreement, enter into any Contract to make any capital expenditures for the Company's fiscal year ending February 2, 2003 obligating the Company or any Company Subsidiary to pay an amount in excess of \$1,000,000 or an amount which together with amounts already expended or committed during its 2002 fiscal year would exceed the amount budgeted for capital expenditures in the 2002 Budget;
- (vi) except with respect to the CK Acquisition or under the Credit Facility, create, incur, assume or guarantee any debt for borrowed money in excess of \$5,000,000, whether due or to become due, except debt incurred in the ordinary course of business;
- (vii) change its accounting principles, methods or practices (including, without limitation, its method of accounting for stock options) other than a change which is required by reason of a concurrent change in Law or GAAP;
- (viii) cancel or terminate any material insurance policy naming it as a beneficiary or a loss payee without obtaining substitute insurance coverage except where such cancellation or termination could not reasonably have a Material Adverse Effect;
 - (ix) amend its Certificate of Incorporation or By-Laws; or
- (x) agree, whether in writing or otherwise, to do any of the foregoing.

Section 6.03. No Solicitation. From the date hereof to the Closing Date, the Company agrees that neither it nor the Company Subsidiaries shall, and shall not permit any of their respective directors, officers, Employees, representatives or agents to, directly or indirectly, solicit or initiate any discussions or negotiations with, participate in any negotiations with, provide any information to or otherwise cooperate in any other way with, or facilitate or encourage any effort or attempt by, any Person, other than the Initial Purchaser, the Investors and their respective, partners, directors, officers, employees, representatives and agents, respecting any Competing Offer. The Company will promptly advise the Initial Purchaser and each Investor Representative of any proposal or inquiry made to it, the Company Subsidiaries or any of their respective directors, officers, Employees, representatives, or agents with respect to any of the foregoing transactions.

Section 6.04. Consent. Each of the parties hereto will use its reasonable best efforts and shall fully cooperate with each other party to make promptly all registrations, filings and applications, give all notices and obtain all Consents in connection with the transactions contemplated hereby.

Section 6.05. HSR. The Company and the Investors shall, as promptly as practicable following the execution and delivery of this Agreement, submit all filings required by the HSR Act to the Federal Trade Commission and the Department of Justice and thereafter provide any supplemental information requested in connection therewith pursuant to the HSR Act and make

any similar filing within, to the extent reasonably practicable, a similar time frame with any other Governmental Body for which such filing is required. Any such notification and report form and supplemental information will be in substantial compliance with the requirements of the HSR Act or other applicable antitrust regulation. The Company and the Investors shall furnish to each other such necessary information and reasonable assistance as the other may request in connection with its preparation of any filing or submission which is necessary under the HSR Act or other applicable antitrust regulation. The Company and the Investors shall request early termination of the applicable waiting period under the HSR Act and any other applicable antitrust regulation; shall respond with reasonable diligence and dispatch to any request for additional information made in response to such filings or information requests made by any other Governmental Body and shall keep each other apprised of any communications with, and inquiries or requests for additional information from the Federal Trade Commission, Department of Justice or any other Governmental Body and shall comply with any inquiry or request made thereby. Notwithstanding anything to the contrary contained herein, the Company will pay the filing fees of the Investors with respect to the notification and report forms required by the HSR Act.

Section 6.06. Notice of Breach. From the date hereof through the Closing Date, as promptly as practicable, and in any event not later than five Business Days after the Company becomes aware thereof, the Company shall provide the Initial Purchaser and each Investor Representative with written notice of (a) any representation or warranty of the CK Companies or the Sellers contained in the CK Purchase Agreement being untrue or inaccurate in any material respect, or (b) any failure of the CK Companies or the Sellers to comply with or satisfy in any material respect any covenant, condition or agreement to be complied with or satisfied by such Person under any of the CK Transaction Documents.

Section 6.07. Other Transaction Documents. The Company shall keep the Initial Purchaser, each Investor Representative and their respective counsel fully informed of, provide the Initial Purchaser, each Investor Representative and their respective counsel with copies of all drafts of, and, at the request of the Initial Purchaser or any Investor Representative, discuss with the Initial Purchaser, each Investor Representative and their respective counsel on a timely basis, each of the CK Transaction Documents and shall not, without the prior consent of the Initial Purchaser and each Investor Representative (which consent shall not be unreasonably withheld or delayed), (i) execute any of the CK Transaction Documents to be executed after the date hereof or (ii) amend, waive, supplement or modify any material provisions of any of the CK Transaction Documents executed on or prior to the date hereof.

Section 6.08. Publicity. The parties agree not to issue any announcement, press release, public statement or other information to the press or any third party with respect to this Agreement or the Contemplated Transactions without obtaining the prior written approval of the other parties hereto (which approval shall not be unreasonably withheld); provided, however, that nothing contained herein shall prevent any party hereto, at any time, from furnishing any required information to any Governmental Body or from issuing any announcement, press release, public statement or other information to the press or any third party with respect to this

Agreement or the Contemplated Transactions if required by Law, although, the parties agree to consult with each other as to the content of any release so required and consider in good faith the comments of the other thereon.

ARTICLE VII CONDITIONS TO CLOSING

Section 7.01. Conditions to Obligations of the Initial Purchaser and the Investors. The obligation of the Initial Purchaser and the Investors to consummate the transactions contemplated hereby shall be subject to the fulfillment on or prior to the Closing Date of the following conditions, any or all of which may be waived by the Initial Purchaser and the Investors, in whole or in part (except with respect to conditions 7.01 (n) (i) and (r) which may be waived only by the Initial Purchaser and except with respect to condition 7.01 (n) (ii) which may be waived only by the Investors) to the extent permitted by applicable Law:

- (a) Consents. The Company shall have obtained the consents set forth in Schedule 3.05, in form and substance reasonably satisfactory to the Initial Purchaser and the Investors.
- (b) Material Adverse Effect. Between the date hereof and the Closing Date, none of the CK Companies shall have suffered or experienced a CK Companies Material Adverse Effect.
- (c) No Governmental Order or Other Proceeding or Litigation. No Order of any Governmental Body shall be in effect that restrains or prohibits the Contemplated Transactions.
- (d) CK Transaction Documents. (i) The CK Transaction Documents executed prior to or simultaneously with this Agreement shall not have been amended, modified, supplemented, or provisions thereof waived, in violation of Section 6.07, and (ii) the CK Transaction Documents executed subsequent to this Agreement shall have been entered into in accordance with Section 6.07.
- (e) CK Acquisition. The transactions contemplated by the CK Transaction Documents that by their terms are to be completed concurrently with, or immediately after, the Closing shall be capable of being completed at the Closing and the closing of the CK Acquisition in accordance with the terms and provisions of the applicable CK Transaction Document and all conditions required in the CK Transaction Documents to be satisfied on or prior to closing of the CK Acquisition shall have been satisfied or capable of being satisfied without waiver, amendment or modification unless otherwise approved in writing by the Initial Purchaser and each of the Investor Representatives. The closing of the CK Acquisition shall occur substantially simultaneous with the Closing hereunder.

- (f) HSR Act. All filing and other requirements under the HSR Act shall have been satisfied, and the applicable waiting period with respect to the Contemplated Transactions under the HSR Act shall have expired or been terminated.
- (g) Closing Fees. The Company shall have paid or reimbursed the Investors for all of the expenses specified in Section 10.01(a).
- (h) Legal Opinion. The Initial Purchaser and the Investors shall have received, dated the Closing Date and addressed to each of the Initial Purchaser and the Investors, an opinion of Katten Muchin Zavis Rosenman, counsel to the Company, substantially in the form attached hereto as Exhibit C.
- (i) Certificate of Designations. The Certificate of Designations shall have been duly filed with the Secretary of State of Delaware and shall be in full force and effect.
- (j) Directors. The Series B Designees shall have been appointed to the Board of Directors, effective as of the Closing and the Board of Directors shall consist of no more than 14 directors.
- (k) Registration Rights Agreement. The Registration Rights Agreement shall have been executed and delivered by the parties thereto and shall be in full force and effect, in substantially the form attached hereto as Exhibit D (the "Registration Rights Agreement").
- (1) Investors' Rights Agreement. The Investors Rights Agreement shall have been executed and delivered by the parties thereto and shall be in full force and effect, in substantially the form attached hereto as Exhibit E (the "Investors' Rights Agreement").
- (m) Amendment to the Rights Agreement. The Fifth Amendment to the Rights Agreement shall have been executed and delivered by the parties thereto and shall be in full force and effect, in substantially the form attached hereto as Exhibit F (the "Fifth Amendment").

(n) Stock Certificates.

- (i) The Company shall have delivered to the Initial Purchaser a certificate or certificates representing the Series B Shares, duly registered in the name of the Initial Purchaser.
- (ii) The Initial Purchaser shall have delivered to the Investor Representatives certificates representing the Series B Shares in the amounts specified on Schedule 2.01, duly registered by the Company in the respective names of the Investors as set forth in Schedule 2.01.
- (o) NYSE Listing. The shares of Common Stock issuable upon conversion of the Series B Stock shall have been listed, pending notice of issuance, on the NYSE and the NYSE shall not have withdrawn its advice that the transactions contemplated under this

Agreement constitute a bona fide private financing under paragraph 312.03 of the NYSE Listed Company Manual.

- (p) Loan Documents. Each of the Loan Documents shall have been executed and delivered by the parties thereto and shall be in full force and effect and the closing of the loan transaction contemplated therein shall occur substantially simultaneously with the Closing hereunder.
- (q) Good Standing; Company Certificates. The Company shall have delivered to the Initial Purchaser and the Investor Representatives:
 - (i) a certificate issued by the appropriate Governmental Body evidencing, as of a recent date, the good standing of the Company in its jurisdiction of incorporation;
 - (ii) a certificate, dated the Closing Date, executed by the Secretary of the Company which certifies that (A) attached to such certificate is a complete and correct copy of the Certificate of Incorporation of the Company certified by the Secretary of State of the State of Delaware, and that there has been no amendment to the Certificate of Incorporation of the Company since that the date of such certification, and (B) attached to such certificate is a complete and correct copy of the By-Laws of the Company, as in full force and effect at the Closing Date; and
 - (iii) a certificate, dated the Closing Date, executed by the Secretary of the Company, which certifies as complete and correct resolutions of the Board of Directors: (A) authorizing the execution, delivery and performance of this Agreement and each of the other Transaction Documents, the issuance and sale of the Series B Stock and the issuance of the shares of Common Stock issuable upon conversion of the Series B Stock, the reservation of such shares of Common Stock and the performance of the transactions contemplated by this Agreement and the other Transaction Documents, (B) electing the Series B Designees, (C) implementing the actions with respect to DGCL Section 203 and the Rights Agreement set forth in Section 3.26, and (D) authorizing the issuance of the Rights and the issuance of the shares of Series A Stock issuable upon exercise of the Rights.
- (r) Comfort Letter. The Initial Purchaser shall have received from Ernst & Young LLP a letter addressed to it, in form and substance reasonably satisfactory to the Initial Purchaser, and dated as of the date hereof, (i) confirming that they are independent public accountants within the meaning of the Securities Act and are in compliance with the applicable requirements relating to the qualification of accountants under Rule 2-01 of Regulation S-X and (ii) stating, as of the date hereof (or, with respect to matters involving changes or developments since the respective dates as of which specified financial information is given in the Commission Filings filed in 2002, as of a date not more than five days prior to the date hereof), the conclusions and findings of such firm with respect to the financial information and other matters ordinarily covered by accountants' "comfort letters" to underwriters.

- Section 7.02. Conditions to Obligations of the Company. The obligation of the Company to consummate the transactions contemplated hereby shall be subject to the fulfillment on or prior to the Closing Date of the following conditions, any or all of which may be waived by the Company:
- (a) No Governmental Order or Other Proceeding or Litigation. No Order of any Governmental Body shall be in effect that restrains or prohibits the Contemplated Transactions.
- (b) CK Acquisition. The transactions contemplated by the CK Transaction Documents that by their terms are to be completed concurrently with, or immediately after, the Closing shall be capable of being completed at the Closing and the closing of the CK Acquisition in accordance with the terms and provisions of the applicable CK Transaction Document.
- (c) HSR Act. All filing and other requirements under the HSR Act shall have been satisfied, and the applicable waiting period with respect to the transactions described herein together with any extensions thereof, under the HSR Act, shall have expired or been terminated.
- (d) NYSE Listing. The shares of Common Stock issuable upon conversion of the Series B Stock shall have been listed, pending notice of issuance on the NYSE and the NYSE shall not have withdrawn its advice that the transactions contemplated under this Agreement constitute a bona fide private financing under paragraph 312.03 of the NYSE Listed Company Manual.
- (e) Registration Rights Agreement. The Registration Rights Agreement shall have been executed and delivered by the parties thereto and shall be in full force and effect.
- (f) Investors' Rights Agreement. The Investors Rights Agreement shall have been executed and delivered by the parties thereto and shall be in full force and effect.
- (g) Purchase Price. The Initial Purchaser shall have delivered to the Company the Initial Purchase Price.

ARTICLE VIII SURVIVAL

Section 8.01. Survival. The representations, warranties and covenants to be performed at or prior to Closing of the parties set forth in this Agreement shall survive for a period of 24 months following the execution and delivery of this Agreement and thereafter shall be of no further force or effect, provided that the representations and warranties set forth in Sections 3.11 (Taxes), 3.16 (ERISA) and 3.20 (Environmental Protection) shall survive for the applicable period of the statute of limitations and the representations and warranties set forth in Sections 3.01 (Organization), 3.02 (Authorization) and 3.04 (Capitalization) shall survive indefinitely (or if indefinite survival is not permitted by Law, then for the maximum period permitted by

applicable Law). No representations, warranties or covenants of the Company in this Agreement shall be affected by any examination made for or on behalf of the Initial Purchaser or the Investors, the knowledge of the Initial Purchaser or the Investors of any certificate or opinion. Following the expiration of the periods set forth above with respect to any particular representation or warranty, no party hereto shall have any further liability with respect to such representation or warranty. Except as set forth herein, all of the covenants, agreements and obligations of the parties hereto shall survive the Closing indefinitely (or if indefinite survival is not permitted by Law, then for the maximum period permitted by applicable Law). Anything herein to the contrary notwithstanding, any claim for indemnification that is asserted by written notice which notice specifies in reasonable detail the facts upon which such claim is made as provided in this Section 8.01 within the survival period shall survive until resolved pursuant to a final non-appealable judicial determination or otherwise.

ARTICLE IX INDEMNIFICATION

Section 9.01. Generally. Subject to the limitations and other provisions of this Article IX, the Company covenants and agrees to indemnify, defend and hold harmless the Investor Group Members and the Initial Purchaser Group Members from and against (but only to the extent of) any and all Losses resulting from, incurred in connection with or arising out of (but only to the extent of) (a) any breach of any representation, warranty, covenant or agreement of the Company contained herein or in any other Transaction Document, (b) the failure of the Company to perform any of the agreements, covenants or obligations contained herein or in any other Transaction Documents, or (c) any claims by the Sellers relating to the CK Acquisition (other than if any such claim was a result of a breach by the Investors under this Agreement). Subject to the limitations and other provisions of this Article IX, each Investor, severally and not jointly, covenants and agrees to indemnify, defend and hold harmless the Company Group Members from and against (but only to the extent of) any and all Losses resulting from, incurred in connection with or arising out of (but only to the extent of) (a) any breach of any representation, warranty, covenant or agreement of such Investor contained herein or in any other Transaction Document, or (b) the failure of such Investor to perform any of the agreements, covenants or obligations of such Investor contained herein or in any other Transaction Document. The term "Loss" or any similar term shall mean any and all damages, reduction in value of the original investment in the Series B Shares, deficiencies, costs, claims, fines, judgments, amounts paid in settlement, expenses of investigation, interest, penalties, assessments, out-of-pocket expenses (including reasonable attorneys' and auditors' fees and disbursements, witness fees and court costs). The party or parties being indemnified are referred to herein as the "Indemnitee" and the indemnifying party is referred to herein as the "Indemnitor."

- (a) Any party who receives notice of a potential claim that may, in the judgment of such party, result in a Loss shall use all reasonable efforts to provide the parties hereto notice thereof, provided that failure or delay or alleged delay in providing such notice shall not adversely affect such party's right to indemnification hereunder, unless and then only to the extent that such failure or delay or alleged delay has resulted in actual prejudice to the Indemnitor, including, without limitation, by the expiration of a statute of limitations. In the event that any party shall incur or suffer any Losses in respect of which indemnification may be sought by such party hereunder, the Indemnitee shall assert a claim for indemnification by written notice (a "Notice") to the Indemnitor stating the nature and basis of such claim. In the case of Losses arising by reason of any third party claim, the Notice shall be given within thirty (30) days of the filing or other written assertion of any such claim against the Indemnitee, but the failure of the Indemnitee to give the Notice within such time period shall not relieve the Indemnitor of any liability that the Indemnitor may have to the Indemnitee, except to the extent that the Indemnitor demonstrates that the defense of such action has been materially prejudiced by the Indemnitee's failure to timely give such Notice.
- (b) In the case of third party claims for which indemnification is sought, the Indemnitor shall, if necessary, retain counsel reasonably satisfactory to the Indemnitee, and have the option (i) to conduct any proceedings or negotiations in connection therewith, (ii) to take all other steps to settle or defend any such claim (provided that the Indemnitor shall not settle any such claim without the consent of the Indemnitee which consent shall not be unreasonably withheld) and (iii) to employ counsel to contest any such claim or liability in the name of the Indemnitee or otherwise. In any event, the Indemnitee shall be entitled to participate at its own expense and by its own counsel in any proceedings relating to any third party claim. The Indemnitor shall, within 15 Business Days of receipt of the Notice, notify the Indemnitee of its intention to assume the defense of such claim. If (i) the Indemnitor shall decline to assume the defense of any such claim, (ii) the Indemnitor shall fail to notify the Indemnitee within 15 Business Days after receipt of the Notice of the Indemnitor's election to defend such claim, (iii) the Indemnitee shall have reasonably concluded that there may be defenses available to it which are different from or in addition to those available to the Indemnitor (in which case the Indemnitor shall not have the right to direct the defense of such action on behalf of the Indemnitee), or (iv) a conflict exists between the Indemnitor and the Indemnitee which the Indemnitee has reasonably concluded would prejudice the Indemnitor's defense of such action, then in each such case the Indemnitor shall not have the right to direct the defense of such action on behalf of the Indemnitee and the Indemnitee shall, at the sole expense of the Indemnitor, defend against such claim and (x) in the event of a circumstance described in clause (i) and (ii), the Indemnitee may settle such claim without the consent of the Indemnitor (and the Indemnitor may not challenge the reasonableness of any such settlement) and (y) in the event of a circumstance described in clause (iii) and (iv), the Indemnitee may not settle such claim without the consent of the Indemnitor (which consent will not be unreasonably withheld or delayed). The reasonable expenses of all proceedings, contests or lawsuits in respect of such claims shall be borne and paid by the Indemnitor if the Indemnitee is entitled to indemnification hereunder

and the Indemnitor shall pay the Indemnitee, in immediately available funds, the amount of any Losses, within a reasonable time of the incurrence of such Losses. Regardless of which party shall assume the defense or negotiation of the settlement of the claim, the parties agree to cooperate fully with one another in connection therewith. In the event that any Losses incurred by the Indemnitee do not involve payment by the Indemnitee of a third party claim, then, the Indemnitor shall, within 20 days after written notice from the Indemnitee specifying the amount of Losses, pay to the Indemnitee, in immediately available funds, the amount of such Losses. Any disagreement between the Indemnitor and Indemnitee relating to the amount of the Losses shall be resolved by arbitration in accordance with the rules of the American Arbitration Association (the "AAA") then pertaining in the City of New York, New York, by a single arbitrator to be mutually agreed upon by the parties or, if they are unable to so agree, by an arbitrator selected by the AAA. Anything in this Article X to the contrary notwithstanding, the Indemnitor shall not, without the Indemnitee's prior written consent, settle or compromise any claim or consent to entry of any judgment in respect thereof which imposes any future obligation on the Indemnitee or which does not include, as an unconditional term thereof, the giving by the claimant or plaintiff to the Indemnitee, a release from all liability in respect of such claim.

Section 9.03. Certain Qualifications. The Material Adverse Effect and other materiality (or correlative meaning) qualifications included in the representations, warranties, covenants and agreements contained herein shall have no effect on any provisions in this Article IX concerning the indemnities of the Company with respect to such representations, warranties, covenants and agreements, each of which shall be read as though there were no Material Adverse Effect or other materiality qualification for purposes of such indemnities. All knowledge qualifications included in the representations, warranties, covenants and agreements contained herein shall have no effect on any provisions of this Article IX concerning the indemnities of the Company with respect to such representations, warranties, covenants and agreements, each of which shall be read as though there were no such knowledge qualifications for purposes of such indemnities.

Section 9.04. Limitations on Indemnification.

- (a) No Investor Group Member shall be entitled to be indemnified hereunder unless and until the aggregate of all Losses incurred by Investor Group Members shall exceed \$7,000,000 (the "Basket"); provided, however, that the Basket shall not apply to any Losses incurred by the Investor Group Members with respect to any third party claim against any Investor Group Member for which the Investor Group Members are entitled to indemnity pursuant to Section 9.01.

 Notwithstanding anything to the contrary contained herein, the liability of (i) the Company under this Article IX shall be limited to an amount equal to the Purchase Price; and (ii) the Investors in the aggregate under this Article IX and under the Loan Documents shall be limited to an amount equal to \$100,000,000.
- (b) The parties acknowledge and agree that the indemnification provisions contained herein shall be the sole and exclusive remedy for Losses arising out of or caused by the breach of any of the representations, warranties, covenants or agreements of the parties

contained in this Agreement, except (i) with respect to claims arising out of fraud, or (ii) for the amounts payable pursuant to Articles X and XI hereof.

ARTICLE X FEES, EXPENSES AND COSTS

Section 10.01. Reimbursement.

- (a) In the event that the transactions contemplated by this Agreement are consummated, the Company agrees to pay at the Closing and hold the Investors harmless against liability for the payment of (i) all reasonable legal fees and expenses owed by the Investors to their counsel, incurred in connection with this Agreement, (ii) stamp and other transfer Taxes which may be payable in respect of (A) the execution and delivery of this Agreement or (B) the issuance of the Series B Stock and the Common Stock issuable upon the conversion of the Series B Stock, (iii) all other reasonable costs and expenses (including, without limitation, accounting expenses and consultants' fees) incurred by the Investors in connection with this Agreement (clauses (i), (ii) and (iii) of this subsection (a) are referred to collectively as, the "Fees and Expenses"), and (iv) all reasonable fees and expenses owed by the Investors to Monitor Company incurred in connection with this Agreement (the "Monitor Expenses").
- (b) In the event that the transactions contemplated by the CK Purchase Agreement or a similar transaction in which the Company acquires voting control of the CK Companies or substantially all of their assets are consummated within one year from the date of termination of this Agreement and the transactions contemplated by this Agreement are not consummated (other than by reason of a material breach hereunder by the Investors), the Company agrees to pay on demand the Maximum Amount.
- (c) In the event that none of the Contemplated Transactions are consummated and the Company has been, or will be, reimbursed by CKI or any of its Affiliates for all of the Company's Expenses relating to the CK Acquisition, the Company agrees to pay on demand the Maximum Amount.
- (d) In the event that none of the Contemplated Transactions are consummated and the Company has been, and will only be, partially reimbursed by CKI or any of its Affiliates for the Company's Expenses relating to the CK Acquisition (the "Partial Reimbursement"), the Company agrees to pay on demand the Minimum Amount. In addition to the foregoing, if the aggregate amount of the Partial Reimbursement exceeds the Minimum Amount (the "Excess Amount"), the Company shall pay to the Investors an amount equal to the Percentage of the Excess Amount; provided, however, that in no event shall the Company pay to the Investors an amount in excess of the Maximum Amount or less than the Minimum Amount.
- (e) In the event that none of the Contemplated Transactions are consummated and Company has not been, and will not be, reimbursed by CKI or any of its Affiliates for any of the Company's Expenses, the Company agrees to pay on demand the Minimum Amount.

- (f) Notwithstanding anything to the contrary contained herein, the Company agrees to reimburse the Initial Purchaser and Investors on demand for the Initial Purchaser's and the Investors' reasonable out-of-pocket fees and expenses incurred in connection with any amendment to, or waiver of, this Agreement and the other Transaction Documents.
- (g) The Company shall pay and hold the Initial Purchaser harmless against liability for the payment of (i) all reasonable legal fees and expenses owed by the Initial Purchaser to its counsel, incurred in connection with this Agreement, (ii) stamp and other transfer Taxes which may be payable in respect of (A) the execution and delivery of this Agreement, (B) the issuance of the Series B Stock and the Common Stock issuable upon the conversion of the Series B Stock, or (C) the transfer of the Series B Stock to the Investors, and (iii) all other reasonable costs and expenses (including, without limitation, accounting expenses and consultants' fees) incurred by the Initial Purchaser in connection with this Agreement.

ARTICLE XI

Section 11.01. Termination. This Agreement may be terminated on or any time prior to the Closing:

- (a) by the mutual written consent of each of the Investors and the Company; or
- (b) by either the Company or the Investors if the CK Purchase Agreement shall have been terminated pursuant to its terms; or
- (c) by the Investors if the Closing shall not have occurred (unless the failure of such occurrence shall be due to the failure of the Investors to perform or observe any material agreement set forth herein required to be performed or observed by the Investors on or before the Closing) on or before March 31, 2003; provided that if the Company and the Investors agree to extend the period within which the Closing must occur beyond March 31, 2003, the Initial Purchaser shall also agree to extend beyond March 31, 2003 (but in any event not beyond June 30, 2003).

Section 11.02. Effect Of Termination. In the event of the termination of this Agreement as provided in Section 11.01, all obligations and agreements of the parties set forth in this Agreement shall forthwith become void except for the obligations set forth in: (i) the last sentence of Section 6.05 (HSR) (which shall remain in full force and effect) (ii) Section 6.08 (Publicity), (iii) Article IX (Indemnification), (iv) Section 10.01(b), (c), (d), (e) and (g) (Fees and Expenses) (as applicable), (v) Section 11.03 (Termination Fee) and (vi) Section 11.04 (Break-up Fee), and there shall be no liability or obligation on the part of the parties hereto except as otherwise provided in this Agreement. Notwithstanding the foregoing, the termination of this Agreement under Section 11.01 shall not relieve either party of any liability for breach of this Agreement prior to the date of termination.

Section 11.03. Termination Fee. The Company shall pay to the Investors (pro rata at their respective commitments) a termination fee of (a) cash in the amount of \$22,500,000, and (b) the Termination Warrants (referred to collectively as the "Termination Fee"), in the event that:

- (i) the transactions contemplated by the CK Purchase Agreement are consummated and the transactions contemplated by this Agreement are not consummated; or
- (ii) the Company consummates a similar transaction, within one year of the termination of this Agreement, in which the Company acquires voting control of CKI or substantially all of its assets and the Investors do not participate in the financing of such similar transaction.

Notwithstanding the foregoing, the Investors shall not be entitled to a Termination Fee in the event that the transactions contemplated by this Agreement are not consummated (x) by reason of a default hereunder by the Investors or (y) due to the waiting period with respect to Investors' filing not having expired by the Termination Date by reason of competitive issues raised by the Federal Trade Commission or the Department of Justice regarding the Investors investment in the Company.

Section 11.04. Break-up Fee. In the event that the Contemplated Transactions are not consummated and the Break-up Fee is paid, the Company agrees to pay to the Investors an amount equal to 33% of the Break-up Fee.

Section 11.05. CK Recovery. The Company shall pay to the Investors 33% of any CK Recovery.

ARTICLE XII MISCELLANEOUS

Section 12.01. Notices and Addresses. Any notice, demand, request, waiver, or other communication under this Agreement shall be in writing and shall be deemed to have been duly given on the date of service, if personally served or sent by facsimile; on the Business Day after notice is delivered to a courier or mailed by express mail, if sent by courier delivery service or express mail for next day delivery; and on the third day after mailing, if mailed to the party to whom notice is to be given, by first class mail, registered, return receipt requested, postage prepaid and addressed as follows:

If to the Company:

Phillips-Van Heusen Corporation 200 Madison Avenue New York, New York 10016 Attention: Vice President, General Counsel and Secretary Facsimile: (212) 381-3970 Telephone: (212) 381-3509

with a copy to:

Katten Muchin Zavis Rosenman

575 Madison Avenue

New York, New York 10022

Attention: David H. Landau, Esq.

Facsimile: (212) 940-8776 Telephone: (212) 940-8800

If to the Initial Purchaser:

Lehman Brothers Inc.

745 Seventh Avenue

New York, New York 10019

Attention: Global Retailing/Consumer Group Facsimile: (212) 520-9602

with a copy to Office of the General Counsel

Lehman Brothers Inc.

399 Park Avenue

10th Floor

New York, New York 10022

Attention: Director of Litigation Facsimile: (212) 526-3633; and

Simpson Thacher & Bartlett

425 Lexington Avenue

New York, New York 10017

Attention: Andrew Keller, Esq. Facsimile: (212) 455-2502

If to any Investor, to the Investor Representatives:

Apax Managers, Inc.

445 Park Avenue, 11th Floor

New York, NY 10022

Attention: David H. Landau

Facsimile: (212) 753-6300 Telephone: (212) 319-6155

Apax Partners Europe Managers Ltd.

15 Portland Place London W1B 1PT United Kingdom

Attention: Adrian Beecroft and Clive Sherling Facsimile: 44-20-7872-6475

Facsimile: 44-20-7872-6475 Telephone: 44-20-7872-6300

With a copy to:

Swidler Berlin Shereff Friedman, LLP The Chrysler Building 405 Lexington Avenue
New York, New York 10174
Attention: Morris Orens, Esq.
Facsimile: (212) 891-9598

Section 12.02. Captions. The captions in this Agreement are for convenience of reference only and shall not be given any effect in the interpretation of this Agreement.

Section 12.03. No Waiver. The failure of a party to insist upon strict adherence to any term of this Agreement on any occasion shall not be considered a waiver or deprive that party of the right thereafter to insist upon strict adherence to that term or any other term of this Agreement. Any waiver must be in writing. Any of the covenants or agreements contained in this Agreement may be waived only by the written consent of the Initial Purchaser and the Investor Representative.

Section 12.04. Severability. If one or more provisions of this Agreement are held to be unenforceable under applicable Law, such provision(s) shall be excluded from this Agreement and the balance of this Agreement shall be interpreted as if such provision were so excluded and shall be enforceable in accordance with its terms so long as the economic or legal substance of the transactions contemplated by this Agreement are not affected in any manner materially adverse to any party.

Section 12.05. Exclusive Agreement; Amendment. This Agreement supersedes all prior agreements among the parties with respect to its subject matter, other than the letter agreement dated January 17, 2002 with respect to the non-disclosure of confidential information, is intended (with the documents referred to herein) as a complete and exclusive statement of the terms of the agreement among the parties with respect thereto and cannot be changed or terminated except by a written instrument executed by the party or parties against whom enforcement thereof is

sought, except that, with respect to the Investors, this Agreement may be amended by a written instrument executed by the Investor Representative.

Section 12.06. Limitation on Assignment; Parties in Interest.

- (a) No assignment of this Agreement or of any rights or obligations hereunder may be made by the Company or the Investors (by operation of Law or otherwise) without the prior written consent of the other parties hereto and any attempted assignment without the required consents shall be void; provided, however, that each Investor may validly assign any of its rights and obligations hereunder to an Affiliate of such Investor without the prior written consent of the other parties hereto; provided, that such Investor shall remain liable for any and all obligations assigned to such Affiliate.
- (b) This Agreement shall be binding upon, and shall inure to the benefit of, and be enforceable by, the parties and their respective successors, transferees and assigns.
- Section 12.07. Obligations of Investors Several. The obligations of the Investors hereunder shall be several and not joint. No Investor shall be responsible for the breach of any provision of this Agreement by any other Investor.
- Section 12.08. Governing Law. This Agreement and (unless otherwise provided) all amendments hereof and waivers and consents hereunder shall be governed by the internal Laws of the State of New York, without regard to the conflicts of Law principles thereof which would specify the application of the Law of another jurisdiction.
- Section 12.09. Jurisdiction. Each of the Investors and the Company (a) hereby irrevocably and unconditionally submits to the exclusive jurisdiction of any state or federal court sitting in New York County, New York for the purposes of any suit, action or other proceeding arising out of this Agreement or the subject matter hereof brought by the Company, or any Investor and (b) hereby waives and agrees not to assert, by way of motion, as a defense, or otherwise, in any such suit, action or proceeding, any claim that it is not subject personally to the jurisdiction of the above-named courts, that its property is exempt or immune from attachment or execution, that the suit, action or proceeding is brought in an inconvenient forum, that the venue of the suit, action or proceeding is improper or that this Agreement or the subject matter hereof may not be enforced in or by such court.
- Section 12.10. No Third Party Beneficiary. The terms and provisions of this Agreement are intended solely for the benefit of each party hereto and their respective successors or permitted assigns, and it is not the intention of the parties to confer third-party beneficiary rights upon any other Person other than any Person entitled to indemnity under Article IX.
- Section 12.11. Injunctive Relief. In the event that any party threatens to take any action prohibited by this Agreement, the parties agree that there may not be an adequate remedy at law. Accordingly, in such an event, a party may seek and obtain preliminary and permanent injunctive

relief (without the necessity of posting any bond or undertaking). Such remedies shall, however, be cumulative and not exclusive and shall be in addition to any other remedies which any party may have under this Agreement or otherwise.

Section 12.12. Counterparts. This Agreement may be executed via facsimile and in any number of counterparts, each of which shall be deemed to be an original instrument and all of which together shall constitute one and the same instrument.

Section 12.13. Actions Simultaneous. All actions to be taken and all documents to be executed and delivered by all parties at the Closing shall be deemed to have been taken and executed and delivered simultaneously and no actions shall be deemed to have been taken nor shall any documents be deemed to have been executed and delivered until all actions have been taken and all documents have been executed and delivered.

[Signature Pages Follow]

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date first above written.

PHILLIPS-VAN HEUSEN CORPORATION

By: /s/ Mark D. Fishcer

Name: Mark D. Fischer Title: Vice President APAX EXCELSIOR VI, L.P.

By: Apax Excelsior VI Partners, L.P.,

Its General Partner
By: Apax Managers, Inc.
Its General Partner

By: /s/ David Landau

Name: David Landau Title: Vice President

APAX EXCELSIOR VI-A C.V.

By: Apax Excelsior VI Partners, L.P.,

Its General Partner
By: Apax Managers, Inc.
Its General Partner

By: /s/ David Landau

Name: David Landau Title: Vice President

APAX EXCELSIOR VI-B C.V.

By: Apax Excelsior VI Partners, L.P.,

Its General Partner
By: Apax Managers, Inc.
Its General Partner

By: /s/ David Landau

Name: David Landau Title: Vice President

PATRICOF PRIVATE INVESTMENT CLUB III, L.P.

By: Apax Excelsior VI Partners, L.P.,

Its General Partner
By: Apax Managers, Inc.
Its General Partner

By: /s/ David Landau

Name: David Landau Title: Vice President APAX EUROPE V A-L.P.

By: Apax Partners Europe Managers Ltd. Its Investment Manager

By: /s/ Adrian Beecroft

Name: Adrian Beecroft Title: Managing Director

By: /s/ Clive Sherling

Name: Clive Sherling Title: Managing Director LEHMAN BROTHERS INC.

By: /s/ Dominic Rispoli

Authorized Representative

FROM: Phillips-Van Heusen Corporation

200 Madison Avenue New York NY 10016

Rubenstein Associates, Inc.
Public Relations - Tel.: 212 843 8000
Contact: Scott Tagliarino/Marcia Horowitz

For Calvin Klein Inc.: Don Nathan/Jim Badenhausen (212) 484-7782

For Apax Partners: Todd Fogarty (212) 521-4854

PHILLIPS-VAN HEUSEN TO ACQUIRE CALVIN KLEIN, INC.
Combines Powerful Global Brand with Strong Infrastructure
and Financial Resources Necessary to Fuel Growth

PVH Will Develop Growth Opportunities with New Lifestyle Offerings, Retail and Global Expansion and Increased Marketing Support

NEW YORK, December 17, 2002 - Phillips-Van Heusen Corporation (NYSE: PVH) today announced that it entered into a definitive agreement to acquire Calvin Klein, Inc. (CKI), one of the world's leading lifestyle brands, in a transaction approaching \$700 million.

The value of the transaction includes \$400 million in cash, approximately \$30 million in PVH stock as well as warrants and an ongoing financial incentive for Mr. Klein based on future sales of the Calvin Klein brand. Apax Partners, a leading private equity firm, partnered with PVH and will provide a major part of the financing for the transaction, which is subject to normal regulatory review and is expected to close within 60 days.

Under the terms of the agreement, CKI's existing design and marketing organization will continue as a separate and distinct operating unit. Mr. Klein will retain a significant financial interest in the combined entity's success and be involved in key strategic issues, including growth opportunities, overall design direction, brand positioning and marketing strategy. CKI President Tom Murry will continue as President and Chief Operating Officer of the organization.

"Calvin Klein is one of the most powerful lifestyle brands in the world and has tremendous untapped growth potential that we look forward to working with Calvin to realize," said Bruce Klatsky, Chairman and Chief Executive Officer of PVH.

"We have tremendous respect for what Calvin, Barry Schwartz and their team have built and are strongly committed to preserving the integrity of the brand as well as its exclusive distribution."

Mr. Klein said, "This combination is all about growing our brand and business over the long term in multiple markets and consumer segments. Phillips-Van Heusen understands and appreciates the unique values and integrity of the brand and has the infrastructure and financial capabilities to expand the business beyond what we can achieve as a private company. Over the past several years, we have assembled design, merchandising and marketing teams at CKI to ensure that our brand philosophy is consistently reflected in all aspects of our business. With those teams in place, I look forward to working with the people of Phillips-Van Heusen, who have a unique position in and understanding of our core business, and are committed to helping us successfully realize the great potential of our brand."

POWERFUL GROWTH OPPORTUNITIES

Mr. Klatsky described the deal as a "transforming transaction" that will have a major impact on accelerating PVH's top line growth, profit margins and earnings. The acquisition combines the extraordinary brand presence of Calvin Klein with the infrastructure of PVH, acknowledged as one of the best-managed companies in its industry.

Key opportunities to grow the Calvin Klein business include the launch of better men's and women's sportswear and accessories lines; global expansion through a strategic plan together with its licensing partners; global expansion through Calvin Klein retail stores, and taking advantage of additional growth opportunities in Europe and Asia. Phillips-Van Heusen expects to launch the new collection of men's and women's better sportswear within the next 24 months. In addition, the company will support growth of the brand by increasing marketing support beyond the \$250 million a year in consumer marketing currently spent on the Calvin Klein brand by its licensees and the company, Mr. Klatsky said.

The company said that that it expects the acquisition to make a major contribution to its growth in earnings per share. The acquisition is anticipated to be somewhat dilutive in 2003 and to make a positive contribution to annual earnings by 2004. During 2003, due to normal integration and transition costs, operating earnings per share are expected to be in a range of \$1.00 to \$1.05. Earnings per share for fiscal 2004 and beyond are expected to grow at an annual rate of 15-20 percent.

-more-

"Acquiring Calvin Klein is the realization of our long-standing strategy of taking world-class brands with great growth potential that we can layer onto our existing infrastructure and expanding them dramatically in ways that fit with the brand's positioning," Mr. Klatsky said. "This is a tremendous opportunity for our company, and we believe that Calvin Klein will thrive as a part of PVH because of our track record of successfully growing and supporting leading brands with a strong, efficient infrastructure."

Apax Partners will provide financing for the transaction in the form of a \$250 million equity investment in PVH convertible preferred stock, as well as a \$125 million, two-year secured note. The preferred stock carries an eight-percent dividend payable in kind or in cash at the company's option and a \$14 conversion price into PVH common stock.

Mr. Klatsky said, "We chose to partner with Apax because of their expertise and track record in the consumer products and retail sectors. They have proven to be highly creative and supportive throughout this process."

David Landau, General Partner of Apax, said, "PVH has an outstanding management team. We are excited to back their acquisition of this powerful brand, which we believe offers significant growth opportunities worldwide."

Phillips-Van Heusen Corporation is the largest shirt company in the world and one of the leading apparel and footwear companies. Its well-known brands include Van Heusen, IZOD, G.H. Bass, as well as licensing agreements with Geoffrey Beene, ck Calvin Klein, Arrow, DKNY and Kenneth Cole.

Calvin Klein, Inc. is one of the leading design and marketing companies in the world. The Company consists of a highly successful collection business and an extensive network of licensing partners that together generate over \$3 billion in annual retail sales worldwide. The Company's brands include the Calvin Klein Collection, cK and cK Calvin Klein Jeans. Products designed under these lifestyles include apparel, accessories, shoes, underwear, sleepwear, hosiery, socks, swimwear, eyewear, watches, coats, and fragrance, as well as products for the home.

Apax Partners is one of the world's leading international private equity investment groups, managing and advising more than \$11 billion worldwide. With over 30 years of direct investing experience, Apax focuses on the following industry sectors: retail/consumer products, information technology, telecommunications, healthcare, media and financial services.

Advisors to the transaction include for Phillips-Van Heusen, the Blackstone Group, Lehman Brothers, JP Morgan Chase, and Katten Muchin Zavis Rosenman; for Calvin Klein, Violy, Byorum & Partners Holding Company LLC, and for Calvin Klein, Inc., Grubman, Indursky & Schindler, P.C., Paul, Weiss, Rifkind, Wharton & Garrison.

_ ______

THE COMPANY WILL WEBCAST A CONFERENCE CALL TO REVIEW THE CALVIN KLEIN TRANSACTION ON TUESDAY, DECEMBER 17, 2002 AT 4:30 P.M. EST.] PLEASE LOG ON EITHER TO OUR WEB SITE AT WWW.PVH.COM AND GO TO THE INVESTOR INFO PAGE OR TO CCBN'S WEBSITE AT WWW.COMPANYBOARDROOM.COM TO LISTEN TO THE LIVE WEBCAST OF THE CONFERENCE CALL. THE WEBCAST WILL BE AVAILABLE FOR REPLAY FOR 30 DAYS AFTER IT IS HELD, COMMENCING APPROXIMATELY TWO HOURS AFTER THE LIVE BROADCAST ENDS. PLEASE LOG ON TO WWW.PVH.COM OR WWW.COMPANYBOARDROOM.COM AS DESCRIBED ABOVE TO LISTEN TO THE REPLAY. IN ADDITION, AN AUDIO REPLAY OF THE CONFERENCE CALL IS AVAILABLE FOR 48 HOURS STARTING ONE HOUR AFTER IT IS HELD. THE REPLAY OF THE CONFERENCE CALL CAN BE ACCESSED BY CALLING 1-800-428-6051 AND USING PASSCODE: 272543. THE CONFERENCE CALL AND WEBCAST CONSIST OF COPYRIGHTED MATERIAL. THEY MAY NOT BE RE-RECORDED, REPRODUCED, RETRANSMITTED, REBROADCAST OR OTHERWISE USED WITHOUT THE COMPANY'S EXPRESS WRITTEN PERMISSION. YOUR PARTICIPATION REPRESENTS YOUR CONSENT TO THESE TERMS AND CONDITIONS, WHICH ARE GOVERNED BY NEW YORK LAW.

SAFE HARBOR STATEMENT UNDER THE PRIVATE SECURITIES LITIGATION REFORM ACT OF 1995: Forward-looking statements in this press release and made during the conference call / webcast, including, without limitation, statements relating to the Company's plans, strategies, objectives, expectations and intentions, are made pursuant to the safe harbor provisions of the Private Securities Litigation Reform Act of 1995. Investors are cautioned that such forward-looking statements are inherently subject to risks and uncertainties, many of which cannot be predicted with accuracy, and some of which might not be anticipated, including, without limitation, the following: (i) the Company's plans, strategies, objectives, expectations and intentions are subject to change at any time at the discretion of the Company; (ii) the levels of sales of the Company's apparel and footwear products, both to its wholesale customers and in its retail stores, and the extent of discounts and promotional pricing in which the Company is required to engage, all of which can be affected by weather conditions, changes in the economy, fuel prices, reductions in travel, fashion trends and other factors; (iii) the Company's plans and results of operations will be affected by the Company's ability to manage its growth and inventory; (iv) the Company's operations and results could be affected by quota restrictions (which, among other things, could limit the Company's ability to produce products in cost-effective countries that have the labor and technical expertise needed), the availability and cost of raw materials (particularly petroleum-based synthetic fabrics, which are currently in high demand), the Company's ability to adjust timely to changes in trade regulations and the migration and development of manufacturers (which can affect where the Company's products can best be produced), and civil conflict or war and political and labor instability in the countries where the Company's products are or are planned to be produced; and (v) acquisitions and issues arising with acquisitions and proposed transactions, and (vi) other risks and uncertainties indicated from time to time in the Company's filings with the Securities and Exchange Commission.

The Company does not undertake any obligation to update publicly any forward-looking statement, including, without limitation, any estimate regarding revenues or earnings, whether as a result of the receipt of new information, future events or otherwise.

#